

WestminsterResearch

http://www.westminster.ac.uk/westminsterresearch

Design and implementation of generalized topologies of timeinterleaved variable bandpass $\Sigma - \Delta$ modulators Kalafat Kizilkaya, I.

This is an electronic version of a PhD thesis awarded by the University of Westminster. © Mrs Isil Kalafat Kizilkaya, 2016.

The WestminsterResearch online digital archive at the University of Westminster aims to make the research output of the University available to a wider audience. Copyright and Moral Rights remain with the authors and/or copyright owners.

Whilst further distribution of specific materials from within this archive is forbidden, you may freely distribute the URL of WestminsterResearch: ((http://westminsterresearch.wmin.ac.uk/).

In case of abuse or copyright appearing without permission e-mail repository@westminster.ac.uk

DESIGN AND IMPLEMENTATION OF GENERALIZED TOPOLOGIES OF TIME-INTERLEAVED VARIABLE BANDPASS Σ - Δ MODULATORS

A Thesis/A Dissertation

By

Işıl KALAFAT KIZILKAYA

Department of Engineering

Submitted in partial fulfilment of the requirements of University of Westminster for the degree of Doctor of Philosophy

July, 2016

ABSTRACT

In this thesis, novel analog-to-digital and digital-to-analog generalized time-interleaved variable bandpass sigma-delta modulators are designed, analysed, evaluated and implemented that are suitable for high performance data conversion for a broad-spectrum of applications. These generalized time-interleaved variable bandpass sigma-delta modulators can perform noise-shaping for any centre frequency from DC to Nyquist. The proposed topologies are well-suited for Butterworth, Chebyshev, inverse-Chebyshev and elliptical filters, where designers have the flexibility of specifying the centre frequency, bandwidth as well as the passband and stopband attenuation parameters. The application of the time-interleaving approach, in combination with these bandpass loop-filters, not only overcomes the limitations that are associated with conventional and mid-band resonator-based bandpass sigma-delta modulators, but also offers an elegant means to increase the conversion bandwidth, thereby relaxing the need to use faster or higher-order sigma-delta modulators.

A step-by-step design technique has been developed for the design of time-interleaved variable bandpass sigma-delta modulators. Using this technique, an assortment of lower- and higher-order single- and multi-path generalized A/D variable bandpass sigma-delta modulators were designed, evaluated and compared in terms of their signal-to-noise ratios, hardware complexity, stability, tonality and sensitivity for ideal and non-ideal topologies. Extensive behavioural-level simulations verified that one of the proposed topologies not only used fewer coefficients but also exhibited greater robustness to non-idealties.

Furthermore, second-, fourth- and sixth-order single- and multi-path digital variable bandpass digital sigma-delta modulators are designed using this technique. The mathematical modelling and evaluation of tones caused by the finite wordlengths of these digital multi-path sigma-delta modulators, when excited by sinusoidal input signals, are also derived from first-principles and verified using simulation and experimental results. The fourth-order digital variable-band sigma-delta modulator topologies are implemented in VHDL and synthesized on Xilinx® SpartanTM-3 Development Kit using fixed-point arithmetic. Circuit outputs were taken via RS232 connection provided on the FPGA board and evaluated using MATLAB routines developed by the author. These routines included the decimation process as well. The experiments undertaken by the author further validated the design methodology presented in the work.

In addition, a novel tunable and reconfigurable second-order variable bandpass sigma-delta modulator has been designed and evaluated at the behavioural-level. This topology offers a flexible set of choices for designers and can operate either in single- or dual-mode enabling multi-band implementations on a single digital variable bandpass sigma-delta modulator.

This work is also supported by a novel user-friendly design and evaluation tool that has been developed in MATLAB/Simulink that can speed-up the design, evaluation and comparison of analog and digital single-stage and time-interleaved variable bandpass sigma-delta modulators. This tool enables the user to specify the conversion type, topology, loop-filter type, path number and oversampling ratio.

ACKNOWLEDGMENTS

I would like to thank life in the first place, for providing me such a good opportunity to move to London and work in such a good university with such good colleagues. Yet it also gave me my beautiful baby girl Feraye who has no idea how inspiring she is. Among all my greatest present from life is my husband Burak, who has always been patient, loving and caring.

I would like to present my deep respect and love to my supervisors Dr. Mohammed Al-Janabi and Prof. Izzet Kale. I always felt exclusive due to Dr. Mohammed Al-Janabi's endless patience, flexibility and availability. I guess I was one of the luckiest PhD researchers in the university as he always stood behind me and gave his time and effort in this work. Prof. Izzet Kale, as my second supervisor, has always shed light on my path therefore encouraging this work with his profound knowledge and experience.

Finally, I want to dedicate this thesis to my parents. To my impatience father and mother, Kahraman and Leyla, who everyday asked about the progress of my thesis. To my father- and mother-in-law, Turgut and Nurcan, who always showed respect to my dream of being a doctor and never complained that I took away their precious son and grandchild away from home to pursue this aim.

TABLE OF CONTENTS

Chapter		Page
ABSTRA	СТ	II
ACKNOV	VLEDGMENTS	
		117
TABLE	DF CONTENTS	IV
LIST OF	TABLES	VII
LIST OF	FIGURES	VIII
ABBREV	/IATIONS	XIII
СНАРТЕ	R 1. INTRODUCTION	2
1.1	INTEGRATION, LOOP-FILTER AND STABILITY	2
1.2	HIGH OSR, SPEED AND BANDWIDTH	3
1.3	OUTLINE	4
СНАРТЕ	R 2. DATA CONVERTERS	6
2.1	SAMPLING AND PRE-FILTERING	
2.2	OUANTIZATION	
2.3	Σ - Δ Modulation	
2.3.1	Noise-Shaping	
2.3.2	OVERSAMPLING	14
2.4	BP Σ - Δ Modulation	14
2.5	Motivations for the TI VBP Σ - Δ Modulators	
Contr	IBUTIONS AND NOVELTIES	20
2.6	Performance Parameters	21
2.6.1	SIGNAL TO NOISE RATIO	21
2.6.2	DYNAMIC RANGE	23
2.6.3	STABILITY	23
2.6.4	TONALITY	25
2.6.5	DITHERING	27
2.7	CONCLUDING REMARKS	27
СНАРТЕ	R 3. A/D TI VBP Σ-Δ MODULATORS	28
3.1	VBP Σ - Δ Modulator Design	
3.2	GENERALIZED VBP NTF DESIGN	
3.3	VBP Σ - Δ Modulator Topologies	33
3.3.1	CHAIN OF INTEGRATORS WITH FEEDBACK AND FEEDFORWARD COEFFICIENTS (CI-FBFF)	
3.3.2	THE TI TECHNIQUE	35
3.3.3	DOWNSAMPLER	
3.3.4	UPSAMPLER	
3.3.5	NODE EQUATION METHOD	
3.3.6	Design Technique to Build Up Generalize TI VBP Σ – Δ Modulators	
3.3.7	CHAIN OF RESONATORS WITH FEEDFORWARD AND LOCAL RESONATOR FEEDBACK COEFFICE	ENTS
(CR-R	FB)	43

(CR-RFB) = 433.4 HARDWARE COMPLEXITY OF THE A/D VBP Σ - Δ MODULATORS = 45

3.5	BEHAVIOURAL LEVEL SIMULATIONS OF THE CI-FBFF TOPOLOGY	46		
3.6	6 ANALYSIS OF THE BEHAVIOURAL-LEVEL TOPOLOGIES			
3.7	COEFFICIENT COMPARISON			
3.8	FILTERING PERFORMANCE			
Butte	RWORTH NTFS	60		
Сневу	SHEV NTFS	66		
INVERS	SE-CHEBYSHEV NTFS	70		
Ellipt	ICAL NTFS	73		
3.9	FILTERING COMPARISON	74		
3.10	Non-Idealities	76		
3.11	QUANTIZATION TONES			
3.11.1	SAWTOOTH QUANTIZATION NOISE MODEL, ONE-BIT QUANTIZER, SINGLE SINUSOID			
3.11.2	SP VBP Σ – Δ Modulators Excited by Single Sinusoids	90		
3.11.3	TI VBP Σ – Δ Modulators Excited by Single Sinusoids	91		
3.12	Conclusion	94		
снарте	Έ <i>Α</i> Ο / Α ΤΙ VRD Σ- Α ΜΟΟΙΙΙ ΑΤΟΡς	07		
	Output Feedback and Feedback Topol ocies	98		
4.2	I OOD-FILTED DESIGN	00		
ч.2 4.2.1		101		
4.2.1		105		
т.2.2 Д З	ΓΑΚΤΟΟΕΑΚΤΚΕΟΟΕΝΟΙΕΣ	105		
т.5 Л.Л.	REHAVIOIDAL I EVEL SIMILIATIONS	100		
т.т <i>1</i> . <i>1</i> . 1	COEFEICIENT COMDADISON	109		
т.т.1 4.4.2	ΕΙΙ ΤΕΦΙΝΟ ΡΕΦΕΩΡΜΑΝΟΕ ΑΝΟ ΓΩΜΡΑΦΙΩΟΝ	107		
4.5	INDI EMENTATION			
4.51	ΙΜΓLΕΜΕΝΤΑΤΙΟΝ Γινερ Δοιντ Δριτμμετίς	112 11 <i>1</i>		
4.5.1		115		
4.5.2	DESIGN DUTTLENECK	113 120		
4.5.5	IMPLEMENTATION RESULTS	120		
4.6	HARDWARE COMPLEXITY OF THE D/A VBP $\Sigma = \Delta$ MODULATORS			
4./	A I UNABLE AND RECONFIGURABLE $D/A VBP \Sigma - \Delta$ MODULATOR	125		
4.8	CONCLUSION	128		
СНАРТЕ	R 5. GRAPHICAL USER INTERFACE	130		
5.1	INSTALLATION	130		
5.2	THE MENU BAR	131		
5.3	OUTPUTS	133		
5.4	Programmer Guide	136		
5.5	Conclusion	140		
СНАРТЕ	R 6. CONCLUDING REMARKS AND RECOMMENDATIONS FOR FUTURE WO	RK.142		
6.1	Concluding Remarks			
6.2	RECOMMENDATIONS FOR FUTURE WORK			
APPEND	DIX A	150		
ΑΡΡΕΝΠ	IX B	162		
APPEND				
		197		
	NCES	102		
NEFERE	NCE3	192		

BIOGRAPHY	05
-----------	----

LIST OF TABLES

Table 1: Comparison of Σ -A Modulators in Use	19
Table 2 · Building Blocks of Simulink	34
Table 3: CI-FBFF Topology SNRs and DRs Butterworth Centre Frequency = 0.3 BW = 0.1	02
OSRSP = 32 OSR2P = 64 OSR4P = 128	65
Table 4: CR-RFB Topology SNRs and DRs Butterworth Centre Frequency = 0.3 BW = 0.0	.05
OSRSP = 32 OSR2P = 64 OSR4P = 128	, <u> </u>
Table 5: CI-FBFF Topology SNRs and DRs Chebyshev Centre Frequency =0.4 BW =0.0	
PBA=2 dB, $OSRSP = 32$, $OSR2P = 64$, $OSR4P = 128$. 68
Table 6: CR-RFB Topology SNRs and DRs, Chebyshev, Centre Frequency =0.4, BW =0.0	1,
PBA=0.5 dB, OSRSP = 32, OSR2P = 64, OSR4P = 128	. 68
Table 7: CI-FBFF Topology SNRs and DRs, Inverse-Chebyshev, Centre Frequency =0.46,	
BW =0.001, SBA = 80 dB , $OSRSP = 32$, $OSR2P = 64$, $OSR4P = 128$. 71
Table 8: CR-RFB Topology SNRs and DRs, Inverse-Chebyshev, Centre Frequency =0.4, B	3W
= 0.01, SBA $= 80$ dB, $OSRSP = 16$, $OSR2P = 32$, $OSR4P = 64$. 71
Table 9: CI-FBFF Topology SNRs and DRs, Elliptical, Centre Frequency =0.3, BW =0.02,	
PBA = 1dB, SBA=80dB, <i>OSRSP</i> = 32, <i>OSR2P</i> = 64, <i>OSR4P</i> = 128	. 74
Table 10: CR-RFB Topology SNRs and DRs, Elliptical, Centre Frequency =0.1, BW =0.01	,
PBA = 1dB, SBA=80dB, <i>OSRSP</i> = 16, <i>OSR2P</i> = 32, <i>OSR4P</i> = 64	. 74
Table 11: Tolerable Error Percentage for SP CI-FBFF Topology	. 79
Table 12: Tolerable Error Percentage for SP CR-RFB Topology	. 80
Table 13: Tolerable Error Percentage for 2P CI-FBFF Topology	. 82
Table 14: Tolerable Error Percentage for 4P CI-FBFF Topology	. 82
Table 15: Tolerable Error Percentage for 2P CR-RFB Topology	. 82
Table 16: Tolerable Error Percentage for 4P CR-RFB Topology	. 83
Table 17: Particular Frequencies and Corresponding Loop-filter's Feedback Coefficients for	r
the OF Topology	106
Table 18: Chosen Design Parameters	113
Table 19: Resulting Feedforward Coefficients	113
Table 20: Resulting Feedback Coefficients	113
Table 21: Simulated SNRs, Behavioural Results	121
Table 22: Measured SNRs, Behavioural Results	121
Table 23: The Allocated FPGA Sources for The EF Topologies	124
Table 24: The Allocated FPGA Sources for The OF Topologies	124
Table 25: Simulink Blocks' Names	135
Table 26: Functions for Loop-filter Transfer Function Calculation	137
Table 27: Variable Names Used in det_variables.m.	138
Table 28: Functions called by det_coef.m	139

LIST OF FIGURES

	Page
Figure 1: Data Conversion Block Diagram	6
Figure 2: Sampled Signal Spectrum	7
Figure 3: AAF and Aliasing	8
Figure 4: PCM A/D Converter	8
Figure 5: Quantization of a Sampled Analog Signal	9
Figure 6: AWGN Block Diagram	9
Figure 7: White Quantization Noise	9
Figure 8: Oversampling and Quantization Noise	11
Figure 9: A/D Σ - Δ Modulator	12
Figure 10: D/A Σ - Δ Modulator	12
Figure 11: Noise Shaping and Oversampling	13
Figure 12: AWGN Model in Σ - Δ Modulator	13
Figure 13: Delayed Resonator Block	15
Figure 14: Resonator-Based Second Order (Effective First Order) Σ - Δ modulator's Outp	out
Plots for Different Centre Frequencies	16
Figure 15: Comparison of ADCs	17
Figure 16: Linear Stability Model	24
Figure 17: Split Quantizer	25
Figure 18: a) Limit Cycle Tones b) Idle Channel Tones	26
Figure 19: Superheterodyne Receiver	28
Figure 20: Homodyne Receiver	28
Figure 21: Prospective Receiver with Tunable ADC	29
Figure 22: Output Plots for Different Clock Frequencies	31
Figure 23: The OF Σ - Δ Modulator Block Diagram	32
Figure 24: 2 nd -order SP CI-FBFF Topology	35
Figure 25: Block Diagram of the TI Topology	36
Figure 26: Downsampler Block Diagram	38
Figure 27: Upsampler Block Diagram	38
Figure 28: 2-level Downsampler Behavioural Model	40
Figure 29: 2 nd -order 2-path CI-FBFF Topology	42
Figure 30: 2 nd -order 4-path CI-FBFF Topology	42
Figure 31: 2 nd -order SP CR-RFB Topology	44
Figure 32: 2 nd -order 2-path CR-RFB Topology	44
Figure 33: 2 nd -order 4-path CR-RFB Topology	45
Figure 34: Windowing and Overlapping	48
Figure 35: SP 2 nd -order, Butterworth Filtering	49
Figure 36: 2P 4 th -order, Chebyshev Filtering	49
Figure 37: 4P 6 th -order, Elliptical Filtering	50
Figure 38: 4P 8 th -order, Inverse-Chebyshev Filtering, Zoomed Plot	50
Figure 39: 2 nd -order, Butterworth NTF Feedback Coefficients, BW is fixed to 0.02	51
Figure 40: 4 th -order, Butterworth NTF Feedback Coefficients, BW is fixed to 0.02	52
Figure 41: 6 th -order, Butterworth NTF Feedback Coefficients, BW is fixed to 0.02	52

Figure 42: 8 th -order, Butterworth NTF Feedback Coefficients, BW is fixed to 0.02	52
Figure 43: 2 nd -order, Butterworth NTF Feedforward Coefficients, BW is fixed to 0.02	54
Figure 44: 4 th -order, Butterworth NTF Feedforward Coefficients, BW is fixed to 0.02	55
Figure 45: 6 th -order, Butterworth NTF Feedforward Coefficients, BW is fixed to 0.02	
Figure 46. 8 th -order Butterworth NTF Feedforward Coefficients, BW is fixed to 0.02	55
Figure 47: 2^{nd} -order Elliptical NTF Feedforward Coefficients Centre Freq = 0.1	57
Figure 48: 2^{nd} -order Elliptical NTF Feedforward Coefficients, Centre Freq = 0.4	57
Figure 49: 4^{th} -order Elliptical NTF Feedforward Coefficients, Centre Freq. = 0.1	57
Figure 50: 4^{th} -order, Elliptical NTF Feedforward Coefficients, Centre Freq. = 0.4	58
Figure 51: 6^{th} -order, Elliptical NTF Feedforward Coefficients, Centre Freq. = 0.1	58
Figure 52: 6^{th} -order, Elliptical NTF Feedforward Coefficients, Centre Freq. = 0.4	58
Figure 53: 8 th -order, Elliptical NTF Feedforward Coefficients, Centre Freq. = 0.1	50 59
Figure 54: 8^{th} -order, Elliptical NTF Feedforward Coefficients, Centre Freq. = 0.4	59 59
Figure 55: Butterworth Pole-Zero Man	60
Figure 56: SNR Plot 2^{nd} -order Both Topologies Butterworth NTE Centre Frequency = 0	375
$BW = 0.01 \ OSRSP - 32 \ OSR2P - 64 \ OSR4P - 128$.575, 61
$Figure 57$: SNR vs Centre Frequency Plot 2^{nd} order 2 nath CLEREE Topology Butterway	01 rth
NTE $DW=0.02$ OCD2D = 14	62
NIF, $BW = 0.02$, $USR2P = 10$ Eigura 58: BW Sween Plot 2^{nd} order 4 noth CP PEP Topology Putterworth NTE Control	02
Figure 58. BW Sweep Piol, 2 -order 4-path CK-KFB Topology, Butterworth NTF, Centre $= 0.2$, $OCDAD = 10$; ()
Fieq0.2, USR4P = 10	02
Figure 59. SNK Piols, Bour Topologies, Butterworth NTF, Different OSKs	03
Figure 60: DR VS BW Plot, Both Topologies, Butterworth NTF, BW=0.02, $OSR2P$, $4P =$	10
Eigura 61: SND Diat All Orders 4 noth Dutterworth CLEDEE Tanalagy Dutterworth NT	03 FE
Figure 61. SINK Piol, All Olders 4-path, Butterworth, CI-FBFF Topology, Butterworth NJ Centre Frequency = 0.1 $DW=0.02$ OCD4D = 16	іг, 64
Centre Frequency – 0.1, BW –0.02, $USR4P$ = 10	04 ITE
Figure 62. SINK Piol, 0 -order, All Patris, Butterworth, CK-KFB Topology, Butterworth N Centre Frequency = 0.1 $DW=0.02$ OCDCD = 64 OCD2D = 22 OCD4D = 16	нг, 65
Centre Frequency -0.1 , $BW-0.02$, $USRSP = 64$, $USR2P = 32$, $USR4P = 16$	03
Figure 63: Chebyshev Pole-Zero Map	00
Figure 64: Zoomed Output Plot, Chebyshev	0/
Figure 65: SNR Plot, 2 "-order Both Topologies, Chebysnev NTF, Centre Frequency = 0.4	+15,
BW = 0.01, OSRSP = 32, OSR2P = 64, OSR4P = 128	67
Figure 66: SNR Plot, 6 th -order, 4P, Both Topologies, Chebyshev NTF,	69
Figure 67: Output Plot, 6 th -order, SP, CI-FBFF Topology, Chebyshev NTF, Centre Freque	ency
= 0.25, BW = 0.01, PBA = IdB	69
Figure 68: Inverse-Chebyshev Pole-Zero Map	70
Figure 69: Zoomed Output Plot, Inverse-Chebyshev	71
Figure 70: SNR vs OSR Plot, Inverse-Chebyshev, 6^{m} -order, Centre Frequency = 0.3,	72
Figure 71: Elliptical Pole-Zero Map	73
Figure 72: Zoomed Output Plot, Elliptical	73
Figure 73: SNR Plot, All Filters, 4^{m} -order, 2P, CR-RFB Topology, BW = 0.01, OSR = 64	75
Figure 74: SNR Plot, All Filters, 4 th -order, 2P, CR-RFB Topology, BW = 0.04, OSR = 64	75
Figure 75: Simulink Model of Non-Ideal Integrator	77
Figure 76: Non-Ideal 2 ^{nu} -order SP CI-FBFF Topology	77
Figure 77: SNR vs Gain Error, 2 ^{na} -order CI-FBFF Topology, Elliptical NTF, Centre	
Frequency =0.1, BW=0.02, PBA= 1 dB, SBA= 80 dB, OSR=16	78

Figure 78: SNR vs Leakage Error, 2 nd -order CI-FBFF Topology, Butterworth NTF, Centre	
Frequency =0.1, BW=0.02, OSR=16	78
Figure 79: Simulink Model of Non-Ideal Resonator	80
Figure 80: Non-Ideal 2 nd -order SP CR-RFB Topology	80
Figure 81: Output Plots, 6 th -order SP CR-RFB Topology, Inverse Chebyshev Filter	. 81
Figure 82: Simulink Model of Non-Ideal Adder	. 81
Figure 83: Aliasing Tones, 6 th -order 4P CR-RFB Topology, Chebyshev Filter	83
Figure 84: Quantization Noise of a Sinusoid for a 1-bit Quantizer, A=0.9	84
Figure 85: Simulated Results for Quantization Noise of a Sinusoid, $vc = 0.12$. 87
Figure 86: Calculated Results for Quantization Noise of a Sinusoid, $vc = 0.12$. 87
Figure 87: Quantization Noise of a Sinusoid $vc = 0.37$. 88
Figure 88: Simulated Quantization Noise of a Sinusoid, $vc = 0.17$	89
Figure 89: Calculated Quantization Noise of a Sinusoid, $\nu c = 0.17$	89
Figure 90: Output Plot, SP CI-FBFF 4^{th} -order, Butterworth NTF, Centre Frequency = 0.11,	,
BW=0.02	. 91
Figure 91: Output Plot, SP CR-RFB 6^{th} -order, Elliptical NTF, Centre Frequency = 0.375,	
BW=0.04	. 91
Figure 92: Output Plot, 2P and SP CI-FBFF 4 th -order, Chebyshev NTF, Centre Frequency =	=
0.23, BW=0.02	94
Figure 93: Output Plot, SP and 8P CR-RFB 8 th -order, Inverse-Chebyshev NTF, Centre	
Frequency = 0.42, BW=0.002	94
Figure 94: D/A Converter Block Diagram	97
Figure 95: The OF Topology	98
Figure 96: The EF Topology	98
Figure 97: 4 th -order FBFF TDA Loop-filter	101
Figure 98: SP 4 th -order OF Topology	103
Figure 99: SP 4 th -order EF Topology	103
Figure 100: 2P 4 th -order EF Topology	104
Figure 101: 2P 4 th -order EF Topology	105
Figure 102: Zoomed Output Plots, D/A 2P 6 th -order Elliptical NTF, OF Topology, Centre	
Frequency = 0.2 , BW = 0.01 , SBA = 1 dB , PBA = 60 dB	107
Figure 103: Zoomed Output Plots, D/A 2P 6 th -order Elliptical NTF, EF Topology, Centre	
Frequency = 0.2 , BW = 0.01 , SBA = 1 dB , PBA = 60 dB	107
Figure 104: Finite Wordlength Input Signal and Its Quantization Tones, Freq.=0.27,	
Amplitude = 0.5, 15-bit Fractional Resolution	108
Figure 105: Feedback and Feedforward Coefficients versus Centre Frequency, Elliptical N	ΓF,
PBA=2dB, $SBA=60dB$	110
Figure 106: Feedback and Feedforward Coefficients versus Bandwidth Elliptical NTF, PBA	4=
2dB, $SBA = 60dB$	110
Figure 107: 2^{na} - Order SP OF Topology, Butterworth NTF, Centre Frequency = 0.35,	
BW=0.01	111
Figure 108: 4 th - Order 2P EF Topology, Chebyshev NTF, Centre Frequency = 0.35, BW=0	.01,
PBA=1 dB	111
Figure 109: 6^{m} - Order 4P 0F Topology, Inverse Chebyshev NTF, Centre Frequency = 0.1,	
BW=0.01, SBA=60 dB	112
Figure 110: SNR Plot, 6 th -order EF Topology	112

Figure 111: Experimental Set-up	114
Figure 112: LFSR Output Plot	114
Figure 113: SNR Plot, 4 th -order SP Σ - Δ modulators, Butterworth NTF, Centre Frequence	cv =
0.2,	115
Figure 114: SNR Plot versus Feedforward Coefficient Resolution, EF Topology, 4th-ord	er SP
Σ - Δ modulator Butterworth NTF, Centre Frequency = 0.2, BW = 0.02 OSR = 64	116
Figure 115: SNR Plot versus Feedforward Coefficient Resolution, OF Topology, 4 th -ord	ler SP
Σ - Δ modulator Butterworth NTF, Centre Frequency = 0.2, BW = 0.02 OSR = 64	116
Figure 116: SNR Plot versus Feedforward Coefficient Resolution, OF Topology, 4 th -ord	ler 2P
Σ - Δ Modulator, Chebyshev NTF, Centre Frequency = 0.2, BW = 0.02, PBA = 1 dB OSI	R = 64
	117
Figure 117: SNR Plot versus Feedforward Coefficient Resolution, OF Topology, 4th-ord	ler 4P
Σ - Δ Modulator, Inverse-Chebyshev NTF, Centre Frequency = 0.2, BW =0.02, SBA = 8	0 dB
OSR = 64	117
Figure 118: Pole Zero Map, Ideal Case, 4^{th} -order SP Σ - Δ modulator Chebyshev NTF,	118
Figure 119: Pole Zero Map for Various Resolution Values of Feedforward Coefficients	(Ks),
OF Topology, Chebyshev NTF, Centre Frequency = 0.2, BW =0.02, PBA = 1dB	118
Figure 120: Pole Zero Map for Various Resolution Values of Feedforward Coefficients	(Ks),
EF Topology, Chebyshev NTF, Centre Frequency = 0.2, BW =0.02, PBA = 1dB	119
Figure 121: The OF Topology, Butterworth Loop-Filter	120
Figure 122: The EF Topology, Chebyshev Loop-Filter	120
Figure 123:Output Plots of the SP topology for Inverse- Chebyshev Loop-Filter	122
Figure 124:Output Plots of the 2P topology for Inverse- Chebyshev Loop-Filter	122
Figure 125:Output Plots of the 4P topology for Inverse- Chebyshev Loop-Filter	123
Figure 126: 2^{nd} -order Tunable and Reconfigurable D/A VBP Σ - Δ modulator	125
Figure 127: Output Plots for Different Centre Frequencies, SP Topology is Activated	127
Figure 128: Output Plots for Different Centre Frequencies, 2P Topology is Activated	128
Figure 129: GUI Menu	131
Figure 130: Selecting the Conversion	132
Figure 131: Selecting the Specifications	132
Figure 132: Selecting the Specifications	133
Figure 133: Error Message	133
Figure 134: GUI Output Plot	134
Figure 135: GUI SNR Plot.	135
Figure 136: Average Noise Calculation Plots	139
Figure 137: Clearing The Input Signal	140
Figure 138: 2 nd -order, SP CI-FBFF Topology	150
Figure 139: 2 th -order, 2-path CI-FBFF Topology	150
Figure 140: 2 -order, 4-path CI-FBFF Topology	151
Figure 141: 2 -order, SP CI-FBFF Topology	152
Figure 142. 4 -older, 2-path CI-FDFF Topology	152
Figure 143. 4 -01001, 4-path CI-FDFF Topology	133
Figure 145: 6 th order 2 path CLEREE Topology	133
Figure 146: 6 th order A path CLEREE Topology	130
rigure 140. 0 -oluei, 4-paul Cl-r Dir Topology	137

Figure 147: 8 th -order, SP CI-FBFF Topology	159
Figure 148: 8 th -order, 2-path CI-FBFF Topology	160
Figure 149: 8 th -order, 4-path CI-FBFF Topology	161
Figure 150: 2 nd -order, SP CR-RFB Topology	162
Figure 151: 2 nd -order, 2-path CR-RFB Topology	162
Figure 152: 2 nd -order, 4-path CR-RFB Topology	163
Figure 153: 4 th -order, SP CR-RFB Topology	165
Figure 154: 4 th -order, 2-path CR-RFB Topology	165
Figure 155: 4 th -order, 4-path CR-RFB Topology	166
Figure 156: 6 th -order, SP CR-RFB Topology	168
Figure 157: 6 th -order, 2-path CR-RFB Topology	168
Figure 158: 6 th -order, 4-path CR-RFB Topology	169
Figure 159: 8 th -order, SP CR-RFB Topology	171
Figure 160: 8 th -order, 2-path CR-RFB Topology	172
Figure 161: 8 th -order, 4-path CR-RFB Topology	172
Figure 162: 2 nd -order, SP EF Topology	173
Figure 163: 2 nd -order, 2-path EF Topology	174
Figure 164: 2 nd -order, 4-path EF Topology	175
Figure 165: 4 th -order, SP EF Topology	176
Figure 166: 4 th -order, 2-path EF Topology	177
Figure 167: 4 th -order, 4-path EF Topology	178
Figure 168: 6 th -order, SP EF Topology	179
Figure 169: 6 th -order, 2-path EF Topology	180
Figure 170: 8 th -order, 4-path EF Topology	181
Figure 171: 2 nd -order, SP OF Topology	182
Figure 172: 2 nd -order, 2-path OF Topology	183
Figure 173: 2 nd -order, 4-path OF Topology	184
Figure 174: 4 th -order, SP OF Topology	185
Figure 175: 4 th -order, 2-path OF Topology	186
Figure 176: 4 th -order, 4-path OF Topology	187
Figure 177: 6 th -order, SP OF Topology	189
Figure 178: 6 th -order, 2-path OF Topology	190
Figure 179: 6 th -order, 4-path OF Topology	191

ABBREVIATIONS

The following project-specific and general technical abbreviations are used in the thesis:

AAF	Anti Aliasing Filter
ADC	Analog to Digital Converter
AWGN	Additive White Gaussian Noise
A/D	Analog to Digital
BS	Band Stop
BP	Bandpass
BPF	Bandpass Filter
BW	Bandwidth
CCD	Charged Coupled Device
CF	Centre Frequency
СТ	Continuous Time
DAC	Digital to Analog Converter
DECT	Digital Enhanced Cordless Telecommunications
DEM	Dynamic Element Matching
D/A	Digital to Analog
ENOB	Effective Number of Bits
EPS	Encapsulated Postscript
FFT	Fast Fourier Transform
GNSS	Global Navigation Satellite System
GSM	Global System for Mobile Communication
GUI	Graphical User Interface
FSM	Finite State Machine
HF	High Frequency
IC	Integrated Circuit
IF	Intermediate Frequency

- **IIR** Infinite Impulse Response
- **IMD** Intermodulation Distortion
- LFSR Linear Feedback Shift Register
- LNA Low Noise Amplifier
- LP Lowpass
- LPF Lowpass Filter
- **NTF** Noise Transfer Function
- **OSR** Oversampling Ratio
- **PBA** Passband Attenuation
- PCM Pulse Code Modulation
- **PSD** Power Spectral Density
- **RF** Radio Frequency
- **SBA** Stopband Attenuation
- SC Switched Capacitor
- **SDR** Software Defined Radio
- SINAD Signal-to-Noise and Distortion Ratio
- **SNR** Signal to Noise Ratio
- SP Single Path
- **STF** Signal Transfer Function
- **TI** Time Interleaved
- **VBP** Variable Bandpass
- **2P** Two Path
- **4P** Four Path
- **Σ-** Δ Sigma-Delta

I, Işıl Kalafat Kızılkaya, declare that all the material contained in this thesis is my own work.

Chapter 1. Introduction

Pervasive technology emerges in people's daily life requiring multi-mode multi-channel circuitry. Therefore, data converters need to be highly integrated into the overall system due to low-power, small-size requirements [1-3]. Moreover, innovative technologies such as Global Navigation Satellite System (GNSS), Software Defined Radio (SDR), Global System for Mobile Communication (GSM), Digital Enhanced Cordless Telecommunications (DECT) require wideband, high-speed data conversion [4-6].

Sigma-Delta (Σ - Δ) modulators offer more benefits compared to Nyquist-rate converters such as high-speed data conversion whilst providing decreased hardware. They are able high resolution even with a one-bit quantizer, since they employ oversampling and noise-shaping. Although using fewer bits in the quantizer inevitably causes higher quantization noise, it also removes the Dynamic Element Matching (DEM) circuitry and the noise arising from it [7, 8].

The basic idea of Σ - Δ modulation is to shape the quantization noise using a loop-filter such that the noise in the band of interest is suppressed, whilst the out-of-band noise is amplified. In addition, Σ - Δ modulators utilize high OverSampling Ratios (OSR)s to decrease the overall noise from DC to Nyquist [7, 8]. However, these benefits of Σ - Δ modulation have their own limitations.

First of all, the loop-filter may cause instability [9]. The type and order of the loop-filter should be chosen carefully after iterative simulations [7]. Secondly, using a high OSR restricts the bandwidth of the conversion therefore limiting the Σ - Δ modulators mostly to narrow-band applications [10]. In Chapter 2, these problems will be defined and discussed with their positive and negative aspects.

1.1 Integration, Loop-filter and Stability

As mentioned, a high OSR limits the conversion bandwidth of the Σ - Δ modulators thus there has been a great amount of work and analysis on LowPass (LP) Σ - Δ modulators. On the one hand, the designed, implemented and mathematically analysed LP Σ - Δ modulators makes it easy to build stable, high-resolution LP Σ - Δ modulators. However, the integration of LP Σ - Δ modulators with high-frequency technologies requires complex building blocks such as mixers and passive filters [11-13]. These extra blocks not only make the structure complex and increase the overall power dissipation but also introduce extra noise and limit linearity of these modulators.

Herein, BandPass (BP) Σ - Δ modulators that are capable of feedforwarding the signal through the data converter are really advantageous as they eliminate the downsampling process [12, 13]. However, the majority of the published papers on BP Σ - Δ modulators are confined to midband ($f_s/4$) [13-15]. In addition, resonator based Σ - Δ modulators result in unequal out-ofband gain levels of the Noise Transfer Function (NTF) especially for frequencies close to DC and Nyquist. Unequal shoulder gains may cause modulator instability and/or ear-detectable noise. Moreover, if channel selection is required within the system, tunable centre-frequency and/or bandwidth are needed [15].

As a result, Variable BandPass (VBP) Σ - Δ modulators are highly suitable candidates to employ for high-speed and high frequency data conversion applications. It should be pointed out that VBP terminology is used here when referring to the variable centre-frequency and variable bandwidth of the Σ - Δ modulator. There are a few published papers on multi-band Σ - Δ modulators but they are restricted to other fixed centre-frequencies [16-18]. As a solution, this study presents generalized VBP Σ - Δ modulators that can accomplish noise-shaping for any centre frequency and bandwidth from DC to Nyquist. Moreover with the help of various filters such as Butterworth, Chebyshev, Inverse Chebyshev and Elliptical, designers have the flexibility to specify the centre frequency, bandwidth and passband/stopband attenuations. With the help of these filters, stable VBP Σ - Δ modulators, the designed generalized VBP Σ - Δ modulators employ more feedback and feedforward coefficients. Furthermore, the complexity of these coefficients requires more hardware when implemented in Digital to Analog (D/A) converters. However, these VBP Σ - Δ modulators have superior ability to shape the quantization noise in the frequency of interest within the desired bandwidth.

1.2 High OSR, Speed and Bandwidth

The high sampling frequency places huge pressure on processor speed, thus constraining discrete-time Σ - Δ modulators to narrow-band signal applications. In order to overcome the

high-speed requirements, researchers have focused on several parallel structures such as frequency-band-decomposition, cascaded Hadamard, and Time-Interleaving (TI) based Σ - Δ modulator topologies [19]. The TI approach employs *P* mutually cross-coupled Σ - Δ modulators, each operating at a sampling frequency f_s , which results in an equivalent sampling rate of $P f_s$. Moreover, this approach offers an elegant means to increase the conversion bandwidth circumventing the need to use faster or higher-order Σ - Δ modulators [20]. Using the TI technique requires increased hardware and design complexity, but provides a powerful means to increase the conversion band. In this study, the TI method is utilized and applied to the designed Single Path (SP) VBP Σ - Δ modulators. The designed TI VBP Σ - Δ modulators will be discussed in Chapter 3 and Chapter 4.

To sum up, generalized VBP Σ - Δ modulators are designed and evaluated in this study to provide flexibility. By providing flexibility, these generalized VBP Σ - Δ modulators eliminate the mixer and passive filtering blocks of conventional receivers. Hence they overcome the integration requirements of data converters within the high frequency and high-speed technology as will be explained in Chapter 3. Moreover, these generalized VBP Σ - Δ modulator topologies help to determine those frequencies that are capable of obtaining high SNRs and/or less hardware and/or less tonality. The stability and performance of these VBP Σ - Δ modulators are enhanced with the help of different filters; Butterworth, Chebyshev, Inverse Chebyshev and Elliptical filters. However, these generalized VBP Σ - Δ modulators still suffer from relatively narrow bandwidths as a result of the high OSRs. Therefore, the TI methodology is applied to the designed VBP Σ - Δ modulators. Whilst applying the TI technique, the positive and negative attributes of these resulting multi-path Σ - Δ modulators will be investigated and presented.

1.3 Outline

In *Chapter 2*, the basics of data conversion are explained. The commonly used blocks are explained and their mathematical models are provided. Thereafter, the theory of Σ - Δ modulators as well as their characteristics are presented. Hence the motivations for the design of TI VBP Σ - Δ modulators are explained. Finally the contributions and novelties of the thesis are listed.

In *Chapter 3*, a step-by-step design technique is developed for TI VBP Σ - Δ modulators. This technique is an extension of the node-equation method that is also explained in Chapter 3.

A generalized NTF that can perform Butterworth, Chebyshev, Inverse-Chebyshev and Elliptical filters, is chosen to build generalized Analog-to-Digital (A/D) VBP Σ - Δ modulators. Two discrete loop-filters of 2nd-, 4th-, 6th- and 8th-orders are designed by mapping the given generalized NTF. Thereafter the developed design technique is applied to these SP VBP Σ - Δ modulators in order to construct their 2-path and 4-path counterparts. The resulting topologies are analysed and compared in terms of their coefficients, Signal-to-Noise Ratio (SNR), Dynamic Range (DR) and immunity to non-idealities.

In *Chapter 4*, D/A TI VBP Σ - Δ modulators are designed using the design technique in Chapter 3. The Error Feedback (EF) and the Output Feedback (OF) topologies are built employing the same loop-filter of 2nd-, 4th- and 6th-orders. Once again, the EF and OF topologies are analysed and compared in terms of their coefficients, SNRs, DRs and immunity to non-idealities. Thereafter, they are implemented in VHDL and synthesized on the Xilinx® SpartanTM-3 Development Kit. These circuits not only validate the developed design technique but also ensure the accuracy of the simulation routines built spesifically for these modulators.

In *Chapter 5*, the developed Graphical User Interface (GUI) is presented. This GUI wraps up all the designed A/D and D/A simulink models and Matlab routines in a single environement. It enables the user to plot the output response or the SNR response of any chosen conversion type, topology, order and path number with the desired NTF provided in Chapter 3.

In *Chapter 6*, the concluding remarks of the overall thesis are reviewed. The recommendations for future work are also discussed.

Chapter 2. Data Converters

Analog circuitry unavoidably suffers from non-ideality issues such as manufacturer errors, thermal noise and component mismatches [21]. Apart from a few non-idealities such as clock jittering, finite wordlength and propagation delays, circuit performance is more settled in the digital world. As long as the values of 1s and 0s do not change their states, the non-idealities in the digital world do not affect the overall circuit performance. Moreover, data storage in the digital domain is much more effective and easy. Thus, data conversion plays a vital role in a wide range of implementations from biomedical devices to communication technologies, as the real world is completely analog.

Data conversion is a dual-way street: A/D and D/A. In Figure 1, the data conversion block diagram is given. As seen, an analog signal is filtered, sampled and then converted to a digital word stream. After the processing of the digital signal, a D/A conversion block and an analog filter reconstruct the processed signal.





This section firstly investigates the sampling and the quantization, as these are the vital blocks of data conversion in which the fundamental amount of noise is added to the signal. By simply utilizing these two blocks, one can build a conventional Pulse Code Modulation (PCM) data converter. It is important to go through the basics of PCM to have a better understanding of Σ - Δ modulation and how it achieves higher resolution in combination with simpler hardware.

Thereafter the principles of Σ - Δ modulation are explained. The limitations of its ability to fulfil the emerging technologies' requirements are identified and addressed. A comprehensive literature survey is also carried out to identify the state-of-the-art and any areas that are worthy

of further investigation in the field of Σ - Δ modulation. As a solution, the Time-Interleaving method and VBP Σ - Δ modulators are presented. Finally, performance parameters are defined in order to compare and evaluate the resolution of the designed Σ - Δ modulator topologies.

2.1 Sampling and Pre-Filtering

Sampling is the conversion of a continuous time analog signal x(t), into a discrete time analog signal x[n]. The sampling frequency f_s should be at least twice the signal bandwidth to meet the Nyquist criteria [22]. This is due to the fact that sampling results in signal images at the frequencies kf_s as given in (1), where T_s is the sampling interval. These images are also depicted in Figure 2 where f_B is the signal band.



Figure 2: Sampled Signal Spectrum

However, in real life, signals are not band-limited and may have high frequency tones, which will result in aliasing after sampling. Therefore, pre-filtering is always required before the sampling process to ensure a band-limited signal. The pre-filter is called the Anti-Aliasing Filter (AAF). As seen in Figure 3, the higher the f_s is chosen, the more relaxed an AAF is needed. Moreover, a high sampling frequency will result in higher resolution both for the A/D and D/A converters.



Figure 3: AAF and Aliasing

2.2 Quantization

The second step of signal processing is the A/D conversion in which sampled and bandlimited data is converted to digital data. The conventional Analog to Digital Converter (ADC) can be constructed by using a quantizer after the sampling block as seen in Figure 4. Since the quantizer itself maps the sampled data to a digital quantization level, the output is purely digital. However, this ADC scheme, in other words PCM technique, demands multi-bit quantizers to achieve high resolution. Thereby it increases the overall hardware complexity [23].



Figure 4: PCM A/D Converter

At this point it is rational to roughly inspect the quantization before getting into the Σ - Δ modulation technique. Hence the A/D and D/A conversion blocks are investigated in detail in Chapter 3 and Chapter 4.

Quantization is the process where discrete amplitude data are converted into a digital data stream. The basic concept is illustrated in Figure 5. Note that uniform sampling is employed within this study where the quantization steps, " Δ_Q "s, are taken to be equal.



Figure 5: Quantization of a Sampled Analog Signal

The sampled input data x[n] is rolled over to the nearest quantization level resulting in a discrete time and discrete amplitude data stream. This rolling process inevitably causes an error at the output, known as the quantization noise.

As depicted in Figure 5, quantization is a highly non-linear operation which is mathematically hard to model. The commonest approximation to define its behaviour is the Additive White Gaussian Noise (AWGN) approach. As seen in Figure 6, the AWGN equates the output of a quantizer, y[n], to the sum of the input signal, x[n], and the quantization noise, e[n]. Moreover, the probability density function of the quantization noise, $P_e(e)$, is assumed to have a uniform distribution between $\pm \Delta_Q/2$ as illustrated in Figure 7.



Figure 6: AWGN Block Diagram



Figure 7: White Quantization Noise

In (2), the power of the white quantization noise is calculated where σ_e is the quantization noise and Δ_Q is the quantization step. Assuming an N-bit quantizer with a maximum output level of *V*, the quantization step can be calculated as $2V/2^N$. So (2) is recalculated in (3) resulting in the SNR formula given in (4) where σ_s^2 is the input signal power. Note that for every *N* increment there is a corresponding 6 dB SNR increment. Equation (4) clearly depicts this for high resolution where a higher number of bits in the quantizer is required.

$$\sigma_e^2 = \int_{-\Delta_Q/2}^{\Delta_Q/2} \frac{1}{\Delta_Q} e^2 de = \frac{\Delta_Q^2}{12}$$
(2)

$$\sigma_e^2 = \frac{\left(\frac{2V}{2^N}\right)^2}{12} \tag{3}$$

$$SNR = 10 \log \frac{\sigma_s^2}{\sigma_e^2} = 10 \log \frac{\sigma_s^2}{V^2} + 4.77 + 6.02N \, dB \tag{4}$$

The SNR calculation in (4) is for an ideal PCM ADC. In real-life implementations, there occurs input referred noise such as thermal noise and flicker noise. Hence the obtained SNR is lower than the calculated SNR. Therefore, designers generally calculate the Effective Number Of Bits (ENOB) as a performance metric. The ENOB measures the actual resolution in bits of an operating ADC. The IEEE standard 1241-2000 defines the ENOB as a measure of the signal-to-noise and distortion ratio (SINAD) used to compare actual ADC performance to an ideal ADC [24]. The ENOB formula is given by the equation in (5) and the equivalent equation to calculate SINAD is given in (6). Note that the IEEE standard 1241-2000 calculates the ENOB assuming a sine wave input to the ADC. Therefore, *A* refers to the amplitude of the sine wave and *V* represents the full-scale range of the ADC/DAC under test.

$$ENOB = \log_2(SINAD) - \frac{1}{2}\log_2(1.5) - \log_2\left(\frac{A}{V/2}\right)$$
(5)

$$SINAD = \sqrt{1.5} \left(\frac{A}{V/2}\right) 2^{ENOB} \tag{6}$$

On the other hand, (4) assumes that the entire band from DC to the sampling frequency is required for the conversion. However, the conversion band is generally limited to the input signal band. If the in-band quantization noise is reduced, the SNR will increase. A common

way to decrease the in-band noise is to use oversampling. The idea of oversampling is that the uncorrelated quantization noise is averaged by a higher number of samples. As a result, the Power Spectral Density (PSD) of the quantization noise is distributed over a larger frequency band decreasing the in-band quantization noise as seen in Figure 8. Assuming a baseband signal, the reduced quantization noise by the use of oversampling can be further decreased with the help of a Low Pass Filter (LPF).



Figure 8: Oversampling and Quantization Noise

Consequently, oversampled converters can achieve the desired SNR with a fewer number of quantization bits. The attractiveness of fewer quantization bits is the drastically reduced Digital to Analog Converter (DAC) circuitry utilized in the feedback loop of the A/D converters [25]. Specifically, if a one-bit quantizer is employed, the resulting DAC circuitry is just a comparator. A one-bit quantizer not only reduces the DAC circuitry but also eliminates the DEM circuitry that is required to cancel the non-ideal behaviour of the DAC.

By the same token, D/A converters suffer from the problem of having a multi-bit input. As the digital data coming through the DAC is multi-bit, the resulting circuitry is complex [25]. Hence, modulation is required before the D/A conversion in which the multi-bit input data is converted to a coarse, lower-bit data in order to reduce the DAC circuitry and mitigate its non-ideal behaviour.

2.3 Σ - Δ Modulation

 Σ - Δ modulators offer high resolution and relatively simpler hardware therefore they are widely chosen in many applications for data conversion. When they were first introduced in the market in the 1960s, they were mostly used for audio implementations due to their high OSR requirements [7], [8], [26]. However, as technologies developed and processors achieved higher speeds, Intermediate Frequency (IF) applications, such as in frequency synthesizers [27], switched-mode power supplies [28] and software defined radios [29]; also used the Σ - Δ modulation technique. The fundamental concept underpinning Σ - Δ modulation is to combine noise-shaping and oversampling to achieve high resolution therefore decreasing the number of bits in the quantizer. This in return significantly reduces the hardware required for the DAC and relaxes the steepness of the AAF behaviour.

 Σ - Δ modulators utilize a feedback loop and a loop-filter to shape the quantization noise arising from the quantizer itself. The basic structure of a Σ - Δ modulator for an A/D conversion is seen in Figure 9. A decimator is placed after the Σ - Δ modulator to filter the output data and downconvert it to the Nyquist rate. Note that for the D/A modulator the DAC in the feedback loop is removed, instead a DAC is placed after the decimator as seen in Figure 10.

To differentiate between A/D and D/A Σ - Δ modulation, N-bit and M-bit quantizers are put within the block diagrams respectively. Once again, an A/D Σ - Δ modulator converts a sampled input data to an N-bit digital data stream whilst a D/A Σ - Δ modulator converts an N-bit digital data to a coarse M-bit digital data where M<N. The D/A conversion is performed by the M-bit DAC placed after the decimator therefore saving both hardware and power consumption. This study utilizes 1-bit quantizers for both topologies resulting in a comparator for the DAC topology and eliminating the DEM circuitry required to amend the non-ideal behaviour of the DAC.







Figure 10: D/A Σ - Δ Modulator

2.3.1 Noise-Shaping

Noise-shaping is the term used for quantization noise filtering. This is where the in-band noise is reduced using a suitable NTF and the suppressed quantization noise is pushed to the out-of-band region as shown in Figure 11.



Figure 11: Noise Shaping and Oversampling

If the AWGN is utilized to mathematically model the quantizer behaviour as seen in Figure 12, the resulting equation is obtained in (7). X(z) is the input to the Σ - Δ modulator whilst E(z) represents the quantization noise. The Signal Transfer Function (STF) shapes the input signal, whereas the NTF shapes the quantization noise. The NTF is the most important part of the design process, as it has to ensure Σ - Δ modulator stability, minimise tonality and at the same time deliver the required SNR and DR.



Figure 12: AWGN Model in Σ - Δ Modulator

$$Y(z) = X(z)\frac{H(z)}{1+H(z)} + E(z)\frac{1}{1+H(z)}$$
(7)

$$STF(z) = \frac{H(z)}{1+H(z)}$$
(8)

$$NTF(z) = \frac{1}{1+H(z)} \tag{9}$$

2.3.2 Oversampling

In addition to the noise-shaping, Σ - Δ modulators employ high OSR to spread the quantization noise over a wider frequency band as well as moving the sampling images further apart, thereby relaxing the AAF [30-32]. This was shown in Figure 3. As known from filter theory, a relaxed AAF can be implemented using simpler hardware. So, higher OSRs not only result in higher SNRs but also reduce the overall hardware. The OSR is defined in (10) where f_s is the sampling frequency and f_B is the bandwidth of the signal. After all, a high OSR is a trade-off between bandwidth and processor speed.

$$OSR = \frac{f_s}{2f_B} \tag{10}$$

Consequently, with the help of noise-shaping and oversampling, Σ - Δ modulators achieve high resolution in conjunction with a reduced number of quantizer bits. The reduction in the quantizer bits and the relaxed AAF topology simplify the overall hardware providing lower power consumption. By using a single-bit quantizer, a Σ - Δ modulator can accomplish resolution as high as the equivalence of 20-bits as reported in [33], [34].

2.4 BP Σ - Δ **Modulation**

The majority of published work presents the Low-Pass (LP) Σ - Δ modulators. Obviously this is due to the baseband applications that extensively use the Σ - Δ modulation technique. These baseband applications, such as in audio implementations, do not suffer from the high-speed requirements of the Σ - Δ modulators [35], [36]. As a result, there is already a huge amount of know-how for the LP Σ - Δ modulators in terms of design, system-level modelling, simulations and implementation.

Therefore, the basic design method of a bandpass Σ - Δ modulator is to convert a LP Σ - Δ modulator prototype to its bandpass counterpart. A basic BP Σ - Δ modulator can be built by replacing the integrator blocks of the already designed LP Σ - Δ modulator with resonator blocks. This assumes a delay-based integrator whose transfer function is given in (11). Basically by transforming the z^{-1} to $-z^{-2}$, one can obtain a mid-band resonator as shown in (12).

$$H_{ID}(z) = \frac{z^{-1}}{1 - z^{-1}} \tag{11}$$

$$H_{RMid}(z) = \frac{-z^{-2}}{1+z^{-2}}$$
(12)

This easy conversion results in mid-band Σ - Δ modulators offering noise-shaping at the normalized frequency of $\nu = 0.25$. There are two main benefits of this conventional design method; firstly, the new BP Σ - Δ modulator preserves the performance metrics of the formerly designed LP Σ - Δ modulator that is the SNR, DR and the stability. Secondly, the mid-band resonator blocks are easy to implement both for Switched Capacitor (SC) circuits and digital circuits [7, 8]. However, these mid-band Σ - Δ modulators limit the flexibility of the overall system due to the fixed centre-frequency and bandwidth. Although the centre-frequency can be changed by appropriate feedback coefficients as shown in Figure 13 and calculated in (13), the bandwidth limitation still exists. More importantly, centre-frequencies close to DC or Nyquist result in unequal out-of-band gain levels of the NTF as depicted in Figure 14 and may result in instability especially for Σ - Δ modulators above second-order. It is worth pointing that 0 dB corresponds to an amplitude value of 1 for all the output frequency response plots in this thesis.



Figure 13: Delayed Resonator Block

$$H_R(z) = \frac{z^{-1}}{1 + (2 - \beta)z^{-1} + z^{-2}}$$
(13)



Figure 14: Resonator-Based Second Order (Effective First Order) Σ-Δ modulator's Output Plots for Different Centre Frequencies

2.5 Motivations for the TI VBP Σ - Δ Modulators

The traditional flash ADCs that directly convert the incoming signal through the comparators offer the fastest conversion technique. State-of-the-art flash ADCs are capable of operating up to 24 GHz of sampling frequency for 4-6 bits of resolution making them appropriate for high-speed and low-resolution technologies such as radar detection [37-42]. Other ADCs such as pipeline [43], [44] or successive approximation [45], [46] can provide higher resolution between 8-16 bits. Their conversion band, however, is limited to a few hundreds of MHz making them unsuitable for Radio Frequency (RF) signal conversion. But these converters are viable in moderate speed applications such as ultrasonic medical imaging, industrial controls and Charged Coupled Device (CCD) imaging. Subsampling may be introduced to such systems in order to overcome the speed limitation at the expense of performance degradation caused by the noise aliasing and the required high-quality BP AAFs [47].

The conventional LP Σ - Δ modulators are mostly employed by audio implementations where the bandwidth is around 20-30 kHz [48-50]. Thus high SNRs can be obtained without sacrificing much effort. On the other hand, there are reported implementations of up to 20 GHz of sampling frequency with a 1 GHz bandwidth range, where Continuous-Time (CT) Σ - Δ modulators are implemented by SiGe BiCMOS technology [51]. SiGe BiCMOS technology might be promising for the future but at the moment SiGe wafers are extremely expensive to use in mass-production. In [52], a CT BP Σ - Δ modulator is built operating at 3.2 GHz of sampling frequency with a 1 GHz of bandwidth and it is implemented in CMOS technology. It should be noted that CT Σ - Δ modulators have the superior performance to operate at higher frequency but they are limited to a set of sampling frequencies [8]. Besides all the aforementioned conversion methods including the LP and mid-band Σ - Δ modulators, configurable bandwidth and centre-frequency BP Σ - Δ modulators are the best candidates to deploy in digitally intensive RF architectures for the emerging sub-micron technologies. These modulators provide high-speed and high-resolution conversion whilst maintaining the advantages of lower hardware complexity and cost.

In Figure 15, the aforementioned ADCs are compared in a single diagram in terms of their resolution and conversion bandwidth. Apparently, Σ - Δ modulators are attractive due to their high-resolution. However, the conversion bandwidth of these modulators limits their use to the narrow-band applications.



Figure 15: Comparison of ADCs

To clarify the motivation of this study, generalized VBP Σ - Δ modulators are designed to evaluate the entire frequency-band from DC to Nyquist with an adjustable bandwidth therefore offering greater flexibility for the design of BP Σ - Δ modulators. By supporting flexible topologies for the signal conditioning and the channelization, these generalized VBP Σ - Δ modulators make it possible to implement reconfigurable, multi-standard signal conversion. In addition, to achieve better resolution and/or to improve the integration of the system, the designer may fix these specifications to the desired values. For instance; in [53] a GSM receiver Integrated Circuit (IC) is designed. However, to allow use of the IC in a broad range of narrow-band applications, including analog-FM, GSM/EDGE, the centre-frequency of the Σ - Δ modulator is fixed to $f_{clk}/8$. Therefore, the overall system's universal clocks can be switched to choose the desired conversion frequency.

Despite the flexibility and integration properties, designing a VBP Σ - Δ modulator has its own challenges. Primarily, bandwidth restriction is still an issue. High OSR requirements still apply limiting the modulator to narrowband applications. The proposed technique to ameliorate this situation is to utilize parallel Σ - Δ modulators to offer higher speed operation or increased conversion band. Specifically the chosen parallel structure is called the Time-Interleaved (TI) topology and is discussed in Section 3.3.2

Table 1 presents a list of Σ - Δ modulators, which compares their implementation performance in order to have a better understanding of the limitations of Σ - Δ modulation. Note that the the reference numbers in Table 1 are listed as presented in the References's chapter on page 192. As listed, with the help of the TI technique, designers managed to increase the overall sampling frequency therefore the required SNR values can be obtained by using lower-order Σ - Δ modulators. When [54-56] and [59] are compared, [59] increases the signal bandwidth with a lower sampling frequency by using an 8-path TI Σ - Δ modulator topology. On the other hand, although the orders of the Σ - Δ modulators in [55], [57] and [58] are the same, it is shown that the Σ - Δ modulator in [58] produces much higher SNR (i.e. 33 dBs and 36 dBs respectively) using lower sampling frequency. This superior SNR performance is attributed to its 2-path TI structure.

An extensive literature survey has revealed that all reported publications on TI topologies have been confined to LP and midband resonator-based Σ - Δ modulators as the recursive operation of Σ - Δ modulators complicate the TI conversion [60], [61]. Moreover, a great number of the architectures in the literature propose 2-path and low-order TI structures derived using the polyphase decomposition method [62-64]. This is due to the fact that polyphase decomposition of high order NTFs particularly for a higher number of interleaving paths is quite cumbersome to calculate [65]. Thereby the first aim of this study is to further develop the method proposed in [20] to enable the design of multi-path VBP Σ - Δ modulators. The utilized and developed method is referred to as the Node Equation technique and is examined in Section 3.3.5.

Signal Band	Sampling Frequency	OSR	Modulator Structure	SNR	Reference
10 MHz	480 MHz	24	CT 3 rd -order A/D	70 dB	[54]
100 MHz	1 GHz	5	CT 2 nd -order A/D	43 dB	[55]
500 kHz	500 MHz	500	DT 1 st -order A/D	60 dB	[56]
10 MHz	50 MHz	2.5	DT 2 nd -order A/D	46 dB	[57]
1.25 MHz	50 MHz	50	DT 2-path 2 nd -order TI D/A	86 dB	[58]
15.625 MHz	250 MHz	64	CT 8-path 1 st -order TI A/D	45 dB	[59]
1.25 MHz	30 MHz	50	DT 5-path 2 nd -order TI A/D	48 dB	[60]
80 MHz	2.2 GHz	27.5	CT 2-path 1 st -order TI A/D	50 dB	[61]

Table 1: Comparison of Σ - Δ Modulators in Use

This thesis presents novel generalized TI VBP Σ - Δ modulators where designers are able to specify the centre-frequency, bandwidth, pass-band and stop-band attenuations. The flexibility of these modulators enables designers to integrate them with the overall system whilst providing larger bandwidth.

A step-by-step design technique is provided and performed for various A/D and D/A TI VBPbased Σ - Δ modulator topologies. This technique initially starts with developing SP VBP Σ - Δ modulator topologies and verifying their stability, tonality, SNRs and DRs by extensive behavioural-level simulations. The process commences with the design of the NTF, which is then mapped to an appropriate topology. Before selecting a suitable topology, the NTF specifications should be determined to ensure stability, adequate SNR and DR. In this sense, employing different filtering types where the designer has the control over the centrefrequency and the bandwidth is indispensible. The NTF design in this study circumvents the imbalance in the shoulder gain levels by making use of Butterworth, Chebyshev, Inverse-Chebyshev and Elliptical narrow-band filters, thus delivering more stable Σ - Δ modulators. The second step is to convert these SP VBP Σ - Δ modulators to their P-path TI counterparts. Thereafter, all the designed topologies are investigated including their performance and immunity to non-idealities. The node-equation technique presented for LP Σ - Δ modulators in [20] and further developed for VBP Σ - Δ modulators is detailed in Section 3.2. Finally, Matlab routines in conjunction with SIMULINK models were wrapped in a GUI environment to automate and speed-up the calculations.

Contributions and Novelties

The developed and novel contributions of this PhD study are listed in bullet points. The developed contributions' bullet points start with a 'D' whereas the novel contributions' bullet points start with an 'N'.

- D: A step-by-step design technique for generalized TI VBP Σ-Δ modulators is developed. This technique is an extension of the node-equation method. The node equation method is applied to a generalized NTF, and resulting generalized SP topologies are converted to their TI counterparts by using the node-equation technique.
- N: Using the developed design method, over 36 SP and multi-path Σ-Δ modulators are designed, analysed and evaluated. These generalized VBP Σ-Δ modulators are capable to perform noise-shaping at a desired centre frequency with the chosen bandwidth, passband and stopband attenuations.
- D: Simulation routines calculating the filter coefficients and overall performance of TI VBP Σ-Δ modulators are developed in MATLAB and evaluated with SIMULINK models.
- **D:** The proposed VBP Σ - Δ modulators are evaluated for many different centre frequencies and bandwidths.
- D: 6th- and higher-order CI-FBFF Σ-Δ modulator topologies whose normalised centrefrequencies lie within [0.25 0.5] are shown to be unsuitable for implementation. This is attributed to their large feedback and feedforward coefficients.

- D: A/D TI VBP Σ-Δ modulators' non-idealities are extensively evaluated at the behavioural level.
- D: A/D and D/A VBP Σ-Δ modulators that use Butterworth, Chebyshev, Inverse-Chebyshev and Elliptical filters are compared and evaluated in terms of SNRs, DRs, tonality and hardware complexity.
- N: Significant reductions to the number of coefficients and internal connections are made for the OF D/A TI VBP Σ-Δ modulators.
- N: The tonal behaviour of quantizers when excited by single sinusoidal signal is mathematically modelled and verified both for A/D and D/A VBP Σ-Δ modulators. The model is called the sawthooth quantization noise model.
- N: The sawtooth quantization noise model is extended for the TI topologies and verified both for TI A/D and D/A VBP Σ - Δ modulators.
- N: 4th-order, SP, 2-path and 4-path D/A VBP Σ-Δ modulators are implemented in VHDL and synthesized on the Xilinx® SpartanTM-3 Development Kit.
- N: To further enhance the flexibility of the TI VBP Σ-Δ modulators, a 2nd-order reconfigurable and tunable D/A TI VBP Σ-Δ modulator topology is built and implemented.
- N: The SIMULINK models and MATLAB codes are combined in a GUI.

2.6 **Performance Parameters**

The main performance parameters that define a Σ - Δ modulator's overall behaviour are the SNR, DR, stability and tonality.

2.6.1 Signal to Noise Ratio

The Signal-to-Noise Ratio (SNR) of a Σ - Δ modulator is the ratio of the signal power to the inband noise power. On page 10, the SNR is defined and calculated in (4) for a PCM converter. The SNR of a Σ - Δ modulator differs from the SNR of a PCM converter in two aspects. Firstly, the in-band noise that is shaped by the NTF should be taken into account for the SNR calculations. Secondly, the noise at the output of the modulator is not only composed of the quantization noise. There are some other noise sources in real life such as thermal noise and flicker noise that need to be considered. However, it is hard to approximate these noise
sources before designing the circuitry. Therefore in (16), equation (14) is recalled where the noise-shaping and bandwidth are taken into the equation in order to have a projection of the expected SNR. σ_n^2 is used to define the shaped noise power whereas the quantization noise power is defined by σ_e^2 . $S_q(f)$ is the PSD of the non-shaped quantization noise and is equal to $S_q(f) = \sigma_e^2/f_s$. f_{c1} and f_{c2} are the cut-off frequencies of the designed band-pass filter. Note that for a lowpass Σ - Δ modulator f_{c1} equals to $-f_{c2}$. By making proper alterations in (15), (22) is calculated.

$$SNR = 10 \log \left(\frac{\sigma_s^2}{\sigma_e^2}\right) \tag{14}$$

$$\sigma_n^2 = \int_{f_{c1}}^{f_{c2}} S_q(f) |NTF(f)|^2 df$$
(15)

$$SNR = 10 \log\left(\frac{\sigma_s^2}{\sigma_n^2}\right) = 10 \log\frac{\sigma_s^2}{\int_{f_{c1}}^{f_{c2}} \sigma_e^2 / f_s \left| NTF(f) \right|^2 df}$$
(16)

$$SNR = 10\log\left(\sigma_s^2\right) - 10\log\left(\sigma_e^2 \frac{f_B}{f_s} \int_{f_{c1}}^{f_{c2}} \left|NTF(f)\right|^2 df\right)$$
(17)

$$SNR = 10 \log(\sigma_s^2) - 10 \log(\sigma_e^2) - 10 \log\left(\int_{f_{e1}}^{f_{e2}} |NTF(f)|^2 df\right) - 10 \log\left(\int_{f_s}^{f_{e2}} |f_s|^2\right)$$
(18)

$$SNR = 10\log(\sigma_s^2) - 10\log(\sigma_e^2) - 10\log\left(\int_{f_{e1}}^{f_{e2}} |NTF(f)|^2 df\right) + 10\log\left(\int_{f_B}^{f_{e2}} |NF(f)|^2 df\right) + 10\log\left(\int_{f_B}^{f_{e2}} |NF(f)|^2 df\right)$$
(19)

$$SNR = 10\log(\sigma_s^2) - 10\log(\sigma_e^2) - 10\log\left(\int_{f_{c1}}^{f_{c2}} |NTF(f)|^2 df\right) + 10\log(2OSR) \quad (20)$$

If
$$OSR = 2^r$$
 (21)

$$SNR = 10\log(\sigma_s^2) - 10\log(\sigma_e^2) - 10\log\left(\int_{f_{e1}}^{f_{e2}} |NTF(f)|^2 df\right) + (3+3r)dB$$
(22)

Every doubling in the OSR is supposed to increase the SNR by 3 dBs. In [7] the NTF integration is calculated for LP, integrator-based NTFs. The resulting SNR estimation is that; for every doubling of OSR an SNR increment of (6L+3) dB is expected, where L is the order of the modulator. In Section 3.6 the obtained SNRs are discussed, where it is shown that the (6L+3) dB increment cannot be obtained for the VBP Σ - Δ modulators when the OSR is doubled.

In this study, the SNR is directly calculated from the actual Σ - Δ modulator output using Matlab routines to obtain a more accurate value. This is due to the fact that the quantization operation is highly non-linear and an assumption is used to model its behaviour. This causes deviations in the noise-shaping. Furthermore, limit cycle tones, quantization tones and dithering noise are also added to the output that cannot be estimated by (16).

2.6.2 Dynamic Range

The Dynamic Range (DR) of a Σ - Δ modulator is defined as the ratio between the maximum and the minimum detectable signal amplitude values. The maximum detectable signal highly depends on the quantizer overload level, which drives the Σ - Δ modulator into unstable mode. The minimum detectable signal level depends on the noise-floor and is generally determined by the dither.

2.6.3 Stability

The stability of a Σ - Δ modulator is affected by many parameters. There is no precise mathematical model presented to-date that determines the stability of a Σ - Δ modulator. This in return, requires researchers to perform numerous iterative simulations in order to acquire an NTF that suitably meets the required design specifications.

A Σ - Δ modulator is said to be unstable if the output produces alternating long strings of 1s and 0s as is the case with low frequency oscillations. Another aspect of an unstable modulator is that the modulator exhibits quite large or unbounded states. In other words, the input

amplitude to the quantizer accumulates and results in extremely low gains for the quantizer. Consequently, the Σ - Δ modulator fails to perform noise-shaping. Both definitions are valid and need to be inspected.

In [66], the stability of a Σ - Δ modulator was shown to correlate with the NTF out-of-band gain where a rule of thumb was developed based on detailed simulations. Lee's stability criterion is widely used and is given by:

$$|NTF(z)| < 2 \tag{23}$$

In [67], the power of the NTF, rather than the gain is inspected. The resulting rule of thumb is called the power gain rule. According to [67], the total power of the normalized NTF needs to be less than 3 for the modulator to be stable. However, these rules are not sufficient to determine the stability of a Σ - Δ modulator. Because [66] is based on the simulation result of a fourth-order LP Σ - Δ modulator and in [67] the white noise assumption is used.

In the linear model, a more deterministic approach is provided. The idea is to model the quantizer as a variable gain block and a binary quantizer as seen in Figure 16. Thus the correlation of the input signal, modulator states and the modulator output are ensured. The resulting equations are calculated in (24) and (25) with the help of the linear quantizer noise model.



Figure 16: Linear Stability Model

$$STF(z,k) = \frac{kH(z)}{1+kH(z)}$$
(24)

$$NTF(z,k) = \frac{1}{1+kH(z)}$$
(25)

To investigate for the stability of the Σ - Δ modulator the root locus is plotted. If for all values of k the poles of the NTF remain within the unit-circle, the modulator is said to be stable. Otherwise the input-signal statistics are needed to determine the stability boundaries of the Σ - Δ modulator. It should be noted that the root locus method is applied to the designed TI VBP Σ - Δ modulators to ensure their stability.

In [9], the modulator is split into two linear subsystems for the signal and for the noise as seen in Figure 17. The idea of this method is to correlate the PSD of the subsystems' quantizer outputs using describing functions. However, for higher-order modulators, solving these nonlinear equations is cumbersome and can be done using numerical methods.



Figure 17: Split Quantizer

2.6.4 Tonality

 Σ - Δ modulators may suffer from unwanted tones that will cause instability and/or ear detectable periodicities. Once again the non-linear nature of quantization interferes with the calculation of these tones, making it very hard to predict the frequencies and the amplitude of these tones.

 Σ - Δ modulators act like Finite State Machines (FSMs) producing a set of states within certain periods. If the output state varies with a period of T_L , it is highly possible to observe tones at f_s/T_L . These tones are known as limit cycle oscillation tones and are mostly seen in non-linear systems. However the presence and the amplitude of these limit cycle oscillation tones also depend on the input amplitude, input frequency, quantization step and modulator order [67].

Idle channel tones describe an alternative type of tonality. They are generally observed at distinct frequencies superimposed on a noise background as depicted in Figure 18. A very common assumption about idle channel tones is that they are aliased high frequency harmonics of the limit cycle tones [69]. There are also some authors who assume that all

tonality sources in Σ - Δ modulators are caused by the periodic behaviour of the states. This depends on a number of variables such as the input amplitude, input frequency, quantization steps, modulator topology and modulator order. These tones may alias, diminish or be amplified.



Figure 18: a) Limit Cycle Tones b) Idle Channel Tones

Papers that calculate the amplitude and frequency of these tones either use the AWGN approximation and/or successive simulation tools. Mostly, papers report individual topologies seeking tone-free or at least less tonal responses. These publications mainly agree on;

- Σ - Δ modulators whose inputs are periodic or DC produce more tones in their outputs.
- Lower order Σ-Δ modulators suffer from more tonality when compared to higher-order Σ-Δ modulators.
- Multi-stage Σ-Δ modulators are less prone to tonality when compared to single-stage Σ-Δ modulators.
- Initial conditions may play a vital role to break the short cycles therefore reducing the unwanted tones at the Σ-Δ modulator's output.
- Randomization is the key solution to these idle channel tones and can be accomplished either by employing more complex Σ-Δ modulator structures; or adding dither noise within the feedback loop.

Another cause for the unwanted tones is the quantization of the input signal and harmonic distortion. This will be addressed and mathematically modelled in Sections 3.11 and 4.3 for sinusoidal input signals. Moreover, TI topologies specifically suffer from channel mismatch tones that are elaborately discussed in Section 3.11.3.

2.6.5 Dithering

Dither refers to a source of white noise that is used to randomise the strong tones that appear at the output. However, dither inevitably increases the noise floor. Still, it is best injected at the quantizer input to take advantage of the noise-shaping process.

2.7 Concluding Remarks

The main objective of this chapter was to present the motivations of generalized TI VBP Σ - Δ modulators. In the first place, the gaps in the theory are defined. Therefore, the case for the design of generalized VBP Σ - Δ modulators was made. The basic principles of data conversion are explained covering quantization, sampling and their mathematical analysis.

Secondly, Σ - Δ modulators are presented as a solution to the conventional Nyquist rate PCM converters. However these modulators lack speed due to their high OSR requirements and are limited to baseband applications. BP Σ - Δ modulators provide solutions for narrow band RF applications as these modulators are confined to mid-band operations and there is no control over the conversion band.

To this extent, generalized VBP Σ - Δ modulators are the best candidates allowing noiseshaping at the desired centre-frequency with an adjustable bandwidth. However, these Σ - Δ modulators still suffer from high OSR requirements and are more suitable for narrow band applications. The TI technique is presented as a solution to overcome this narrow band restriction. A detailed up-to-date literature survey is discussed covering the fundamental TI conversion techniques, their applications and implementations. The node equation method is identified for its ease of application and therefore is chosen in the conversion of SP VBP Σ - Δ modulators to their TI counterparts. The node-equation method is subsequently extended for VBP Σ - Δ modulators and a step-by-step technique is developed. This process can be used to construct the P-path generalized TI VBP Σ - Δ modulators and is detailed in Section 3.3.5.

Finally, the main performance criteria that can be used to compare and evaluate the performance of the designed Σ - Δ modulators are defined and explained.

Chapter 3. A/D TI VBP Σ - Δ Modulators

An A/D converter is the key building block for a highly digitized and monolithic receiver. To clarify this statement, a superheterodyne receiver architecture is depicted in Figure 19. Superheterodyne receivers firstly invented in 1910s are still widely used these days [70]. The idea is to modulate the input RF signal to a fixed IF frequency by means of a mixer. The modulated IF signal is then filtered and amplified to ensure a band-limited signal for A/D conversion. In order to have multi-channel implementation, a superheterodyne receiver requires discrete analog RF amplifiers and mixer blocks, which are highly complex to build and power consuming. Clearly the superheterodyne receiver is not suitable for integration.



Figure 19: Superheterodyne Receiver

On the other hand, a direct conversion scheme also known as the homodyne receiver eliminates the passive filtering therefore providing better integration when compared to the superheterodyne receiver. The principle is to convert the RF signal to baseband frequency using a quadrature mixer [71]. Thereafter the analog baseband signal is applied to an ADC. The block diagram of a homodyne receiver is given in Figure 20.



Figure 20: Homodyne Receiver

However, the direct conversion suffers from DC offset, I/Q mismatch as well as flicker noise problems. Rather than applying baseband conversion, low-IF conversion may be utilized to avoid DC offset and to mitigate the flicker noise. Yet, the LPF needs to be replaced with complex Band Pass Filters (BPF)s. Complex BPFs are not only hard to build but also use the double the chip area and therefore increase power consumption [72].

All these problems can be overcome by employing BP Σ - Δ modulators as depicted in Figure 21. Note that the mixer circuit and passive circuitry are excluded. The flicker noise and DC offsets are also avoided.



Figure 21: Prospective Receiver with Tunable ADC

The underlying problems of BP Σ - Δ modulators are defined in Section 2.4. A summary of the issues is listed below.

- A conventional BP Σ-Δ modulator design can be derived from its LP prototype using the z⁻¹ to -z⁻² transformation. The resulting modulators are resonator based midband Σ-Δ modulators, which may have stability issues.
- 2. There is no control over the bandwidth of the NTF, which results in Σ - Δ modulators that are confined to narrow-band applications.
- 3. Flexibility of the overall system is restricted since the conversion frequency is fixed. In addition, multi-mode, multi-channel implementations suffer from signal aliasing.

The VBP Σ - Δ modulators, presented in Section 2.4, overcome the above limitations and offer the advantages listed below.

- 1. Flexibility is introduced where the designer is able to specify the center frequency bandwidth, stopband and passband attenuations from a menu of filters. This in return provides:
 - a. More stable Σ - Δ modulator topologies.
 - b. Improved integration for IC circuits.
 - c. Improved SNR and DR.
- 2. The narrow band restriction is overcome by the time-interleaving technique thus enabling higher frequency and wider band conversion.

3.1 VBP Σ - Δ Modulator Design

It is anticipated due to the above benefits that VBP Σ - Δ modulator will become more widespread. Several Σ - Δ modulator structures are proposed and the entire frequency band from DC to Nyquist is investigated and evaluated in this study. To do so, generalized topologies are built that are capable of noise-shaping at any desired frequency for Butterworth, Chebyshev, Inverse-Chebyshev and Elliptical filters.

A VBP Σ - Δ modulator can be designed for specific applications where the centre frequency, bandwidth, pass-band and stop-band attenuations can be prescribed by the designer. The system's clock and signal's bandwidth can be directly utilized whilst keeping the modulator stable. The required SNR can also be obtained with the help of different filters. In this sense the designer can attain a fully integrated signal conversion as no mixer circuitry and pre-filtering is required.

Another benefit is that the digital processor can be clocked with the highest value if required, as there is no frequency restriction of the data conversion. For example, the input signal with a centre-frequency equal to f_x and the processor can be clocked at $10f_x$. If a conventional BP Σ - Δ modulator is employed, the sampling frequency needs to be $4f_x$. However, if a VBP Σ - Δ modulator is employed with a centre frequency of $\nu = 0.1$ as seen in Figure 22, the designer will get the benefit of high speed processing as well as the higher OSR.



Figure 22: Output Plots for Different Clock Frequencies

Another advantage of VBP Σ - Δ modulators is that tunable, multi-mode and/or multi-channel data conversion is made feasible. As already stated, there is not much work reported in the open literature on VBP Σ - Δ modulators with the exception of a few publications on tunable or programmable ADCs that focus on tuning the centre-frequency of the modulator to the already defined fixed values [73-75]. In [73], 9 different centre-frequencies from $0.1f_s$ to $0.4f_s$ are predetermined and can be tuned. In [74], the centre-frequency can be tuned from DC to $0.31f_s$ with intervals of $0.052 f_s$. In [75], 5 different centre-frequencies can be selected from DC to 1GHz.

The terms tunable and programmable slightly differ from each other. Tunable converters are mostly built with CT Σ - Δ modulators where a set of frequencies can be tuned to the chosen frequency band. In other words, channel selection is achieved where the bandwidth is fixed. On the other hand, programmable converters are constructed using discrete-time Σ - Δ modulators, where the user can select the conversion band and centre-frequency using switches. Yet, during the literature survey, a few publications have drawn the attention of the author that tunable switched capacitor circuits are being investigated mostly for MEMs applications, making it possible to build tunable, discrete-time Σ - Δ ADCs [76-80]. For those who are interested in the implementation of these circuits, the general idea is to tune the time constant by switches in series with the capacitors.

3.2 Generalized VBP NTF Design

The design of a Σ - Δ modulator starts with the NTF. The SNR, stability and DR are all dependent on the NTF. Choosing a flat response for the NTF provides more stable Σ - Δ modulators as this evens the out-of-band gain levels close to DC and Nyquist. As a result overloading of the quantizer and accumulators is alleviated. Therefore Butterworth filters are widely employed to ensure stability of the Σ - Δ modulator. On the other hand, Chebyshevbased filters offer higher resolution as their transition bands are narrower and steeper when compared to Butterworth filters [7].

Since the intention in this thesis is to build generalized topologies, a generalized NTF is chosen to realize Butterworth, Chebyshev and Elliptical filter-based Σ - Δ modulators. The generalized NTF of an L^{th} -order Butterworth, Chebyshev and Elliptical band-stop filter is given in (26). The coefficients of a suitable band-stop Butterworth, Chebyshev, Inverse Chebyshev or Elliptical filters can be obtained using Matlab. The generalized transfer function of an L^{th} -order NTF is given by:

$$NTF_{gen}(z) = \frac{\prod_{k=1}^{L/2} \left(1 - 2b_k z^{-1} + z^{-2}\right)}{1 + a_1 z^{-1} + \dots + a_{L-1} z^{-L+1} + z^{-L}}$$
(26)

An Output Feedback (OF) topology is chosen to explain the design methodology since its NTF and STF have already been derived as already presented in (7). The block diagram given in Figure 23 is modelled in Matlab/Simulink where an Infinite Impulse Response (IIR) filter block is used to represent the loop-filter. The system level design in Matlab/Simulink helps practitioners to improve their understanding of the operation of VBP Σ - Δ modulators for different loop-filters, centre frequencies and bandwidths.



Figure 23: The OF Σ - Δ Modulator Block Diagram

After building the OF Σ - Δ modulator block diagram in Figure 23, designers can calculate the loop-filter's numerator and denominator coefficients in MATLAB without building the loop-filter topology. Therefore, the final loop-filter and Σ - Δ modulator coefficients may need to be adjusted following extensive simulations to meet the required specifications including the SNR, DR, stability and tonality. It should be noted that regarding the NTF filter type, the following functions can be employed, i.e., butter for Butterworth filters, cheby1 for Chebyshev filters, cheby2 for Inverse-Chebyshev filters and ellip for Elliptical filters.

3.3 VBP Σ - Δ Modulator Topologies

The second step in the design of VBP Σ - Δ modulator is mapping the generalized NTF in (26) to a generalized topology. The choice of the topology does not affect the performance at the behavioural level. From an implementation perspective, however, different topologies that use the same initial NTF may produce slightly different output response and resolutions. This is attributed to the sensitivity of coefficient variation of the Σ - Δ modulator structures. Therefore, this section mainly focuses on the choice of the building blocks: resonator-based or integrator-based.

3.3.1 Chain of Integrators with FeedBack and FeedForward Coefficients (CI-FBFF)

As implied from its name, the CI-FBFF topology utilizes integrators as its main building blocks. Integrators process the variations of the input signal as they accumulate and low-pass filter the in-coming signal. Therefore, integrators are the key building blocks of Σ - Δ modulators. The block diagrams of delayless and delayed discrete-time integrators are boxed and displayed in Figure 24 in which the second-order SP CI-FBFF topology is shown. This topology is also used to explain the node-equation technique and its 2-path TI counterpart equations are calculated in (35)- (48). So internal nodes are named as n1, n2, n3, n4 and n5.

The behavioural-level drawings in the thesis are composed using the Matlab Simulink R2010b. For readers who are not familiar with the Simulink, Table 2 is given before the SP CI-FBFF topology explaining the building blocks. It must be noted that, the Simulink drawings are converted to EPS (Encapsulated PostScript) format using the "print" function of Matlab. Therefore, the converted EPS files do not owe the same precision for all topologies.

Moreover, in Appendix A, Appendix B, Appendix C and Appendix D the sizes of the building blocks may differ from one topology to another to fit individual topologies into a single page. Table 2 : Building Blocks of Simulink

Building Block	Function
INPUT	The sampled single-sinusoidal input signal.
Dither Noise	The white noise source used as the dither noise.
× × ×	Gain block used to apply the feedback and feedforward multiplications.
>+ >	Adder block.
$\left \frac{-1}{z} \right $	Delayer.
> ↓ _D >	Downsampler.
> ↑ ∪ >	Upsampler.
Quantizer	Quantizer.
Out1 > LFSR	Linear feedback shift register built by the author using Simulink.



Figure 24: 2nd-order SP CI-FBFF Topology

The topology basically determines the numerator of the NTF by feedback coefficients and the denominator of the NTF by feedforward coefficients given in (27). Therefore, without even considering the non-idealities, it can be asserted that non-idealities of the feedback coefficients will degrade the SNR and DR performance. This is due to the fact that the in-band quantization noise is determined by the numerator of the NTF. On the other hand, the feedforward coefficients are calculated using the denominator coefficients of the NTF. In other words, the pole locations are affected by the feedforward coefficients. So the feedforward coefficients will affect the SNR, DR and stability performance.

$$NTF_{2^{nd}CI}(z) = \frac{1 + (K_1 - 2)z^{-1} + z^{-2}}{1 + (L_1 + (K_1 - 2))z^{-1} + (L_2 + 1)z^{-2}}$$
(27)

In addition, the feedforward and feedback multiplications of each node are summed up together at the same adder. This in return may cause excessive signal accumulation leading to an unstable modulator. It can be foreseen that this topology gives a low dynamic range.

3.3.2 The TI Technique

One of the major issues of Σ - Δ modulation is the bandwidth restriction, less than a few MHz, arising from the oversampling requirements. Using higher-order NTFs may relax the oversampling demands as the required SNR is obtained by a steeper noise-shaping. However, higher-order NTFs drastically suffer from stability problems and low DRs [81]. Herein parallel working Σ - Δ modulators offer an appropriate solution to increase the conversion

bandwidth without sacrificing the performance. Yet the fundamental drawback of parallelism is the increased hardware.

There are various parallel structures in the open literature that extend the conversion bandwidth using different approaches. For instance, the frequency-band decomposition technique involves partitioning the conversion bandwidth into sub-bands each being processed by different channels at the same time [82]. This technique is challenging due to the design complexity since each channel requires different NTFs and therefore different structures [83]. Eventually larger conversion bandwidths or lower speed requirements can be accomplished at the expense of design complexity.

Another parallel Σ - Δ modulation technique is modulated-based architectures in which the input and output sequences are modulated by external signals [83]. Their circuitry and overall performance differ depending on the modulation sequence. A popular parallel structure is the Time-Interleaved Σ - Δ modulator, as it provides efficient and easy conversion from a SP Σ - Δ modulator. The input signal samples are distributed over the parallel, cross-connected channels by means of a commutator hence each sample is processed by different channels. The output is then combined in a sorted manner to obtain a single output sequence. Basically the input and the output signals are modulated by an identity matrix.



Figure 25: Block Diagram of the TI Topology

As depicted in Figure 25, the TI approach employs P mutually cross-coupled Σ - Δ modulators, each operating at a sampling frequency of f_s , which results in an equivalent sampling rate of Pf_s [65]. Therefore, the TI Σ - Δ modulators also operate as multi-rate signal processors.

The design procedure of a TI Σ - Δ modulator commences with the design of a SP structure. Afterwards using one of the methods in the literature, the SP Σ - Δ modulator is converted to its TI counterpart. Mainly, there are two reported methods that enable the conversion of a SP Σ - Δ modulator to its P-path TI counterpart. The first method involves mathematically developing the polyphase decomposition elements of the loop-filter and mapping the resulting matrix to the TI topology [65]. It is clear that this method is challenging due to the cumbersome mathematical operations. Furthermore, for a VBP Σ - Δ modulator with the given NTF in (26), these mathematical operations are much more complex.

The second method, which will be further developed in this thesis, is mathematically less intensive and involves writing node equations of the SP Σ - Δ modulator topology in the time domain [20]. The fundamental idea underlying the node equation conversion is to share the samples with the subsequent and/or adjacent channel for the next time interval if needed. This in return results in fewer components when compared to the polyphase decomposition technique. In addition, it is an efficient and easy technique to apply. Another elegant feature of the node equation method is that it allows designers to build generalized topologies, as the coefficient values in the SP node-equations do not affect the resulting TI topology. However, the polyphase decomposition differs depending on the NTF coefficients.

3.3.3 Downsampler

The downsampler and the upsampler are the two vital building blocks of the TI topologies as they multiply the incoming signal by an identity matrix. The downsampling operation, generally known as decimation, is defined as reducing the sampling rate of a discrete-signal by a certain downsampling ratio.

In Figure 28, the time-domain behaviour of a 2-path downnsampler is shown. In Figure 26 the block diagram of a D-factor downsampler is depicted and in (29) the mathematical model of a downconverter is given where D is the dowsampling ratio.

$$\stackrel{x(n)}{\checkmark} D \stackrel{x_d(n)}{\checkmark}$$

Figure 26: Downsampler Block Diagram

$$x_d(n) = x[nD] \tag{28}$$

$$X_d\left(e^{j\omega}\right) = \frac{1}{D} \sum_{i=0}^{D-1} X\left(e^{j\left(\frac{\omega}{D} - \frac{2\pi}{D}\right)}\right)$$
(29)

If the z-transform is applied;

$$X_d(z) = \frac{1}{D} \sum_{i=0}^{D-1} X(z^{1/D} W^i)$$
(30)

where;

$$W^{i} = e^{-(2\pi/D)^{i}}$$
(31)

Downsampling produces D-1 shifted copies of the stretched input signal. Stretching occurs because of $z^{1/D}$ and shifting occurs because of W^i . These images are cancelled by the upsampler within the same topology as long as they are band-limited to π/D .

3.3.4 Upsampler

The upsampling operation, generally known as interpolation, is defined as increasing the sampling rate of a discrete-signal by an upsampling ratio. The block diagram of an upsampler is given in Figure 27 and its mathematical model is provided in (34) where U is the upsampling ratio.

$$\stackrel{x(n)}{\frown} U \stackrel{x_u(n)}{\frown}$$

Figure 27: Upsampler Block Diagram

$$x_u(n) = x[\frac{n}{U}] \tag{32}$$

$$X_u(e^{j\omega}) = X(e^{j\omega U})$$
(33)

If the z-transform is applied;

$$X_u(z) = X(z^U) \tag{34}$$

Upsampling creates U-fold compressed images of the input signal. These images are caused by the sampled-nature of the input signal.

3.3.5 Node Equation Method

The node-equation method was firstly presented in [20] in which a step-by-step methodology is developed for the conversion of LP SP Σ - Δ modulators to their P-path TI counterparts. In this thesis, the node equation method is further extended to VBP Σ - Δ modulators. Furthermore, using this technique 2-path and 4-path D/A TI VBP Σ - Δ modulators are implemented on VHDL validating the node-equation method and the resulting circuitry's performance. Assuming an already designed and examined SP topology such as Figure 24, the node equation method is applied as follows.

Step 1: Each node is named as seen in Figure 24 and the equations of these nodes are written in the time domain as in (35). Note that the DAC in the feedback loop is assumed to be ideal, in other words its gain is equal to unity and can be regarded as short circuited when the node equations are written. In addition, the non-ideal behaviour of the quantizer is modelled by the Q(n) function whereas the dither noise is represented by d(n).

$$n_{1}(n) = x(n) - y(n)$$

$$n_{2}(n) = n_{1}(n) - K_{1}n_{4}(n)$$

$$n_{3}(n) = n_{2}(n) + n_{3}(n-1)$$

$$n_{4}(n) = n_{3}(n-1) + n_{4}(n-1)$$

$$n_{5}(n) = L_{1}n_{4}(n) + L_{2}n_{4}(n-1) + d(n)$$

$$y(n) = Q(n_{5}(n))$$
(35)

Step 2: This step involves modelling the behaviour of the downsampling in the time domain. A downsampler is basically a p-state commutator. For a downsampling level of two, the downsampler is a two-state commutator that distributes the input signal samples over the time interleaving paths as seen in Figure 28. It is advisable to make a P-path TI signal distribution diagram before moving to the third step.



Figure 28: 2-level Downsampler Behavioural Model

Step 3: The third step is to label each node in accordance with the desired path number of the TI topology. Since a 2-path TI topology is intended, the nodes are labelled as in (36).

$$x(2n) = x_{1}(n), x(2n-1) = x_{2}(n)$$

$$n_{1}(2n) = n_{1,1}(n), n_{1}(2n-1) = n_{1,2}(n)$$

$$n_{2}(2n) = n_{2,1}(n), n_{2}(2n-1) = n_{2,2}(n)$$

$$n_{3}(2n) = n_{3,1}(n), n_{3}(2n-1) = n_{3,2}(n)$$

$$n_{4}(2n) = n_{4,1}(n), n_{4}(2n-1) = n_{4,2}(n)$$

$$n_{5}(2n) = n_{5,1}(n), n_{5}(2n-1) = n_{5,2}(n)$$

$$y(2n) = y_{1}(n), y(2n-1) = y_{2}(n)$$
(36)

Step 4: The final step is to convert each equation individually to its corresponding path-node equations calculated in (37)-(48).

$$n_{1}(2n) = x(2n) - y(2n)$$

$$n_{1,1}(n) = x_{1}(n) - y_{1}(n)$$
(37)

$$n_{1}(2n-1) = x(2n-1) - y(2n-1)$$

$$n_{1,2}(n) = x_{2}(n) - y_{2}(n)$$
(38)

$$n_{2}(2n) = n_{1}(2n) - K_{1}n_{4}(2n)$$

$$n_{2,1}(n) = n_{1,1}(n) - K_{1}n_{4,1}(n)$$
(39)

$$n_{2}(2n-1) = n_{1}(2n-1) - K_{1}n_{4}(2n-1)$$

$$n_{2,2}(n) = n_{1,2}(n) - K_{1}n_{4,2}(n)$$
(40)

$$n_{3}(2n) = n_{2}(2n) + n_{3}(2n-1)$$

$$n_{3,1}(n) = n_{2,1}(n) + n_{3,2}(n)$$
(41)

$$n_{3}(2n-1) = n_{2}(2n-1) + n_{3}(2n-2)$$

$$n_{3,2}(n) = n_{2,2}(n) + n_{3,1}(n-1)$$
(42)

$$n_{4}(2n) = n_{3}(2n-1) + n_{4}(2n-1)$$

$$n_{4,1}(n) = n_{3,2}(n) + n_{4,2}(n)$$
(43)

$$n_{4}(2n-1) = n_{3}(2n-2) + n_{4}(2n-2)$$

$$n_{4,2}(n) = n_{3,1}(n-1) + n_{4,1}(n-1)$$
(44)

$$n_{5}(2n) = L_{1}n_{4}(2n) + L_{2}n_{4}(2n-1) + d(2n)$$

$$n_{5,1}(n) = L_{1}n_{4,1}(n) + L_{2}n_{4,2}(n) + d(n)$$
(45)

$$n_{5}(2n-1) = L_{1}n_{4}(2n-1) + L_{2}n_{4}(2n-2) + d(2n-1)$$

$$n_{5,2}(n) = L_{1}n_{4,2}(n) + L_{2}n_{4,1}(n-1) + d(n)$$
(46)

$$y(2n) = Q(n_5(2n))$$

$$y_1(n) = Q(n_{5,1}(n))$$
(47)

$$y(2n-1) = Q(n_5(2n-1))$$

$$y_2(n) = Q(n_{5,2}(n))$$
(48)

Following the conversion of the equations, the corresponding TI Σ - Δ modulator topology is constructed as demonstrated in Figure 29. Moreover, the 4-path counterpart is shown in Figure 30.



Figure 29: 2nd-order 2-path CI-FBFF Topology



Figure 30: 2nd-order 4-path CI-FBFF Topology

3.3.6 Design Technique to Build Up Generalize TI VBP Σ - Δ Modulators

A step-by-step design methodology of TI VBP Σ - Δ modulators is given below:

- 1. Build an SP Σ - Δ modulator with an IIR block in Simulink to simulate the NTF given in (26).
- 2. Choose appropriate coefficients for the generalized NTF to implement Butterworth, Chebyshev, Inverse Chebyshev or Elliptical filters.
- 3. Analyse the SP Σ - Δ modulator topology in terms of its stability, resolution and dynamic range. If the required performance metrics are not met go to 2.
- 4. Map the chosen NTF to an SP Σ - Δ modulator topology.
- 5. Build the TI counterpart of the SP Σ - Δ modulator topology using the node-equation method.
- 6. Analyse the TI Σ - Δ modulator in terms of its stability, resolution and dynamic range.

3.3.7 Chain of Resonators with Feedforward and Local Resonator FeedBack Coefficients (CR-RFB)

This second topology is constructed using resonator blocks as shown in the dashed square of Figure 31 in which the 2nd-order SP CR-RFB topology is illustrated. Earlier in Section 2.4, a resonator structure was given in Figure 13, where two cascaded integrator blocks were utilized. This conventional topology has been employed in many applications due to its well-known structure and non-ideal model [84-86]. However, in [87] and [88], single-opamp SC resonators are proposed where double-sampling clocking is required. Single-opamp resonator blocks are gaining popularity as they lower the power dissipation and hardware of the overall circuitry. Moreover, in [89], a double-sampling SC resonator is proposed whose numerator is immune to circuit non-idealities, hence the resonant frequency and the Q-factor are not affected by the capacitance values.



Figure 31: 2nd-order SP CR-RFB Topology

The corresponding symbolic NTF of the 2^{nd} -order SP CR-RFB topology is given in (49). Once again the feedback coefficients determine the zeros of the NTF whereas the feedforward coefficients determine the poles.

$$NTF_{2^{nd}CR}(z) = \frac{1 + (K_1)z^{-1} + z^{-2}}{1 + (L_1 + K_1)z^{-1} + (L_2 + 1)z^{-2}}$$
(49)



Figure 32: 2nd-order 2-path CR-RFB Topology



Figure 33: 2nd-order 4-path CR-RFB Topology

3.4 Hardware Complexity of the A/D VBP Σ-Δ Modulators

Although the A/D VBP Σ - Δ modulators are not implemented in this thesis it worth pointing out the hardware complexity of the resulting topologies. Regardless of the topology type, the TI topologies utilizes PxN_{ad} adders and PxN_{mul} multipliers where P is the TI path number, N_{ad} and N_{mul} are the numbers of the adders and multipliers used to build the SP counterpart of the TI topology. Hence the relaxed sampling frequency or the higher OSR ratio is obtained at the expense of increased hardware. However, the SP and TI topologies strictly differ from each other in terms of their building blocks.

As mentioned earlier, the node-equation method shares the samples of the internal nodes between the adjacent and/or subsequent channels. Therefore, the resonators and integrators of the SP topology cannot be observed for the TI topologies. If Figure 29, Figure 30, Figure 32

and Figure 33 are investigated it can be seen that, the resonator and integrators are distributed over the channels. So, the main building blocks of the TI topologies are delayless and/or delayed adders and multipliers.

In this sense, if the TI topologies are implemented using voltage-mode circuits, designers will end using PxN_{ad} adders and PxN_{mul} multipliers. However, if the TI topologies are implemented using current-mode circuits, designer can save the PxN_{ad} adders, because adding occurs through the internal nodes of the current-mode circuits. Consequently current mode TI topologies will use PxN_{mul} multipliers only. Once again, this advantage can be benefited for the TI topologies built using the node-equation method. Hence their main building blocks are delayless and/or delayed adders and multipliers.

Finally, if the CR-RFB and the CI-FBFF topologies are compared, they both utilize LxP feedforward coefficients where L is the order of the modulator and P is the TI path number. On the other hand, the CR-RFB topology uses PxL/2 - 1 fewer feedback coefficients with respect to CI-FBFF topology. Obviously, for multi-path TI topologies it is wise to choose the CR-RFB topology in order to save multipliers and adders. Hence each reduction in the feedback coefficients equals to the reduction of one adder and one multiplier. Consequently, the CR-RFB topology can be implemented saving PxL/2 - 1 adders and PxL/2 - 1 multipliers.

3.5 Behavioural Level Simulations of the CI-FBFF Topology

The behavioural level-simulations are explained in this section for two reasons; firstly it is intended not to disturb the natural flow of the thesis, as the node equation technique needs to be well covered before getting into the analysis of the designed modulators. Secondly, this section also involves the comparison of the designed SP and TI CI-FBFF and SP and TI CR-RFB topologies for different orders.

2nd-, 4th-, 6th-, 8th-order SP, 2-path, 4-path topologies of CI-FBFF and CR-RFB are built and the resulting structures including their symbolic NTFs are given in Appendix A and Appendix B.

Simulation Methodology: The behavioural level simulations of the SP CI-FBFF topologies of any order starts with measuring its SNR for different centre frequencies. Because the SNR plots involve the DR and output plots within the code, the output plot can be observed for discrete input amplitude values for each iteration. If any unstable output is noticed, the pole and zero locations of the designed filter type and resulting feedback and feedforward coefficients are noted to determine the hazardous region. If not, iterative simulations sweep the centre frequency for a fixed bandwidth until an unstable coefficient set is determined.

Afterwards, for each filtering type the safe centre frequencies are fixed and the bandwidth is swept to determine the bandwidth boundaries for each discrete centre frequency. These preliminary and time-consuming simulations are applied to each designed topology mentioned in the anterior sections to predict the stability of each topology.

After the stability, SNR and DR are provided, the designed topologies are compared and discussed in Section 3.9. Moreover the performance of Butterworth, Chebyshev, Inverse-Chebyshev and Elliptical are compared and examined in Section 3.9. Finally and more importantly, the non-idealities are modelled in each designed topology and the resulting performances are given in Section 3.10.

SNR Code: The major concern in the SNR code is to correctly select the Fast Fourier Transform (FFT) samples, windowing function and overlapping number. The FFT conversion inevitably suffers from signal energy leakage due to the aperiodic nature of the Σ - Δ output data. To reduce this leakage, a Hanning window is selected as it provides good frequency resolution whilst keeping the signal amplitude and spectral leakage under control [90]. However, Hanning window is not sufficient to depict all the harmonics and other tones in the frequency band. Therefore, the overlapping technique is utilized to minimize the leakage. The idea of overlapping is to apply overlapping window frames to the output in order to reduce the loss of information as seen in Figure 34.



Figure 34: Windowing and Overlapping

Since windowing and overlapping are applied, the FFT size only needs to cover the period of the processing data to ensure good resolution. For this, as long as the periodicity of the output data is covered, the length of the FFT just determines the resolution. The wider the FFT window is chosen, the better resolution is provided in the frequency domain and the poorer resolution is obtained in the time-domain. In this study, an FFT window size of 2^{20} is chosen with an overlapping 8 windows.

Finally, oversampling is achieved by simply filtering the output data using the FFT sample number and OSR. The signal-band is calculated as in (50) where N_{FFT} is the FFT sample number and P_{TI} is the time-interleaving path number.

$$BW_{signal} = \frac{N_{FFT}/2}{P_{TI} \times 2 \times OSR}$$
(50)

The negative frequency is discarded for the calculations and therefore N_{FFT} is divided by 2. Moreover, to accomplish accurate resolution for the TI topologies, the signal bandwidth is narrowed, i.e., the bin size is reduced for the ease of calculation. As a result, the OSR is increased.

3.6 Analysis of the Behavioural-Level Topologies

In Figure 35 - Figure 38, the output plots of discrete-time CI-FBFF and CR-RFB are given when the same NTF is applied to both topologies. These simulation results validate the node-equation method and resulting TI VBP Σ - Δ modulators' performance. Secondly, they highlight

the characteristics of the topologies. For instance; the tones seen in Figure 36 - Figure 37 need to be investigated. More importantly, in the ideal case the CI-FBFF and CR-RFB topologies are supposed to perform the same NTF function with different feedback and feedforward coefficients. However in Figure 38, it is seen that CR-RFB topology has a deeper notch.

Different VBP Σ - Δ modulators are compared and evaluated at the behavioural level in terms of their order, filter type, topology and path number.



Figure 36: 2P 4th-order, Chebyshev Filtering



Figure 38: 4P 8th-order, Inverse-Chebyshev Filtering, Zoomed Plot

3.7 Coefficient Comparison

This section will discuss in detail the effects of the coefficients on the output amplitude frequency response of the proposed VBP Σ - Δ modulator topologies. As will be explained and illustrated in the upcoming sections, stability, tonality, DR and SNR degradation are mostly caused by the coefficients' values and coefficient mismatches.

Feedback Coefficients: The CR-RFB topology uses PxL/2 feedback coefficients whereas the CI-FBFF topology uses (PxL) - 1 feedback coefficients where *P* is the TI path number and *L* is the order of the modulator. Consequently a P-path Lth –order CR-RFB topology saves PxL/2 - 1 feedback coefficients with respect to the same path number and order CI-FBFF topology.

Another major difference between the two topologies is that the CR-RFB topology utilizes local feedback coefficients. Thereby the resonant frequency is directly determined by the local feedback coefficients. Hence, for the CR-RFB topology, the local feedback coefficients always fall within the range [-2, 2], which is map to the normalized frequency values of [0, 0.5].

For the CI-FBFF topology, the resonant frequency is determined by multiple feedback coefficients as given in Appendix A. This in turn may cause the feedback coefficients of the CI-FBFF topology to be excessively large up to 900.

Large feedback coefficients of the CI-FBFF topology unfortunately increase the accumulation in each node causing the integrators to be saturated even for small input amplitudes. This in turn, has a deleterious effect on stability and can only be observed in the case of non-ideal integrators.

The resulting values of feedback coefficients versus centre frequency of the Butterworth NTFs are given in Figure 39 - Figure 42.



Figure 39: 2nd-order, Butterworth NTF Feedback Coefficients, BW is fixed to 0.02



Figure 40: 4th-order, Butterworth NTF Feedback Coefficients, BW is fixed to 0.02



Figure 41: 6th-order, Butterworth NTF Feedback Coefficients, BW is fixed to 0.02



Figure 42: 8th-order, Butterworth NTF Feedback Coefficients, BW is fixed to 0.02

The frequency range between [0.25, 0.5] significantly degrades the DR as frequencies larger than $\nu = 0.25$ exponentially increase the feedback coefficients of the CI-FBFF topology. In addition, for higher order NTFs the mid-coefficients get the highest values. This is due to the fact that the cascaded topology forms the NTF numerator from the polynomial function. In other words, Pascal's Triangle polynomials are obtained simply by multiplying binomials of the resonator function of the generalized NTF [91].

As mentioned earlier in this section, the two topologies are supposed to produce the same NTF for the ideal-case behavioural-level simulations. However, it is also shown that some tones and notch dips do not match in the output plots. For the SNR plots depicted in Figure 56 on page 61, the ideal-case behavioural simulations are compatible for the CI-FBFF and CR-RFB topologies.

To give an initial idea on the implementation of large coefficients, it must be kept in mind that the coefficient values of the Switched Capacitor (SC) circuitry are determined by the ratio of the capacitors. In the meantime, the sampling speed is provided by two factors: the on-resistance of the switches and the value of the sampling capacitors. To achieve high-speed operation small capacitors must be utilized. In other words, the capacitor values need to be reduced for an area and/or power efficient implementation. Therefore, the implementation of the large feedback coefficients shown in Figure 41 and Figure 42 is physically impossible despite the fact that the circuit works perfectly well in a simulation environment.

Generally speaking;

- Centre frequencies higher than $\nu = 0.25$ increase the feedback coefficients of the CI-FBFF topology. Whilst the CR-RFB topology's feedback coefficients are always between [-2, 2
 - From the implementation perspective, the CI-FBFF topology cannot perform stable, low-power, low-hardware implementations for orders of 6 and above at the frequency interval [0.25 0.5].
 - The frequency interval [0 0.25] seems feasible for the implementation of CI-FBFF topology.

- The CI-FBFF feedback coefficients get their highest values in the mid-coefficients due to the nature of Pascal Triangle polynomials.
- The NTF type and filter specifications such as the pass-band and stop-band attenuations and bandwidth do not affect both topologies' feedback coefficients' range.

Feedforward Coefficients: The feedforward coefficients are determined by the pole locations of the NTF for both topologies. Therefore, the bandwidth, stop-band and pass-band attenuations strictly affect only the feedforward coefficient values. In other words, the feedforward coefficients depend on the NTF filter type.

The denominator of the symbolic NTFs of the CR-RFB topologies is affected by the feedback and the feedforward coefficients as given in Appendix B. The limited interval of the CR-RFB feedback coefficients keeps the feedforward coefficients to closer values of the same order magnitude to the denominator coefficients. However, the feedback coefficients of the CI-FBFF increase the feedforward coefficients especially for orders 6 and above.



Figure 43: 2nd-order, Butterworth NTF Feedforward Coefficients, BW is fixed to 0.02



Figure 44: 4th-order, Butterworth NTF Feedforward Coefficients, BW is fixed to 0.02



Figure 45: 6th-order, Butterworth NTF Feedforward Coefficients, BW is fixed to 0.02



Figure 46: 8th-order, Butterworth NTF Feedforward Coefficients, BW is fixed to 0.02

In Figure 47 - Figure 54 the BW sweep plots for the feedforward coefficients of both topologies are given for two distinct centre-frequencies; $\nu = 0.1$ and $\nu = 0.4$ for the Elliptical NTF. These plots are presented to show how the coefficient interval is affected by the centre-frequency when bandwidth is swept. As expected, the CR-RFB feedforward coefficient interval is not affected much (maximum of ± 2) by the bandwidth sweep for these different centre-frequencies.

However, the CI-FBFF feedforward coefficients for orders 6 and 8, exponentially increase for the centre-frequency interval [0.25 0.5]. As mentioned, this is actually caused by the cascaded structure of the CI-FBFF topology. Frequencies over 0.25 increase the feedback coefficients thus the internal node accumulation can be balanced by larger feedforward coefficients. This leads to quantizer overload as the feedforward coefficients are multiplied by the internal signals and added up just before the quantizer. Quantizer overload may cause unstable modulators.

If Figure 51 is compared to Figure 52 and Figure 53 is compared to Figure 54, it is clearly seen that the CI-FBFF topology's feedforward coefficients extensively increase; i.e., up to 1500 for the 6th-order NTF and up to 350 for the 8th-order NTF when the centre frequency equals to 0.4. However Figure 47, Figure 48, Figure 49 and Figure 50 show that the centre frequency does not cause significant change in the feedforward coefficient interval for the 2nd- and 4th-order CR-RFB and CI-FBFF topologies.

If a multi-frequency and/or multi-band system using the CI-FBFF topology is required, centrefrequency and bandwidth sweep simulations need to be held together. Once again, the CI-FBFF topology is not suitable for VBP Σ - Δ modulators whose input's centre frequencies fall within the interval of [0.25 0.5].



Figure 47: 2nd-order, Elliptical NTF Feedforward Coefficients, Centre Freq. = 0.1, Pass-Band Attenuation (PBA) = 1dB, Stop-Band Attenuation (SBA) = 80 dB



Figure 48: 2nd-order, Elliptical NTF Feedforward Coefficients, Centre Freq. = 0.4, PBA = 1dB, SBA= 80 dB



Figure 49: 4th-order, Elliptical NTF Feedforward Coefficients, Centre Freq. = 0.1, PBA = 1dB, SBA = 80 dB


Figure 50: 4th-order, Elliptical NTF Feedforward Coefficients, Centre Freq. = 0.4, Pass-Band Attenuation = 1dB, Stop-Band Attenuation = 80 dB



Figure 51: 6th-order, Elliptical NTF Feedforward Coefficients, Centre Freq. = 0.1, PBA = 1dB, SBA = 80 dB



Figure 52: 6th-order, Elliptical NTF Feedforward Coefficients, Centre Freq. = 0.4, PBA = 1dB, SBA= 80 dB



Figure 53: 8th-order, Elliptical NTF Feedforward Coefficients, Centre Freq. = 0.1, PBA = 1dB, SBA = 80 dB



Figure 54: 8th-order, Elliptical NTF Feedforward Coefficients, Centre Freq. = 0.4, PBA = 1dB, SBA = 80 dB

Generally speaking;

- Centre frequencies higher than $\nu = 0.25$ increase the feedforward coefficient of the CI-FBFF topology. However these centre frequencies do not have a noticeable effect on the CR-RFB topology's feedforward coefficients.
- The NTF type and filter specifications affect the topologies feedforward coefficients' value
 - The more aggressive the NTF is designed, the smaller the coefficients become.

3.8 Filtering Performance

The NTF filter type affects the performance and stability of the VBP Σ - Δ modulators as examined below.

Butterworth NTFs

Basically a band-stop Butterworth filter has its poles on a semi-circle with a radius of v_{co} where v_{co} is the cut-off frequency. Butterworth filters provide maximally flat out-of-band gain as their gain function derivatives of frequency approximates to zero. The output plots of the Butterworth-based VBP Σ - Δ modulators are depicted in Figure 35 on page 49.



Figure 55: Butterworth Pole-Zero Map

To give an overall understanding, Butterworth NTFs have a wider transition band compared to the other filter types. Hence Butterworth-based NTFs achieve much better SNRs and DRs for larger BW values. Due to its maximally flat out-of-band gain, the stability performance of Butterworth based NTFs are comparatively more robust. As far as the analyses allow, they have delivered stable outputs for all the designed VBP Σ - Δ modulators in the thesis for any centre frequency and up to a normalized bandwidth of 0.1.

The numerator and denominator coefficients of the desired Butterworth NTF are obtained using the 'butter' function in Matlab. These coefficients are mapped to the desired topology by using the Matlab Symbolic Toolbox.

2nd Order Topologies Employing Butterworth NTF

Both topologies produce noise shaping at their outputs with a few tones. These observed tones are mainly caused by the quantization operation and are analysed in Section 3.11. As both topologies have their coefficients within the same interval, the resulting DR and SNR values

have a perfect match. The SNR plots of both topologies for different OSRs and number paths are shown in Figure 56. It should be noted that all SNR figures in this thesis is plotted such that 0 dB indicates a signal to noise ratio of 1.



Figure 56: SNR Plot, 2nd-order Both Topologies, Butterworth NTF, Centre Frequency = 0.375, BW = 0.01, $OSR_{SP} = 32, OSR_{2P} = 64, OSR_{4P} = 128$

In the first place, this plot validates the node-equation method and its extension to VBP Σ - Δ modulators since a higher path number provides better SNR without increasing the individual path OSR. In addition, if the SNR estimation stated in Section 2.6 and calculated in [13] is recalled; every doubling of OSR is supposed to increase the SNR value by (6L + 3) dB where L is the order of the modulator. Increasing the path number without changing the path-OSR increases the overall OSR in multiples of P, where P is the path number.

The resulting SNR values depicted in Figure 56 are quite compatible with their theoretical SNR increment values. Note that an Lth-order VBP NTF's transition band roll-off is equivalent to the $(L/2)^{th}$ -order LP NTF. So for a 2nd-order VBP Σ - Δ modulator, an SNR increment of is expected. Refer to Table 3 and Table 4 on pages 65 and 66 for a compared SNR versus OSR summary.

As expected, the centre frequency sweep in other words the feedback coefficients' variations do not affect the performance as shown in Figure 57. Whereas the bandwidth sweep affects the obtained SNR up to 15 dBs and DR up to 14 dBs as shown in Figure 58. Normalised bandwidths larger than 0.04 do not increase the SNR more than 1-2dBs. Different bandwidth values can be chosen to increase the DR.



Figure 57: SNR vs Centre Frequency Plot, 2^{nd} -order 2-path CI-FBFF Topology, Butterworth NTF, BW=0.02, $OSR_{2P} = 16$



4th – 6th – 8th Order Topologies Employing Butterworth NTF

The output and SNR plots have an adequate matching between the CI-FBFF and CR-RFB for 4^{th} - 6^{th} - 8^{th} orders of the SP, 2-path and 4-path topologies. However, minor differences occur for different OSRs as shown in Figure 59. These variations are caused by the in-band tones.



Figure 59: SNR Plots, Both Topologies, Butterworth NTF, Different OSRs



Moreover, the SNR versus centre frequency relationship remains the same; the SNR value is not affected by the chosen centre frequency. On the other hand, the DR clearly changes with the bandwidth and the order of the modulator as depicted in Figure 60. This is due to the fact that larger bandwidths and higher-order VBP Σ - Δ modulators tend to possess larger coefficients. Hence the resulting VBP Σ - Δ modulators overload the quantizer for smaller input amplitudes, decreasing the DR as seen in Figure 61.



a)CI-FBFF, 2-path b)CR-RFB, 4-path Figure 60: DR vs BW Plot, Both Topologies, Butterworth NTF, BW=0.02, *OSR*_{2P,4P} = 16



Figure 61: SNR Plot, All Orders 4-path, Butterworth, CI-FBFF Topology, Butterworth NTF, Centre Frequency = 0.1, BW=0.02, *OSR*_{4P} = 16

Note that the overall OSR employed in Figure 61 is 16 and the SNR values for the 4-path topologies are depicted. The individual path OSR equals to 16/4. With such a low OSR, even the 2nd-order 4P CI-FBFF topology delivers an SNR of 39 dB and 8th-order 4P CI-FBFF topology accomplishes 65 dB SNR whilst providing DRs of 45 dB and 75 dB respectively.

In Figure 62, the SNR plot of the 6th-order CR-RFB topology for all path numbers are displayed. But in this plot, the overall OSRs for each SP, 2-path and 4-path topologies are chosen to equate each other. In other words, individual OSRs are chosen to be 64, 32 and 16 respectively. To do so, it is expected to obtain a well-matched SNR plot since all the SP, 2-path and 4-path topologies of same order have the same overall NTF. However, it is clearly seen that, even in the ideal-case simulations the 4-path topology SNR and DR values are smaller. This is due to the tones caused by the time-interleaving nature of the topology and is explained in Section 4.3. To summarize this concept, it must be stated that the TI idea depends on the perfect cancellation of the downsampling images. However before being cancelled at the upsampling circuitry, these images cause quantization tones, which might be folded back into the signal band.



Figure 62: SNR Plot, 6th-order, All Paths, Butterworth, CR-RFB Topology, Butterworth NTF, Centre Frequency = 0.1, BW=0.02, OSR_{SP} = 64, OSR_{2P} = 32, OSR_{4P} = 16

In Table 3 and Table 4, the summary of obtainable SNRs of Butterworth-based NTFs are given for a specific centre frequency, BW and OSR. The expected SNR increment for every doubling of OSR for different orders are as follows; 2^{nd} -order: 9 dB, 4^{th} -order: 15 dB, 6^{th} -order: 21 dB, 8^{th} -order: 27dB. Since the OSR is fixed in Table 3 and Table 4, these results are expected for every doubling of the TI path number. However, this increment cannot be obtained especially for higher-order Butterworth-based VBP Σ - Δ modulators. This performance degradation is attributed to the in-band tones acquired in the TI topologies and modelled in Section 4.3.

	SNR (dB)	DR (dB)
	(SP/2P/4P)	(SP/2P/4P)
2 nd	33/42/47	40/45/47
4 th	38/52/59	47/65/73
6 th	40/58/70	53/69/81
8 th	39/65/78	53/76/92

Table 3: CI-FBFF Topology SNRs and DRs, Butterworth, Centre Frequency =0.3, BW =0.02, $OSR_{SP} = 32, OSR_{2P} = 64, OSR_{4P} = 128$

	SNR (dB)	DR (dB)
	(SP/2P/4P)	(SP/2P/4P)
2 nd	33/42/47	38/44/47
4 th	38/52/60	46/65/73
6 th	40/58/70	51/70/79
8 th	40/58/78	52/70/94

Table 4: CR-RFB Topology SNRs and DRs, Butterworth, Centre Frequency =0.3, BW =0.02 , $OSR_{SP} = 32$, $OSR_{2P} = 64$, $OSR_{4P} = 128$

Chebyshev NTFs

Chebyshev filters produce amplitude fluctuations in exchange for a steeper transition band. These variations are called pass-band attenuations. To do so; poles lie on an ellipse whereas zeros are placed at the same frequency as shown in Figure 63. Poles closer to the zeros make the filter more aggressive; thus providing a narrower transition band when compared to the Butterworth filters of the same order.



Figure 63: Chebyshev Pole-Zero Map

Figure 64 shows a zoomed output plot of the 8th-order SP CI-FBFF topology when a Chebyshev filter is applied. Choosing higher pass-band attenuation eventually causes a more aggressive filter characteristic and raises the possibility of an unstable modulator.



Figure 64: Zoomed Output Plot, Chebyshev

The numerator and denominator coefficients of the desired Chebyshev based NTFs are obtained using the 'cheby1' function in Matlab. These coefficients are mapped to the desired topology by using the Matlab Symbolic Toolbox.

2nd-, 4th-, 6th-, 8th-Order Topologies Employing Chebyshev NTF

The resulting 2nd-order output plots, the SNR and DR ranges perfectly match as depicted in Figure 65.



Figure 65: SNR Plot, 2nd-order Both Topologies, Chebyshev NTF, Centre Frequency = 0.415, BW = 0.01, $OSR_{SP} = 32, OSR_{2P} = 64, OSR_{4P} = 128$

The 6th- and 8th-order SNR and DR performance have minor differences due to the coefficient interval mismatches between the topologies. These values are given in Table 5 and Table 6. In Table 6, a more relaxed Chebyshev NTF is designed whereas in Table 5, a more aggressive Chebyshev NTF is designed. As expected the NTF designed in Table 5 is capable of providing higher SNR and DR. The increases in SNR and DR are obtained by adjusting the passband attenuation rather than modifying the bandwidth. This feature is quite functional for wideband and high-resolution implementations. Note that the (6L+3) dB of SNR increase for every doubling of the OSR does not apply for Chebyshev NTF-based VBP Σ - Δ modulators for any path number.

	SNR (dB)	DR (dB)
	(SP/2P/4P)	(SP/2P/4P)
2 nd	33/40/43	49/50/48
4 th	43/55/60	57/69/75
6 th	50/69/76	62/78/88
8 th	57/70/84	75/81/105

Table 5: CI-FBFF Topology SNRs and DRs, Chebyshev, Centre Frequency =0.4, BW =0.01, PBA=2 dB, $OSR_{SP} = 32$, $OSR_{2P} = 64$, $OSR_{4P} = 128$

Table 6: CR-RFB Topology SNRs and DRs, Chebyshev, Centre Frequency =0.4, BW =0.01, PBA=0.5 dB, $OSR_{SP} = 32, OSR_{2P} = 64, OSR_{4P} = 128$

	SNR (dB)	DR (dB)
	(SP/2P/4P)	(SP/2P/4P)
2 nd	21/30/36	39/50/48
4 th	29/40/49	40/51/62
6 th	32/45/58	40/56/67
8 th	39/48/67	53/61/77



Figure 66: SNR Plot, 6^{th} -order, 4P, Both Topologies, Chebyshev NTF, BW = 0.01, PBA = 1dB, $OSR_{4P} = 128$

An interesting observation about Chebsyhev NTF-based VBP Σ - Δ modulators is that they perform quite higher SNR and DR at mid-band even if the pass-band attenuation and bandwidths are fixed. This observation is plotted in Figure 66. This could be attributed to the presence of more or higher amplitude tones at the output since some strong out-of band tones are observed. These limit cycle tones result in deeper notches for Chebyshev NTFs since the overall power at the output spectrum of a quantizer is fixed. This is due to the fact that the expected quantization noise is distributed over $\left[-\frac{\nu_s}{2}, \frac{\nu_s}{2}\right]$.



Figure 67: Output Plot, 6th-order, SP, CI-FBFF Topology, Chebyshev NTF, Centre Frequency = 0.25, BW = 0.01, PBA = 1dB

These tones may or may not be a problem depending on the implementation. If they are filtered and do not alias with other signals within the system, mid-frequency Chebyshev NTFs give higher resolution. Yet if they are not filtered and pass through other cascade circuits, they may be amplified and therefore degrade the quality of the output signal.

Note that these tones are also observed for other NTF types. An SNR increment as in the case of Chebyshev-based filters cannot be obtained at mid frequency. This is due to the fact that the limit cycle tones appear at close frequencies to the passband attenuations. In a way they help the passband frequency distortion hence steeper transition and higher SNRs are provided.

Inverse-Chebyshev NTFs

Inverse-Chebyshev filters distribute zeros over the stop-band to fluctuate the frequency response in order to provide narrow transition-bands. This time, attenuations are at the passband region. The numerator and the denominator coefficients of the designed Inverse-Chebyshev based NTF are obtained using the 'cheby2' function in Matlab. These coefficients are then mapped to the desired topology by using the Matlab Symbolic Toolbox.



Figure 68: Inverse-Chebyshev Pole-Zero Map



Figure 69: Zoomed Output Plot, Inverse-Chebyshev

2nd-, 4th-, 6th-, 8th-Order Topologies Employing Inverse-Chebyshev NTF

The SNR and DR values are given for all orders and all path numbers of the both CI-FBFF and CR-RFB topologies in Table 7-Table 8. Once again, these values are listed to give an idea of the Inverse-Chebyshev NTF-based topology performance. The (6L+3) SNR increment rule does not work for Inverse-Chebyshev filters as well.

	SNR (dB)	DR (dB)
	(SP/2P/4P)	(SP/2P/4P)
2 nd	41/51/57	38/42/46
4 th	48/63/73	64/74/84
6 th	35/52/63	52/69/78
8 th	30/41/48	42/58/73

Table 7: CI-FBFF Topology SNRs and DRs, Inverse-Chebyshev, Centre Frequency =0.46, BW =0.001,SBA = 80 dB, OSR_{SP} = 32, OSR_{2P} = 64, OSR_{4P} = 128

Table 8: CR-RFB Topology SNRs and DRs, Inverse-Chebyshev, Centre Frequency =0.4, BW = 0.01, SBA= 80 dB, OSR_{SP} = 16, OSR_{2P} = 32, OSR_{4P} = 64

	SNR (dB)	DR (dB)
	(SP/2P/4P)	(SP/2P/4P)
2 nd	34/40/47	31/38/40
4 th	29/46/53	43/60/70
6 th	UNSTABLE	UNSTABLE
8 th	32/US/US	50/US/US

The Inverse-Chebyshev filters accomplish the highest SNR and DR values when compared to the other NTFs. This is attributed to the distributed zeros over the signal-band. These zeros strongly reduce the overall in-band noise. This becomes more observable for higher-order Inverse-Chebyshev NTFs, as there are more zeros within the signal band. However, there are two main restrictions of Inverse-Chebyshev filter-based NTFs. Firstly, Inverse-Chebyshev filters operate in narrow-band designs; otherwise they drive the Σ - Δ modulator to unstable mode as shown in Table 8.

Secondly the choice of the OSR becomes vital. It was mentioned earlier that distributed zeros over the signal band reduces the in-band quantization noise hence delivering higher SNRs. If the NTF stop-band is not designed such that its zeros remain within the signal band, the Inverse-Chebyshev NTFs lose their attraction. As shown in Figure 70 for higher OSRs, the SNR does not increase within the same slope of the lower OSRs. This is due to the fact, if high OSR is chosen the output frequency response is stretched over a larger frequency scale causing NTF zeros to be placed out of the signal band.



Figure 70: SNR vs OSR Plot, Inverse-Chebyshev, 6th-order, Centre Frequency = 0.3, BW = 0.04, SBA = 80dB

Elliptical NTFs

Elliptical filters use both passband and stopband attenuations to accomplish the narrowest transition band with respect to the aforementioned filters. The numerator and denominator coefficients of the desired Elliptical based NTF are obtained using the 'ellip' function in Matlab. These coefficients are then mapped to the desired topology by using the Matlab Symbolic Toolbox. In Figure 71 pole-zero map of an Elliptical band stop filter is given. In addition, to depict the pass-band and stop-band attenuations, a zoomed output response of a 6th-order Elliptical band stop filter is given in Figure 72.



Figure 71: Elliptical Pole-Zero Map



Figure 72: Zoomed Output Plot, Elliptical

2nd-, 4th-, 6th-, 8th-Order Topologies Employing Elliptical NTF

See Table 9 and Table 10 for the SNR and DR performance of some Elliptical NTF-based VBP Σ - Δ modulators. The (*6L*+*3*) SNR increment rule does not work for Elliptical filters.

	SNR (dB)	DR (dB)
	(SP/2P/4P)	(SP/2P/4P)
2 nd	33/42/47	40/45/47
th	38/52/59	47/65/73
))	40/58/70	53/69/81
3 th	39/65/78	53/76/92

Table 9: CI-FBFF Topology SNRs and DRs, Elliptical, Centre Frequency =0.3, BW =0.02, PBA = 1dB, SBA=80dB, OSR_{SP} = 32, OSR_{2P} = 64, OSR_{4P} = 128

Table 10: CR-RFB Topology SNRs and DRs, Elliptical, Centre Frequency =0.1, BW =0.01, PBA = 1dB, SBA=80dB, OSR_{SP} = 16, OSR_{2P} = 32, OSR_{4P} = 64

	SNR (dB)	DR (dB)
	(SP/2P/4P)	(SP/2P/4P)
2 nd	14/20/23	33/39/41
4 th	19/28/34	33/43/61
6 th	23/31/32	32/44/33
8 th	24/32/36	35/46/55

The main benefit is that they can accomplish superior SNRs and DRs compared to Butterworth and Chebyshev-based $\Sigma - \Delta$ modulators of the same order for relatively wide narrow-band designs. This is due to the fact that they utilize both pass-band and stop-band attenuations in exchange for a narrower transition band. Moreover, the passband attenuation compensates the effect of the distributed zeros in the out-of-band gain. It lowers out-of-band gain thus providing more stable circuits when compared to Inverse-Chebyshev filters.

3.9 Filtering Comparison

The discrepancies in performance of the employed NTF filters were discussed and evaluated in the above section. This section sums up the main diversities.

In Figure 73 a narrow-band NTF's SNR plot is given whereas in Figure 74 an SNR plot for larger-band design is given for all filters. These plots are obtained by setting the cut-off frequencies of each filter as close as possible to each other in order to make a fair comparison of the topologies.



Figure 73: SNR Plot, All Filters, 4th-order, 2P, CR-RFB Topology, BW = 0.01, OSR = 64



Figure 74: SNR Plot, All Filters, 4th-order, 2P, CR-RFB Topology, BW = 0.04, OSR = 64

As seen in Figure 74, the Inverse-Chebyshev based NTF delivers negative SNR for all input amplitude values whilst the other filters provide positive SNR values. This is caused by the increased bandwidth. It can be concluded after several simulations performed within the thesis that the Inverse-Chebyshev NTF based VBP Σ - Δ modulators provide mostly unstable outputs when bandwidth is larger than 0.04 for the 4th-, 6th- and 8th-order topologies. For narrow-band applications the Inverse-Chebyshev based VBP Σ - Δ modulators are highly suitable as long as

the in-band zeros are kept within the signal band. Otherwise, there occurs an apparent SNR reduction as explained in Section 3.8 within the subtitle of Inverse-Chebyshev NTFs.

The Butterworth-based VBP Σ - Δ modulators can be chosen for relatively wide-band applications because they remain stable whilst providing high SNRs and DRs. This is attributed to the maximally flat out-of-band gain of the Butterworth NTFs. As detailed in Section 3.8 within the subtitle of Butterworth NTFs, Butterworth NTF based VBP Σ - Δ modulators they have delivered stable outputs up to a normalized bandwidth of 0.1.

In Section 3.8 the subtitle Chebyshev NTFs explains that the Chebyshev filters actually produce stable outputs with quite strong out-of-band tones at the centre frequency of 0.25. These out-of-band tones might be problematic for implementation purposes as they may alias with other signals within the overall receiver and/or they may be ear-detectable as well.

Finally, the Elliptical and Chebyshev NTFs seem to preserve stability for relatively large bandwidths whilst maintaining the same level of SNR and DR performance to that of their Butterworth counterparts.

3.10 Non-Idealities

The designed VBP Σ - Δ modulators' non-ideal behaviours are modelled in order to investigate their immunity to non-idealities. Since discrete-time analogue VBP Σ - Δ modulators are designed, the switched-capacitor circuits' non-ideal behaviours are examined.

Switched capacitor circuits are exposed to several non-idealities such as capacitance voltage dependency, capacitor mismatches and op-amp finite gain. One single non-ideality may result in multiple performance degradation. For instance, finite op-amp gain causes integrator leakage and integrator gain error. Therefore, non-ideal building blocks are investigated rather than addressing components' non-idealities.

Non-Ideal Integrator: The transfer function of a non-ideal integrator is given in (51) where g is the gain error and α is the leakage error.

$$H_{ID,non-id}(z) = \frac{gz^{-1}}{1 - \alpha z^{-1}}$$
(51)

Gain blocks model the gain error, g, and the leakage error, α , as shown in Figure 75.



Figure 75: Simulink Model of Non-Ideal Integrator

Non-ideal SP CI-FBFF Topologies

The designed SP CI-FBFF topologies are investigated by applying non-ideal integrators, as their main building blocks are delayless and delayed integrators. The non-idealities are applied to the Simulink model by generating random values for g and α . The random parameters g and α lie between [(-1 + e) (1 + e)] where e has a white distribution between a chosen percentage.

In Figure 76 the non-ideal 2nd-order SP CI-FBFF topology is given. In Figure 77 and Figure 78, SNR plots of non-ideal 2nd-order SP CI-FBFF topology are given when 10% gain errors and 10% leakage errors are applied.



Figure 76: Non-Ideal 2nd-order SP CI-FBFF Topology



a)g₁ applied

b) g_2 applied

Figure 77: SNR vs Gain Error, 2nd-order CI-FBFF Topology, Elliptical NTF, Centre Frequency =0.1, BW=0.02, PBA= 1 dB, SBA= 80 dB, OSR=16



The 10% error causes a maximum of 4 dB SNR variations for an OSR of 16. In the mean time, the stability is preserved. If higher OSR values are employed, the tolerable error percentage decreases because of the in-band noise-shaping degradation.

Various simulations showed that SP CI-FBFF topologies are quite sensitive to integrator nonidealities especially for orders above second. This is due to the fact that non-ideal integrators cause both pole and zero displacements. Therefore, both g and α may lead to poor SNR and/or unstable modulator. In Table 11, tolerable gain and leakage error percentages are summarized depending on the numerous simulations. These values are concluded such that the modulators do not become unstable and provide 4-8 dBs of SNR and 3-9 dBs DR variations.

Order	Error Percentage
2 nd	5 %
4 th	1 %
6 th	0.1 %
8 th	0.1 %

Table 11: Tolerable Error Percentage for SP CI-FBFF Topology

In Section 3.7, the coefficient analysis of the CI-FBFF topologies was given. In this analysis it was depicted that the coefficients of the CI-FBFF topology incredibly increases for centre frequencies above 0.25. This is caused by the cascaded nature of the topology. Hence cascaded integrators accumulate signals at the internal nodes, the feedback and feedforward coefficients will need to increase to balance this accumulation. This situation results in extra sensitivity to non-idealities for centre frequencies above 0.25. In other words, any attempt to impair the ideal behaviour of the topology causes significant variations for pole zero locations. Therefore the CI-FBFF modulators can easily become unstable for centre frequencies above 0.25.

Moreover, 4th- and 6th-orders of the SP CI-FBFF topology is quite sensitive to non-ideal behaviours of the integrators. As seen in Table 11, they tolerate only up to 0.1% of errors without going into the unstable region.

Non-Ideal Resonator: The transfer function of a non-ideal resonator is given in (52) where g is the gain error, α and β are the leakage errors. Note that β causes the centre frequency to shift whilst α results in a finite gain at the resonant frequency.

$$H_{ID,non-id}(z) = \frac{gz^{-1}}{1 + \beta K_1 z^{-1} + \alpha z^{-2}}$$
(52)

The behavioural simulation model of a non-ideal resonator is depicted in Figure 79.



Figure 79: Simulink Model of Non-Ideal Resonator

Non-ideal SP CR-RFB Topologies All orders of SP CR-RFB topologies are investigated with non-ideal resonators. The non-ideal 2nd-order SP CR-RFB topology is shown in Figure 80.



Figure 80: Non-Ideal 2nd-order SP CR-RFB Topology

Conversely to the CI-FBFF topologies, the CR-RFB topologies are quite immune to nonidealities from DC to Nyquist. Although the resonators are cascaded, the local feedbacks prevent signal accumulation therefore keeping the coefficients within a suitable interval for different centre frequencies. Once again, in Table 12 tolerable percentage errors are listed. These values have been arrived at to provide 4-8 dBs of SNR and 3-9 dBs DR variations as well as ensuring modulator stability.

Table 12: Tolerable Error Percentage for SP CR-RFB Topology

Order	Error Percentage
2 nd	20%
4 th	5%
6 th	5%
8 th	5%

Moreover, the resonant centre-frequency and resonant gain are formed directly by α and β respectively. The gain error g only affects the pole locations. So, as long as the poles do not cause instability, the gain error does not affect the output response much as shown in Figure 81.



Figure 81: Output Plots, 6th-order SP CR-RFB Topology, Inverse Chebyshev Filter

Adder Non-Idealities: The node-equation technique shares the samples of each node. Therefore, the resulting TI topologies do not require integrator/resonator blocks. Instead resonators and integrators are distributed over the channels just like the samples. The TI topologies only use delayless and delayed adders to perform integration and resonation operations.

The TI topologies are investigated using the non-ideal adder model given in (53). Note that a double delayer adder is depicted in Figure 82. Using a delayed or delayless adder does not change the defined errors; a_1 and a_2 .

$$Out_{adder,non-id}(z) = a_1 z^{-1} I n_1 + a_2 z^{-1} I n_2$$
(53)



Figure 82: Simulink Model of Non-Ideal Adder

Non-ideal TI Topologies: The TI topologies' tolerances to non-idealites are listed in Table 13-Table 16. Once again, these values have been evaluated such that the modulators provide 4-8 dBs of SNR and 3-9 dBs DR variations ensuring modulator stability.

Order	Error Percentage
2 nd	6%
4 th	3%
6 th	3%
8 th	3%

Table 13: Tolerable Error Percentage for 2P CI-FBFF Topology

Table 14: Tolerable Error Percentage for 4P CI-FBFF Topology

Order	Error Percentage
2 nd	5%
4 th	2%
6 th	2%
8 th	0.5 %

As seen, the TI CI-FBFF topologies are more immune to non-idealities when compared to their SP counterparts. This is because of the cross-connected structure of the TI topologies. These non-idealities do not affect the resonant frequency and/or centre frequency directly as is the case with SP topologies. The non-ideal adders inherently create tones and white noise therefore mitigating the in-band NTF degradation.

Table 15: Tolerable Error Percentage for 2P CR-RFB Topology

Order	Error Percentage
2^{nd}	8%
4 th	5%
6 th	5%
8 th	2%

Order	Error Percentage
2 nd	8%
4 th	5%
6 th	4%
8 th	1%

Table 16: Tolerable Error Percentage for 4P CR-RFB Topology

However, as will be discussed in Section 3.11.3, the mismatches among channels may create strong tones. These tones are called aliasing tones and tend to appear at frequencies $2\pi i/M$ where $i = 1, 2, \dots M - 1$ due to the imperfect cancellation of the downsampling images. They may cause instability or they can be folded back into the signal-band degrading the SNR and DR performance of the modulator.



Figure 83: Aliasing Tones, 6th-order 4P CR-RFB Topology, Chebyshev Filter

3.11 Quantization Tones

The tonality sources are already defined in Section 2.6.4. The frequency and amplitude of idle channel tones and limit cycle tones are hard to predict. However, there is another source of tonality seriously influencing the performance of VBP Σ - Δ modulators, i.e. the quantization tones. Basically the quantization of a sinusoid creates tones whose amplitudes and frequencies can be mathematically determined. Since the VBP Σ - Δ modulators are excited by single sinusoids, these tones may become observable in some cases. In this section, these tones are investigated for one-bit quantizers.

The truth is that D/A VBP Σ - Δ modulators are more likely to produce these tones since the finite wordlength also quantizes the signal. The quantization-tone performance of the D/A VBP Σ - Δ modulators is discussed in Section 4.3.

The quantizaton tones can be modelled using the sawtooth quantization noise model. However, since the quantization noise is highly input signal dependent, the actual power of these tones may become extremely difficult to predict. As a result, a mathematical model is developed for the tones that are caused by the quantization of the sinusoids. This mathematical model is developed for SP VBP Σ - Δ modulators and further extended for the TI VBP Σ - Δ modulators.

3.11.1 Sawtooth Quantization Noise Model, One-bit Quantizer, Single Sinusoid

The first step of the sawtooth quantization noise model is to use the additive noise model as already shown in the Section 2.1.

The second step is to define this additive quantization noise as an input signal dependent function. To have a better understanding of this theory, the simulated quantization noise of a sampled-sinusoid for a 1-bit quantizer is shown in Figure 84. In Figure 84-b the quantization noise clearly converges to a sawtooth signal with a frequency of $2\nu_c$ multiplied by a sinusoidal signal with a frequency of ν_c , where ν_c is the input signal frequency.



Figure 84: Quantization Noise of a Sinusoid for a 1-bit Quantizer, A=0.9 a) Input to the Quantizer, Output of the Quantizer and the Quantization Noise b)The Quantization Noise

Before getting into the mathematical analysis it must be mentioned that a mathematical model for the triangle wave is also built and compared with the simulation results and a decision is made that the sawtooth error model gives closer amplitude and frequency estimation.

The quantization noise is modelled accordingly in (54) where SW[n] represents the reverse sawtooth function.

$$Q_{err}[n] = SW[n]A_{\sin}\sin[2\pi(v_c n)]$$
(54)

An ideal sawtooth wave function can be written as the sum of sinusoids with integer multiples of the fundamental frequency which is $2\nu_c$ in this case.

$$SW[n] = \frac{A_{sw}}{2} - \frac{A_{sw}}{\pi} \sum_{k=1}^{\infty} (-1)^k \frac{\sin\left[2\pi(2\nu_c k)n\right]}{k}$$
(55)

The amplitudes are set such that the multiplication of two signals' amplitudes equal to Δ_Q . This is due to the fact that, if the sinusoidal signal sample value is zero, the quantizer will map it to the Δ_Q value. Moreover, the amplitude values given in (56) are determined after extensive simulations and comparisons of the provided mathematical model to the quantization noise obtained by simulations.

$$A_{sw} = \left(\Delta_Q - A - \varepsilon_1\right), \quad A_{sw} = \frac{\Delta_Q}{\left(\Delta_Q - A - \varepsilon_2\right)}$$
(56)

The trigonometric identity in (57) should be kept in mind to calculate the overall quantization noise in (60).

$$A_{1}\sin(a)A_{2}\sin(b) = \frac{A_{1}A_{2}}{2}\left(\cos(a-b) - \cos(a+b)\right)$$
(57)

 Δ_Q is the quantization step size and *A* is the input amplitude. Also, ε_1 and ε_2 are the input amplitude dependent errors that are assumed to have a white distribution over $[-\Delta_Q \Delta_Q]$. However, in this case they are assumed to be negligible and are approximated to zero for ease

of calculation. Depending on the input signal, a statistical approximation can be made. In addition, it is assumed that there is no overload of the quantizer.

$$Q_{err}[n] = \frac{A_{sw}}{2} A_{sin} \sin[2\pi\nu_c n] - \left(A_{sin} \sin[2\pi\nu_c n]\right) \left(\frac{A_{sw}}{\pi} \sum_{k=1}^{\infty} (-1)^k \frac{\sin[2\pi(2\nu_c k)n]}{k}\right)$$
(58)

Since the one-bit quantizer is examined; $\Delta_Q = 1$;

$$Q_{err}[n] = \frac{A_{sw}A_{sin}}{2} \sin[2\pi v_c n] - \left(\frac{A_{sw}A_{sin}}{2\pi} \sum_{k=1}^{\infty} (-1)^k \frac{\cos[2\pi (v_c(2k-1))n] - \cos[2\pi (v_c(2k+1))n]}{k}\right)$$
(59)

$$Q_{err}[n] = \frac{1}{2} \sin[2\pi v_c n] - \left(\frac{1}{2\pi} \sum_{k=1}^{\infty} (-1)^k \frac{\cos[2\pi (v_c(2k-1))n] - \cos[2\pi (v_c(2k+1))n]}{k}\right)$$
(60)

Hence, the sawtooth signal is multiplied by a sinusoid, the process of calculating the quantization noise signal is similar to double-sided AM modulation. It is well known that, if a sinusoidal signal with a frequency of f_1 is AM-modulated by another sinusoid with a frequency of f_2 , the resulting tones will be at $(f_2 - f_1)$ and $(f_2 + f_1)$. Since $f_1 = 2f_c$, $4f_c$, $6f_c$... and $f_2 = f_c$, the resulting tones will be at $[...(6-1)f_c, (4-1)f_c, (2-1)f_c, f_c, (2+1)f_c, (4+1)f_c, (6+1)f_c$...].

This mathematical model in (59) has given some knowledge to estimate the frequencies and amplitudes of the tones. In order to visualize the derived mathematical model, an example of the sawtooth quantization noise model is given below. A sinusoid with a normalized frequency of 0.12, given in (61), is directly applied to a 1-bit quantizer. This results in a quantization noise of (62). The simulation results are given in Figure 85 for two different input amplitudes. The output plots of the quantization noise validate the proposed mathematical model. The quantization noise harmonics of a sinusoid are not input amplitude dependent. It should be remembered that tones beyond 0.5 are folded back and added to the already existing tones within the range of [0, 0.5] as their frequency is mapped to v_N -[v_N + 0.5]. The operator [x] represents the largest integer less than or equal to [x] and v_N is the normalized frequency of the signal.

$$S_{input} = A \sin\left[2\pi (0.12)n\right] \tag{61}$$

$$Q_{err}[n] = \frac{1}{2} \sin\left[2\pi (0.12)n\right] - \left(\frac{1}{2\pi} \sum_{k=1}^{\infty} (-1)^k \frac{\cos\left[2\pi ((0.12)(2k-1))n\right] - \cos\left[2\pi ((0.12)(2k+1))n\right]}{k}\right)$$

(62)

The simulated and mathematical results are compared in Figure 85-Figure 86. It should be noted that only fundamental harmonic amplitude is misleading as the sawtooth error model calculates the fundamental signal amplitude using fixed value of 0.5. However, the fundamental harmonic's frequency is actually where the input signal frequency is. So there is no point in calculating it correctly. On the other hand, other harmonics of the quantization noise is calculated such that the maximum difference occurring between the calculated and simulated harmonics is 5 dBs.



Figure 85: Simulated Results for Quantization Noise of a Sinusoid, $\nu_c = 0.12$



Figure 86: Calculated Results for Quantization Noise of a Sinusoid, $\nu_c=0.12$

The sawtooth quantization noise model can be summed up as follows:

- 1. The expected high tones in a Σ - Δ modulator resulting from the quantization of a sinusoid, not the limit cycle tones, can be calculated in terms of their frequency and amplitude.
- 2. The quantization harmonics are not input-amplitude dependent but are dependent on the quantization step. Only the fundamental harmonic's amplitude is determined by the input signal and cannot be calculated by the developed mathematical model.
- 3. These tones can be whitened by dithering, especially for multi-level quantizers. Due to their smaller Δ_Q s, multi-bit quantizers result in lower amplitude tones when compared to the 1-bit quantizer as expected.
- 4. At some particular frequencies such as 0.25, 0.125, 0.375...etc dithering may not work sufficiently for all the tones. This is because harmonics of the sawtooth signal are mapped and added to each other at the same frequencies resulting in higher amplitude tones as depicted in Figure 87.

On the other hand, utilizing a variable centre frequency exhibit more but smaller amplitude tones that can be reduced sufficiently when dithering is employed. This makes the proposed VBP Σ - Δ modulator topologies more attractive compared with their mid-band counterparts.



a) Simulated b) Calculated Figure 87: Quantization Noise of a Sinusoid $\nu_c = 0.375$

5. Input frequencies, whose frequencies are irrational, result in a higher number of tones since the harmonics of the sawtooth signal are not folded back to the same frequencies. Therefore, the resulting quantization tones have lower amplitudes as shown in Figure 88b. This is due to the fact that irrational frequency tones are mapped close to each other and may not be sufficiently suppressed within the signal-band, thus resulting in significant SNR reduction.



Figure 88: Simulated Quantization Noise of a Sinusoid, $\nu_c = 0.17$

Moreover, the presented mathematical model becomes less accurate when the input frequency is irrational. This is due to the fact that, the estimated sawtooth signal is deterioted when the input signal has an irrational frequency as depicted in Figure 88a. Strong in-band tones are still presented with the sawtooth quantization noise model although tones close to DC and Nyquist amplitude values are quite different from the simulation results. Yet their frequencies are calculated precisely as seen in Figure 89.



Figure 89: Calculated Quantization Noise of a Sinusoid, $\nu_c = 0.17$

6. The presented quantization noise mathematical model can be extended to any input signal that can be decomposed to a Fourier series. Every single harmonic should be treated as a single input to the quantizer and the resulting quantization noise will be a sum of the individually calculated quantization noise.

3.11.2 SP VBP Σ - Δ Modulators Excited by Single Sinusoids

This section presents some output plots for the designed SP VBP Σ - Δ modulator topologies. These Σ - Δ modulators are excited by single sinusoids to characterise and evaluate the effects of quantization tones. Since A/D VBP Σ - Δ modulators work in the discrete analog domain, the only source of quantization tones is the quantizer. Note that in the digital domain the finite wordlength of each path causes the quantization tones as will investigated in Section 4.3.

In A/D modulators, a sampled sinusoid is firstly filtered by the STF and then quantized. The quantization tones are shaped by the NTF hence they pass through the feedback path. Depending on extensive simulations, observations and calculations, it can be concluded that the frequencies of expected quantization tones of a VBP Σ - Δ modulators can be calculated whereas the amplitude of these tones depend on the NTF and STF of the modulator. Moreover, dithering may whiten some of the harmonics.

In Figure 90a, the output plot of a 4th-order SP-FBFF topology is depicted where the out-of band quantization tones are observed. The in-band tones are mostly suppressed by the NTF. In Figure 90b, the dither noise amplitude is increased to randomise the strong in-band tones at the expense of increased noise floor. Despite this, the second and third harmonics of the developed mathematical model are still present.



Figure 90: Output Plot, SP CI-FBFF 4th-order, Butterworth NTF, Centre Frequency = 0.11, BW=0.02

In Figure 91, a centre frequency of 0.375 is chosen and applied to the CR-RFB 8th-order topology. Although a 6th-order NTF is employed, the only occurring harmonic cannot be diminished. Since it is way out of the signal band, it does not cause any performance degradation.



Figure 91: Output Plot, SP CR-RFB 6th-order, Elliptical NTF, Centre Frequency = 0.375, BW=0.04

3.11.3 TI VBP Σ-Δ Modulators Excited by Single Sinusoids

The TI topologies are likely to produce quantization tones due to the downsampling operation. In addition, if mismatches occur, these tones are not cancelled by the upsamplers. If downsampler's mathematical model is recalled;

$$X_d\left(e^{j\omega}\right) = \frac{1}{D}\sum_{i=0}^{D-1} X\left(e^{j\left(\frac{\omega}{D} - \frac{2\pi i}{D}\right)}\right)$$
(63)

Due to the frequency scaling property given in (64), the quantization noise of a sinusoid is recalculated in (65) for a P-path TI structure.

$$f(at-\theta) = \frac{1}{|a|} F\left(\frac{\omega}{a}\right) e^{-\theta/a}$$
(64)

$$Q_{err,Downsample}[n] = \sum_{i=0}^{P-1} \left\{ \frac{\frac{1}{2} \sin[2\pi P \nu_c n - 2\pi i]}{-\frac{1}{2\pi} \sum_{k=1}^{\infty} (-1)^k} \left(\frac{\frac{\cos[2\pi (P \nu_c (2k-1))n - 2\pi i]}{k}}{-\frac{\cos[2\pi (P \nu_c (2k+1))n - 2\pi i]}{k}} \right) \right\}$$
(65)

These tones are shaped by the NTF and thereafter pass through the upsamplers. Harmonics will be attenuated and folded back to the signal band by the upsamplers. If the upsampler's mathematical model is recalled and combined with the frequency scaling property of Fourier analysis, the expression given in (69) is obtained where P is the path number.

The sawtooth quantization noise mathematical model given in (69) is developed for a single downsampled and upsampled sinusoidal signal that is directly applied to a quantizer. As a result, each path of the TI topologies produce the resulting tones as they are all stimulated by downsampled sinosidal signals. Note that, the amplitude of the quantization tones are shaped by the overall NTF for a TI Σ - Δ modulators. Therefore, the amplitudes in (69) are replaced by *B* and *C* representing the NTF shaped amplitudes.

$$X_u(e^{j\omega}) = X(e^{j\omega U})$$
(66)

$$X_u(n) = \frac{1}{|U|} X_u\left(\frac{n}{U}\right) \tag{67}$$

$$Q_{err,Upsample}[n] = \sum_{i=0}^{p-1} \begin{cases} \frac{1}{2P} \sin\left[\frac{2\pi P v_c n - 2\pi i}{P}\right] \\ -\left(\frac{1}{2\pi P} \sum_{k=1}^{\infty} (-1)^k \left(\frac{\cos\left[\frac{2\pi (P v_c(2k-1))n - 2\pi i}{P}\right]}{k}\right)\right) \\ -\frac{\cos\left[\frac{2\pi (P v_c(2k+1))n - 2\pi i}{P}\right]}{k}\right) \end{cases}$$
(68)
$$Q_{err,TI}[n] = \sum_{i=0}^{p-1} \begin{cases} B \sin\left[2\pi v_c n - \frac{2\pi i}{P}\right] \\ -\left(\sum_{k=1}^{\infty} (-1)^k \left(\frac{\cos\left[2\pi (v_c(2k-1))n - \frac{2\pi i}{P}\right]}{k}\right)\right) \\ -\frac{\cos\left[2\pi (v_c(2k+1))n - \frac{2\pi i}{P}\right]}{k}\right) \end{cases}$$
(69)

The quantization tones seen in the SP topologies are also observed in TI topologies. Yet the shifted copies of these tones are also observed. More importantly, they are folded back into the signal band and can cause in-band tones. Another fundamental problem is that the strongest quantization harmonic will not occur at the input frequency where the NTF zeros are placed. Therefore, of course depending on the input frequency, some very strong tones can be observed within the signal band. Specifically the odd-path numbered TI topologies may produce odd frequency multiples that cannot be mapped onto each other but rather harmonics are distributed over the frequency band. Hence this is the main reason to design even-path numbered TI topologies.

The output plots of some TI topologies are given below where the quantization tones can be clearly observed whereas in their SP counterparts these tones do not occur. This basic phenomenon has not been mentioned in any previous publications to the best knowledge of the author. In [92] these tones are observed and referred to as limit-cycle tones caused by the path
mismatches. However, it is proved that these tones can be observed even for the ideal-case simulations.



Figure 92: Output Plot, 2P and SP CI-FBFF 4th-order, Chebyshev NTF, Centre Frequency = 0.23, BW=0.02



Figure 93: Output Plot, SP and 8P CR-RFB 8th-order, Inverse-Chebyshev NTF, Centre Frequency = 0.42, BW=0.002

3.12 Conclusion

This chapter constitutes the backbone of the overall thesis. Firstly, the motivations to build up A/D VBP Σ - Δ modulators are detailed whilst inspecting the conventional receiver architectures; superheterodyne and homodyne receivers. It is shown that VBP Σ - Δ modulators are promising for future technologies as they eliminate the mixer and passive circuitry. Moreover, they provide flexibility hence making it easier to build tunable architectures.

A generalized NTF of a VBP Σ - Δ modulator was developed from filter principles. By utilizing the given NTF and mapping it to an appropriate Σ - Δ modulator topology, a generalized SP VBP Σ - Δ modulator could be constructed. However, the high OSR requirements of Σ - Δ modulation technique needed to be compensated. So, the node-equation method was introduced and applied to VBP Σ - Δ modulators. The node-equation method is an easy-to-apply mathematical technique. Moreover, it is not coefficient-dependent like the polyphase decomposition technique. It simplifies the process of designing and building generalized TI VBP Σ - Δ modulators. Finally, a step-by-step design technique to construct generalized TI VBP Σ - Δ modulators was provided and supported with an example as presented in Section 3.3.6.

Generalized 2-path and 4-path TI VBP Σ - Δ modulators of 2nd-, 4th-, 6th- and 8th-orders were developed using this design technique. Two different loop-filter topologies were selected to map the NTF. The CI-FBFF utilizes the integrators whereas the CR-RFB utilizes the resonators as their building blocks. Therefore their ideal and non-ideal simulation results give different responses.

These two generalized topologies are evaluated in terms of their coefficients, stability, SNR and DR performance as well as immunity to non-idealites. It is concluded that the CI-FBFF topology is not suitable to implement for frequencies above 0.25 because of its very large feedback and feedforward coefficients. However, the CR-RFB topology's feedback and feedforward coefficients are within a realizable interval. Thereby, the CR-RFB topology provides good resolution and is suitable for implementation for any centre frequency.

On the other hand, the non-ideality simulations revealed that the node-equation technique and resulting TI topologies are quite immune to non-idealities. They can provide their resolution and stability up to the given percentage errors listed in Table 13-Table 16. This is the result of their cross-connected architectures. The resonant behaviour of the overall TI structure is not determined by individual adders. All adders contribute to the NTF zeros and poles' locations. Hence, the presence of a highly non-ideal adder can be compensated by other adders. Therefore another advantage of building TI topologies has been established.

Finally, the quantization tones that are produced by a quantizer that is excited by a single sinusoid were mathematically modelled. The developed mathematical model uses the sawtooth error model and extends it to calculate the amplitude and frequency of these tones. The simulation tones and calculated tones were compared and the findings of this model were listed in Section 3.11. The SP topologies and the TI topologies are investigated in terms of their quantization tones. It has shown that the TI topologies were more likely to produce the quantization tones.

Chapter 4. D/A TI VBP Σ - Δ Modulators

D/A Σ - Δ modulators are basically digital data processors, in which a multi-bit digital data stream is converted to coarse digital data. By doing so, the actual DAC circuitry saves hardware, area and power [93], [94]. In this chapter, one-bit D/A Σ - Δ modulators are investigated. Hence the required DAC circuitry is represented by a comparator that maps the one-bit data to a positive or negative supply voltage [95], [96]. Moreover the expensive trimming and/or calibration circuit is excluded [97], [98]. The block diagram of the intended D/A converter scheme is given in Figure 94.



Figure 94: D/A Converter Block Diagram

The interpolation filter mainly accommodates the signal rate to the desired oversampling-ratio and suppresses the spectral images. Then the VBP Σ - Δ modulator converts the N-bit digital data to one-bit digital data. Thereafter the DAC, in other words the comparator, produces the analog signal. This signal is either band-pass or low-pass filtered regarding its-band.

The subject of interest in this study is the single-bit VBP Σ - Δ modulators in which the analog design challenges still exist such as high OSR, BW limitation, stability and tonality. Therefore generalized D/A VBP Σ - Δ modulators are designed and implemented that are capable of achieveing noise shaping at any desired center frequency and bandwidth with a choice of filter types: Butterworth, Chebyshev, Inverse-Chebyshev and Elliptical. As mentioned, these NTFs ensure stability whilst giving the ability to control the bandwidth, pass-band and stop-band attenuations. This in return helps to tune the Σ - Δ modulator to the desired signal band without the need for extra filter blocks.

Once again the node-equation method is applied to the designed SP topologies in order to overcome the high OSR requirement.

The key feature of the VBP Σ - Δ modulator topologies built in this section is that they use delay elements as their main building blocks rather than integrators. Integrators are especially advantageous in A/D Σ - Δ modulators for two reasons. First of all; they are easy to build using switched capacitor circuits. Secondly, they supress high frequency noise, therefore resulting in more stable circuits. However, they cause signal accumulation through the internal nodes. The accumulation in the internal path increases the overall hardware of the digital implementations. This is due to the fact that the accumulated internal path requires an increased number of bits. In addition, the pitched analog signals do not occur in digital circuits, thus making integrators less attractive. Delay blocks not only prevent signal accumulation but are also easy to build in digital circuits requiring less power and area.

In this section; two fundamental Σ - Δ modulator topologies, ErrorFeedback (EF) and OutputFeedback (OF), are built in MATLAB, implemented in VHDL and synthesized on the Xilinx® SpartanTM-3 Development Kit. The step taken further is that the hardware consumption of the OF topology is significantly decreased as will be covered in Section 4.2.1.

4.1 Output Feedback and Error Feedback Topologies

The block diagrams of the EF and the OF topologies are illustrated in Figure 95 and Figure 96 respectively.



Figure 95: The OF Topology



Figure 96: The EF Topology

The NTFs and the STFs of both topologies are given in (70)-(73) where H(z) is the transfer function of the loop-filter. Note that the NTFs and the STFs are calculated using the white-noise approximation.

$$STF_{OF}(z) = \frac{H(z)}{1 + H(z)}$$
(70)

$$STF_{EF}(z) = 1 \tag{71}$$

$$NTF_{OF}(z) = \frac{1}{1+H(z)}$$
(72)

$$NTF_{EF}(z) = 1 - H(z) \tag{73}$$

The EF topology filters the quantization noise, i.e. the quantization noise. The quantization noise is obtained by subtracting the quantization input from its output in the block diagram. However in the implementation, the quantization and subtraction operations are performed by the truncation of the quantizer's input signal. The MSB is the output of the Σ - Δ modulator and the remaning LSBs are fedback to the loop-filter. That is why in EF topologies, the quantization noise is usually referred to as the truncation error. In addition, the EF topology has another advantage when implemented digitally; it does not cause any signal corruption due to its unity STF.

On the other hand, the OF topology is generally chosen in analog implementations, since analog non-idealities of the loop-filter are not directly added to the input as in the EF topology. However, this study reveals that for a digital VBP Σ - Δ modulator implementation, the OF topology is more beneficial when implementation limitations such as internal data-path number and propagation delays arise.

4.2 Loop-Filter Design

The NTF is chosen to provide generalized Σ - Δ modulators that can accomplish noise shaping from DC to Nyquist for various filter types such as Butterworth, Chebyshev, Inverse-

Chebyshev and Elliptical. The generalized transfer function of an Lth-order generalized Band Stop (BS) NTF is given in (74).

$$NTF_{gen}(z) = \frac{\prod_{k=1}^{L/2} \left(1 - 2b_k z^{-1} + z^{-2}\right)}{1 + a_1 z^{-1} + \dots + a_{L-1} z^{-L+1} + z^{-L}}$$
(74)

As given in (74), the numerator is an L^{th} -order resonator in which b_k determines the resonant frequency. The 4th-order resonator transfer function is calculated in (75) and (76).

$$H_{res,4^{th}}(z) = \left(1 - 2b_1 z^{-1} + z^{-2}\right) \left(1 - 2b_2 z^{-1} + z^{-2}\right)$$
(75)

$$H_{res,4^{\pm}}(z) = \left(1 - 2(b_1 + b_2)z^{-1} + (1 + 4b_1b_2)z^{-2} - 2(b_1 + b_2)z^{-3} + z^{-4}\right)$$
(76)

It can be clearly seen in (76) that the second and fourth coefficients of a 4th-order resonator transfer function are symmetrical. Moreover, the fifth coefficient is unity. To simplify the equation, the coefficients of (76) are renamed as in (77). The transfer function of an L^{th} -order resonator is given in (78).

$$H_{res,4^{in}}(z) = \left(1 + c_2 z^{-1} + c_3 z^{-2} + c_2 z^{-3} + z^{-4}\right)$$
(77)

$$H_{res,L^{th}}(z) = \left(1 + c_2 z^{-1} + c_3 z^{-2} + \dots + c_{(L/2)+1} z^{-(L/2-1)} + \dots + c_3 z^{-(L-2)} + c_2 z^{-(L-1)} + z^{-L}\right)$$
(78)

To sum up, the L^{th} and the 1st coefficients of an L^{th} -order resonator are unity and the remaining coefficients are symmetrically equal. This symmetry can be very useful in the design of the loop-filter since it eliminates half of the multiplication blocks. The NTFs of the OF and EF topologies are given in (72) and (73) respectively. The NTFs are re-calculated in (79) and (80), where the transfer functions of the loop-filters are replaced by num_H/den_H .

$$NTF_{OF}(z) = \frac{1}{1 + \frac{num_H}{den_H}}$$

$$NTF_{OF}(z) = \frac{den_H}{den_H + num_H}$$
(79)

$$NTF_{EF}(z) = 1 - \frac{num_H}{den_H}$$

$$NTF_{EF}(z) = \frac{den_H - num_H}{den_H}$$
(80)

(79) reveals that the denominator of the loop-filter directly determines the numerator of the NTF_{OF} . To take the advantage of the symmetrical numerator of the NTF, a loop-filter must be designed such that the denominator coefficients are individually determined by the loop-filter's coefficients.

4.2.1 FBFF TDA Loop-filter

In Figure 97, the FeedBack and FeedForward (FBFF) Time Delay and Accumulate (TDA) topology, commonly known as the Direct Form-1 IIR filter topology is shown. With properly chosen coefficients, both EF and OF topologies can perform Butterworth, Chebyshev, Inverse-Chebyshev and Elliptical filters. Thus these topologies provide flexibility by enabling designers to specify the centre frequency, signal bandwidth as well as the passband and stopband attenuation parameters.



Figure 97: 4th-order FBFF TDA Loop-filter

The key point of the designed FBFF TDA topology is that the delay blocks are distributed between adders and multipliers to prevent long combinational delays and to balance the registering [99]. These delayers even the in-coming signal timings to each adder block. As

long as the multiplication operations are performed within a clock interval there is no path delay for the loop-filter.

The transfer function of the FBFF TDA loop-filter is given in (81). The symbolic NTF formulae of the OF and EF topologies are calculated in (82) and (83) respectively.

$$H(z) = \frac{K_1 z^{-1} + K_2 z^{-2} + K_3 z^{-3} + K_4 z^{-4}}{1 + L_1 z^{-1} + L_2 z^{-2} + L_3 z^{-3} + L_4 z^{-4}}$$
(81)

$$NTF_{OF}(z) = \frac{1 + L_1 z^{-1} + L_2 z^{-2} + L_3 z^{-3} + L_4 z^{-4}}{1 + (K_1 + L_1) z^{-1} + (K_2 + L_2) z^{-2} + (K_3 + L_3) z^{-3} + (K_4 + L_4) z^{-4}}$$
(82)

$$NTF_{EF}(z) = \frac{1 + (L_1 - K_1)z^{-1} + (L_2 - K_2)z^{-2} + (L_3 - K_3)z^{-3} + (L_4 - K_4)z^{-4}}{1 + L_1z^{-1} + L_2z^{-2} + L_3z^{-3} + L_4z^{-4}}$$
(83)

For a 4th-order SP Σ - Δ modulator there are supposed to be 4 feedback multiplications for the OF topology. But using the FBFF TDA loop-filter results in only 2 feedback multiplications. Note that in (82), the numerator of the *NTF*_{OF} is determined directly by the feedback coefficients. Since the numerator of an NTF of a VBP Σ - Δ modulator is composed of resonators as analysed in Section 4.2.1. L_4 is unity and L_1 equals to L_3 . The designed 4th-order 2P OF Σ - Δ modulator topology saves 4 multiplication blocks and the 4P OF Σ - Δ modulator saves 8 multiplication blocks. To generalize this rule; an Lth order P-Path OF Σ - Δ modulator utilizing the FBFF TDA loop-filter saves $P \times L/2$ multiplication blocks.

The 4th-order SP OF and EF topologies are depicted in Figure 99. Their 2P and 4P counterparts are also built using the node-equation method and illustrated in Figure 100 and Figure 101 respectively.





Figure 99: SP 4th-order EF Topology



Figure 100: 2P 4th-order EF Topology



Figure 101: 2P 4th-order EF Topology

4.2.2 Particular Frequencies

Although significant savings have been achieved in the feedback multiplication blocks of the OF topology, there are some particular frequencies at which designers can even save more. These particular normalized frequencies and their corresponding feedback coefficients are listed in Table 17 for the OF topology. When $\nu = 1/6$ and $\nu = 1/3$, the multiplication blocks that perform the L₁ and L₃ multiplications are replaced with a shifter. When $\nu = 1/4$, there are no multiplication blocks, but only one shifter is needed for the L₂ multiplication.

Normalized	L ₁	L ₂	L ₃	L_4
Frequency				
1/6	$-2.0009 = \sim -2$	3.0009 = ~3	$-2.0009 = \sim -2$	1
1/4	0	2	0	1
1/3	$-2.0009 = \sim -2$	$3.0009 = \sim 3$	$-2.0009 = \sim -2$	1

 Table 17: Particular Frequencies and Corresponding Loop-filter's Feedback Coefficients for the OF Topology

4.3 Quantization Tones

In Section 3.11 the quantization tones of a single-bit quantizer were mathematically modeled for a single sinusoidal input signal. In Sections 3.11.2 and 3.11.3 this mathematical model was evaluated for SP and multi-path VBP Σ - Δ modulators respectively. The main findings of this analyses were that the expected tone frequencies could be determined if the input signal frequency were known. The amplitude of these tones depends on the loop-filter, order of the Σ - Δ modulators as well as the frequency of the input signal. However, dither can be used to reduce these tones, but this will be at the expense of an increased noise level.

For the D/A VBP Σ - Δ modulators these characteristics are still applicable. However, D/A VBP Σ - Δ modulators tend to be more tonal. The finite wordlength also behaves as a quantization operation as it truncates the input signal and internal signals. Therefore quantization tones need to be investigated.

The finite wordlength effect is generally referred to as truncation noise. In [100], [101], finite wordlength errors are assummed to be uniformly distributed, white and uncorrelated. They are added to the existing path whenever a truncation occurs. This assumption serves quite well for internal path, as they are added up together and/or multiplied to each other. Moreover, if the finite worlength effect is thought to be a quantization operation, these internal quantizers are not excited by a single sinusoid. Therefore the quantization noise converges to white noise. In this sense, D/A VBP Σ - Δ modulators tend to have an increased noise floor level when compared to their A/D counterparts. Moreover, for the TI topologies where the number of multipliers and adders is increased, the overall noise floor is increased as well when compared to their SP counterparts. In addition, the finite wordlength of coefficients cause NTF shaping

degradation. The elevated noise floor and degraded noise-shaping can be seen in Figure 102 and Figure 103.



Figure 102: Zoomed Output Plots, D/A 2P 6th-order Elliptical NTF, OF Topology, Centre Frequency = 0.2, BW = 0.01, SBA = 1 dB, PBA = 60 dB



Figure 103: Zoomed Output Plots, D/A 2P 6th-order Elliptical NTF, EF Topology, Centre Frequency = 0.2, BW = 0.01, SBA = 1 dB, PBA = 60 dB

The white noise produced by the finite wordlength of the coefficients diminishes the discrete quantization tones caused by the quantizer.

Note that the EF and OF topologies result in different responses when non-ideal behaviours, finite wordlengths of input signal and coefficients, are applied. The finite wordlength effect on the overall performance of the EF and OF topologies is examined in Section 4.5.1.

Another source of quantization tones is the finite wordlength input signal itself. The mathematical model given in (65) can be recalled and modified for an N-bit fractional finite wordlength acting as a quantizer. If an N-bit fractional resolution is chosen, M equals to 2^{-N} .

$$Q_{err,TI}[n] = \sum_{i=0}^{P-1} \left\{ -\left[\frac{M}{2\pi} \sin\left[2\pi v_{c}n - \frac{2\pi i}{P}\right] - \left[\frac{M}{2\pi P} \sum_{k=1}^{\infty} (-1)^{k} \left(\frac{\cos\left[2\pi \left(v_{c}(2k-1)\right)n - \frac{2\pi i}{P}\right]}{k} \right) - \frac{\cos\left[2\pi \left(v_{c}(2k+1)\right)n - \frac{2\pi i}{P}\right]}{k} \right) \right] \right\}$$
(84)

The input signal wordlength is chosen to be 16 bits (1-bit for the sign and 15-bits for the decimal part). The resulting quantization tones are displayed in Figure 104. As clearly seen, the resulting quantization tones have quite low amplitudes and can easily be masked by the overall quantization noise produced within the modulator and shaped by the NTF. Note that these tones act as input signals to the Σ - Δ modulator and are shaped by the STF. Yet again, the visible tones at the output are produced by the one-bit quantizer.



Figure 104: Finite Wordlength Input Signal and Its Quantization Tones, Freq.=0.27, Amplitude = 0.5, 15bit Fractional Resolution

4.4 Behavioural Level Simulations

The 2^{nd} -, 4^{th} - and 6^{th} -order D/A VBP Σ - Δ modulators are built using the FBFF TDA loop-filter for both the EF and OF topologies. Moreoever their 2P and 4P TI counterparts are designed using the node equation method. These structures are shown in Appendix C and Appendix D.

4.4.1 Coefficient Comparison

The OF and EF topologies' feedback and feedforward coefficients have comparable amplitudes for a chosen filter type and filter specifications. This is due to the fact that the delayers do not produce any accumulation, hence the feedback and feedforward coefficients are solely dependent on the filter coefficients as given in (82) and (83).

The relationship between these coefficients versus the normalised centre frequency is illustrated Figure 105. Similarly, the relationship between these coefficients versus the normalised bandwidth is shown in Figure 106. The order of the magnitude as well as the actual values of these coefficients can be implemented with relative ease



a) The EF Topology



b) The OF Topology

Figure 105: Feedback and Feedforward Coefficients versus Centre Frequency, Elliptical NTF, PBA= 2dB, SBA = 60dB



Figure 106: Feedback and Feedforward Coefficients versus Bandwidth Elliptical NTF, PBA= 2dB, SBA = 60dB

4.4.2 Filtering Performance and Comparison

Behavioural level simulations are performed for the ideal case where finite wordlength is only applied to the input signal. As expected, the behavioural level simulations results are highly similar to the A/D behavioural level simulations given in Section 3.4. To demonstrate some examples, the output plots of some D/A VBP Σ - Δ modulators for different path numbers and orders are illustrated in Figure 107-Figure 109. The quantization tones can be clearly observed for these ideal case simulations especially for the TI topologies.



Figure 107: 2nd- Order SP OF Topology, Butterworth NTF, Centre Frequency = 0.35, BW=0.01



Figure 108: 4th- Order 2P EF Topology, Chebyshev NTF, Centre Frequency = 0.35, BW=0.01, PBA=1 dB



Figure 109: 6th- Order 4P 0F Topology, Inverse Chebyshev NTF, Centre Frequency = 0.1, BW=0.01, SBA=60 dB

In Figure 110, the SNR plots for different orders and path numbers are shown. The Inverse-Chebyshev filter yields the highest SNR due to the distribution of its NTF zeros across the entire signal band region.



Figure 110: SNR Plot, 6th-order EF Topology

4.5 Implementation

The designed 4th-order digital VBP Σ - Δ modulators are implemented in VHDL and synthesized on the Xilinx® SpartanTM-3 Development Kit. The universal clock of the Xilinx® SpartanTM-3 Development Kit is 66 MHz. Hence, the resulting clock frequencies are 33MHz and 16.5MHz for the 2P and 4P modulators' individual paths respectively. In Table

18, the chosen parameters for the designed modulators are listed. The resulting feedforward and feedback coefficients are given in Table 19 and Table 20.

Design Specs.	Butterworth	Chebyshev	Inv. Chebyshev	Elliptical
Centre-Frequency	0.2	0.2	0.2	0.2
Bandwidth	0.02	0.02	0.004	0.02
PBA	-	1 dB	-	1 dB
SBA	-	-	60 dB	80 dB

Table 18: Chosen Design Parameters

Table 19: Resulting Feedforward Coefficients

Loop-Filter		K ₁	<i>K</i> ₂	<i>K</i> ₃	K ₄
Butterworth	OF	0.0549	-0.2099	0. 55	-0.1628
	EF	0.0549	-0.2099	0.1557	-0.1628
Chebyshev	OF	0.0406	-0.1548	0.1133	1175
	EF	0.0406	-0.1548	0.1 3	-0.1175
Inv. Chebyshev	OF	0.5753	2 1182	1.0852	-0.8250
	EF	0.5753	-2.1182	1.0852	-0.8250
Elliptical	OF	0.0406	-0.1548	0.1133	-0.1175
	EF	0.0406	-0.1548	0.1133	-0.1175

Table 20: Resulting Feedback Coefficients

Loop-Filter		L ₁	L ₂	L ₃	L_4
Butterworth	OF	-1.2385	2.3835	-1.2385	1
	EF	-1.1836	2.1736	-1.0828	0.8372
Chebyshev	OF	-1.2385	2.3835	-1.2385	1
	EF	-1.1980	2.2287	-1.1252	0.8825
Inv. Chebyshev	OF	-1.2361	2.38 7	-1.2361	1
	EF	-0.6608	0.2635	-0. 5 8	0.1750
Elliptical	OF	-1.2385	2.3835	-1.2385	1
	EF	-1.1980	2.2287	-1.1252	0.8825

Fixed-point arithmetic is chosen and the circuits are excited by a 16-bit single sinusoidal input signal. A one-bit quantizer is employed and the output data of the VBP Σ - Δ modulators are saved in the RAM. Thereafter, the data is read via RS232 connection and processed in Matlab. No decimation filter is built in the FPGA; instead the decimation filter is applied within the Matlab routines. The experimental set-up is shown in Figure 111.



Figure 111: Experimental Set-up

Dither is used to randomize the tonal response of the Σ - Δ modulators. A 16-bit Fibonacci Linear Feedback Shift Register (LFSR) is built because of its relatively white output spectrum. As depicted in Figure 112, -70dB of white noise is obtained for normalized frequencies above 0.055.



Figure 112: LFSR Output Plot

4.5.1 Fixed Point Arithmetic

Finite wordlengths are applied to coefficients by rounding. Simulations are iteratively repeated to determine the optimum wordlength in order to acquire loop-filters with suitable noise-shaping characteristics. The fractional resolution is chosen to be 16-bits. Note that the decimal part of the number is represented by extra bits such as sD.15-bits where s is the sign bit, D is

the number of the decimal bits and the 15-bits represent the fractional part of the number. As seen in Figure 113, the use of 15-bits of fractional resolution provides very comparable results with the ideal case for both the EF and OF topologies.



Figure 113: SNR Plot, 4th-order SP Σ - Δ modulators, Butterworth NTF, Centre Frequency = 0.2, BW =0.02, OSR = 64

4.5.2 Design Bottleneck

In the design of any digital circuits, the multipliers require quite large areas and lead to long propagation delays. Therefore, any attempt to discard or minimize the multiplication blocks will end up in faster Σ - Δ modulators. Substantial savings from the feedback multipliers have already been achieved for the OF topology. This is accomplished with the proper design of the loop-filter such that the numerator of the NTF is only feedback coefficient dependent.

On the other hand, the SNR of the Σ - Δ modulator highly depends on the in-band noise that is suppressed by the NTF numerator. At the system level, it is obvious that the feedforward coefficients, K_m s, do not affect the resonant performance of the NTF of the OF topology. They only affect the location of the poles. If their wordlengths are reduced to a number where the poles remain within the unit-circle, namely in the stable region, the feedforward multipliers will result in faster circuits. However for the EF topology both the feedback and feedforward coefficients determine the zeros. Therefore any attempt to further reduce the wordlength of the feedback and/or feedforward coefficients leads to poor SNR as depicted in Figure 114.



Figure 114: SNR Plot versus Feedforward Coefficient Resolution, EF Topology, 4th-order SP Σ-Δ modulator Butterworth NTF, Centre Frequency = 0.2, BW =0.02 OSR = 64

Again, iterative simulations are performed to decide the resolution of these coefficients for the OF topology. After all, it is observed that 8-bits of resolution for the feedforward coefficients of the OF topology gives fairly good results. Simulations revealed that fractional resolution below 7-bits may cause strong in-band tones and can lead up to instability even if the poles are still in the stable region. The relationships between the SNR and feedforward coefficients for different multi-path topologies are given in Figure 115-Figure 118. In Figure 115-Figure 118 SNR versus feedforward coefficient resolution plot is shown.



Figure 115: SNR Plot versus Feedforward Coefficient Resolution, OF Topology, 4th-order SP Σ–Δ modulator Butterworth NTF, Centre Frequency = 0.2, BW =0.02 OSR = 64



Figure 116: SNR Plot versus Feedforward Coefficient Resolution, OF Topology, 4th-order 2P Σ-Δ Modulator, Chebyshev NTF, Centre Frequency = 0.2, BW =0.02, PBA = 1 dB OSR = 64



Figure 117: SNR Plot versus Feedforward Coefficient Resolution, OF Topology, 4^{th} -order 4P Σ - Δ Modulator, Inverse-Chebyshev NTF, Centre Frequency = 0.2, BW =0.02, SBA = 80 dB OSR = 64

In Figure 118 and Figure 119, the pole zero locations are shown for the ideal case and for various resolution values. The zeros of the OF topology are not affected by the $K_m s$ whereas the poles clearly change their locations specifically for fractional resolution values below 7-bits. Moreover, one of the poles moves out of the unit-circle and may cause instability. This depends on the size of the shift outside the unit-circle as the modulator stability depends on the overall closed-loop transfer function.

In Figure 120, the pole-zero patterns of the EF topology to the various fractional resolution values is seen, again for the feedforward coefficients. As expected the feedforward coefficients alter the locations of the poles and zeros. It should be noted that if the chosen filter specifications are changed, the resolution of the feedback and feedforward coefficients needs to be reinvestigated.



Figure 118: Pole Zero Map, Ideal Case, 4th-order SP Σ–Δ modulator Chebyshev NTF, Centre Frequency = 0.2, BW =0.02, PBA = 1dB



a) 15-bits Fractional Resolution

b) 7-bits Fractional Resolution



c) 5-bits Fractional Resolution

Figure 119: Pole Zero Map for Various Resolution Values of Feedforward Coefficients (K_s), OF Topology, Chebyshev NTF, Centre Frequency = 0.2, BW =0.02, PBA = 1dB



b) 5-bits Fractional Resolution

Figure 120: Pole Zero Map for Various Resolution Values of Feedforward Coefficients (K_s), EF Topology, Chebyshev NTF, Centre Frequency = 0.2, BW =0.02, PBA = 1dB

The Spartan-3 family has a dedicated array of 18x18 multipliers. Reducing the wordlength of the coefficients unfortunately does not affect the performance of the overall topology. Because the coefficients are actually inputs to the multipliers and they allocate the 18x18 multiplier blocks. Therefore the effects of reduced wordlength coefficients on the overall propagation delay and hardware reduction cannot be observed. A slight increase in speed may be achieved since the MSBs of the multipliers are not in calculation, as the MSBs require the longest calculation delay in a multiplier. Nevertheless, the universal clock frequencies of the Spartan-3 are fixed and the multiplier outputs are registered at every clock cycle. Therefore, the circuit speed cannot be improved in the VHDL implementations. The advantage of reduced wordlengths of the feedforward coefficients would be a huge benefit when it comes to integrated circuit implementations.

4.5.3 Implementation Results

The implementation results not only verify the node-equation method, but also support the behavioural-level analysis of the designed modulators. The output plots for some of the circuits are shown in Figure 121 and Figure 122. This data is saved in RAM and read through the RS232. The time-domain output data is processed within the MATLAB routines and its frequency spectrum is obtained.



Figure 121: The OF Topology, Butterworth Loop-Filter



Figure 122: The EF Topology, Chebyshev Loop-Filter

As clearly observed, the 4P topologies have deeper notches in comparison to their SP and 2P counterparts. Obviously, the 2P topologies' notches are deeper when compared to their SP counterparts. This is due to the relaxed path frequencies. All the multipliers and adders of the TI topologies are clocked with a lower clock frequency; hence the propagation delay does not affect the circuit performance. However, for the SP-topologies, the propagation delay causes data-loss therefore degrading the noise-shaping performance.

Moreover, the provided mathematical model of the quantization tones for the TI topologies clearly works. Generally, the 4P topologies are more likely to produce these quantization tones.

In Table 21 and Table 22, the behavioural-level simulations and implementation results for the SNR values are compared. These tables are for an overall OSR of 64. Note that for the 2-path topologies, the individual path OSR is 32 and for the 4-topologies the individual path OSR is 16.

Loop-Filter	Topology	SP	2-Path	4-Path
Butterworth	OF-16 bits	42 dB	55 dB	67 dB
	OF- 8 bits	41 dB	56 dB	67 dB
	EF-16 bits	41 dB	55 dB	65 dB
Chebyshev	OF-16 bits	40 dB	55 dB	67 dB
5	OF-8 bits	42 dB	57 dB	68 dB
	EF-16 bits	41 dB	54 dB	67 dB
Inv. Chebyshev	OF-16 bits	64 dB	72 dB	72 dB
5	OF-8 bits	64 dB	72 dB	72 dB
	EF-16 bits	64 dB	72 dB	72 dB
Elliptical	OF-16 bits	40 dB	55 dB	67 dB
	OF-8 bits	42 dB	57 dB	68 dB
	EF-16 bits	40 dB	55 dB	64 dB

Table 21: Simulated SNRs, Behavioural Results

Table 22: Measured SNRs, Behavioural Results

Loop-Filter	Topology	SP	2-Path	4-Path
Butterworth	OF-16 bits	41 dB	58 dB	69 dB
	OF- 8 bits	45 dB	57 dB	50 dB
	EF-16 bits	45 dB	58 dB	67 dB
Chebyshev	OF-16 bits	40 dB	54 dB	65 dB
	OF-8 bits	42 dB	58 dB	62 dB
	EF-16 bits	40 dB	55 dB	60 dB
Inv. Chebyshev	OF-16 bits	63 dB	64 dB	68 dB
č	OF- 8 bits	60 dB	42 dB	66 dB
	EF-16 bits	22 dB	4 dB	6 dB
Elliptical	OF-16 bits	40 dB	55 dB	60 dB
*	OF-8 bits	42 dB	58 dB	59 dB
	EF-16 bits	42 dB	55 dB	52 dB

Another substantial finding from the implementation results is that the Inverse-Chebyshev filter causes low-quality notches for the EF topology as depicted in Figure 123-Figure 125. As seen, the out-of band noise level matches well with the OF topology. However the in-band noise floor is elevated. It is attributed to the aggressive nature of the designed Inverse-

Chebyshev filter. The Inverse-Chebyshev NTF is designed for a bandwidth of 0.04 to acquire high resolution. As it has the narrowest bandwidth when compared to the other filters, the finite wordlength deteriorates its resonant frequencies more than the other filters.

It should be reminded that the finite wordlength of feedback and feedforward coefficients of the EF topology have a combined effect on the resonant frequency. While the OF topology's resonant frequencies are independent of its feedforward coefficients' resolution. As seen from Figure 123 - Figure 125, 4P and 2P topologies result in deeper notch for the OF topology compared to their SP counterparts due to the relaxed internal clocks. However, the finite wordlength of the feedback and feedforward coefficients of the EF topology impairs all SP, 2P and 4P topologies' resonant behaviour of the EF topology. Once again, it is shown that the designed OF topology is more appropriate to implement VBP Σ - Δ modulators.



Figure 123:Output Plots of the SP topology for Inverse- Chebyshev Loop-Filter



Figure 124:Output Plots of the 2P topology for Inverse- Chebyshev Loop-Filter



Figure 125:Output Plots of the 4P topology for Inverse- Chebyshev Loop-Filter

4.6 Hardware Complexity of the D/A VBP Σ–Δ Modulators

In Section 3.4, the hardware complexity of the A/D VBP Σ - Δ modulators is discussed comparing the topologies and the implementation techniques. The general rule is that the TI topologies utilize PxN_{ad} adders and PxN_{mul} multipliers where N_{ad} is the number of the adders and N_{mul} is the number of the multipliers of the SP counterpart of the TI topology. It was also mentioned in Section 3.4 that designers could save all of the PxN_{ad} adders if the TI topology is implemented using current-mode circuits for A/D VBP Σ - Δ modulators. However in digital world this advice does not apply. Therefore, the OF and EF topologies are investigated to save coefficients as well as the wordlength of the coefficients of the D/A VBP Σ - Δ modulators.

To summarize the outcomes of hardware reduction of the D/A VBP Σ - Δ modulators, using the FBFF TDA loop-filter the OF topology saves PxL/2 feedback coefficients where L is the order of the modulator. Moreover, depending on the NTF coefficients' values feedforward coefficients' resolution can be reduced by simulations. For detailed explanations please go to Sections 4.2.1, 4.2.2 and 4.5.2.

In Table 23 and Table 24 the allocated FPGA sources are listed for the 4th-order OF and EF topologies. As expected, the utilized LUTs and flip-flop slices are around the same range. This is due to the fact that, both topologies employ the same loop-filter structure and their coefficients do not exceed 18-bits of resolution. However, the OF topology uses 4xL/2 less multipliers when compared to the EF topology.

	FPGA Sources	Butterworth	Chebyshev	Inverse Chebyshev	Elliptical
SP	# Slice Flip Flops	87 / 40,960 ~ 1%	87 / 40,960 ~ 1%	124 / 40,960 ~ 1%	86 / 40,960 ~ 1%
	# LUTs	168 / 40,960 ~ 1%	168 / 40,960 ~ 1%	175 / 40,960 ~ 1%	164 / 40,960 ~ 1%
	# Multipliers 18x18	8 / 40 = 20%	8 / 40 = 20%	8 / 40 = 20%	8 / 40 = 20%
2P	# Slice Flip Flops	234 / 40,960 ~ 1%	232 / 40,960 ~ 1%	247 / 40,960 ~ 1%	215 / 40,960 ~ 1%
	# LUTs	501 / 40,960 ~ 1%	499 / 40,960 ~ 1%	666 / 40,960 ~ 1%	448 / 40,960 ~ 1%
	# Multipliers 18x18	16/ 40 = 40%	16/ 40 = 40%	16/ 40 = 40%	16/ 40 = 40%
4P	# Slice Flip Flops	278 / 40,960 ~ 1%	276 / 40,960 ~ 1%	294 / 40,960 ~ 1%	276 / 40,960 ~ 1%
	# LUTs	833 / 40,960 ~ 2%	831 / 40,960 ~ 2%	2501 / 40,960 ~ 6%	831 / 40,960 ~ 2%
	# Multipliers 18x18	32/ 40 = 80%	32/ 40 = 80%	32/ 40 = 80%	32/ 40 = 80%

Table 23: The Allocated FPGA Sources for The EF Topologies

Table 24: The Allocated FPGA Sources for The OF Topologies

	FPGA Sources	Butterworth	Chebyshev	Inverse Chebyshev	Elliptical
SP	# Slice Flip Flops	107 / 40,960 ~ 1%	99 / 40,960 ~ 1%	149 / 40,960 ~ 1%	101 / 40,960 ~ 1%
	# LUTs	156 / 40,960 ~ 1%	140 / 40,960 ~ 1%	198 / 40,960 ~ 1%	147 / 40,960 ~ 1%
	# Multipliers 18x18	6 / 40 = 15%	6 / 40 = 15%	6 / 40 = 15%	6 / 40 = 15%
2P	# Slice Flip Flops	215 / 40,960 ~ 1%	213 / 40,960 ~ 1%	213 / 40,960 ~ 1%	215 / 40,960 ~ 1%
	# LUTs	448 / 40,960 ~ 1%	446 / 40,960 ~ 1%	446 / 40,960 ~ 1%	448 / 40,960 ~ 1%
	# Multipliers 18x18	12/ 40 = 30%	12 / 40 = 30%	12/ 40 = 30%	16/ 40 = 30%
4P	# Slice Flip Flops	259 / 40,960 ~ 1%	257 / 40,960 ~ 1%	284 / 40,960 ~ 1%	257 / 40,960 ~ 1%
	# LUTs	725 / 40,960 ~ 1%	723 / 40,960 ~ 1%	994 / 40,960 ~ 2%	723 / 40,960 ~ 1%
	# Multipliers 18x18	24/ 40 = 60%	24 / 40 = 60%	24/ 40 = 60%	24 / 40 = 60%

4.7 A Tunable and Reconfigurable D/A VBP Σ–Δ Modulator

A second-order tunable and reconfigurable D/A VBP Σ - Δ modulator is designed and implemented. The topology is highly flexible providing a set of choices for designers. In Figure 126, the block diagram of the designed modulator is given. Basically it is a second order 2P OF topology and is built using the developed design methodology in Section 3.3.6.



Figure 126: 2^{nd} -order Tunable and Reconfigurable D/A VBP Σ - Δ modulator

By programming the switches, the topology can either operate as a SP VBP Σ - Δ modulator or as a 2P TI VBP Σ - Δ modulator. If the red route is selected, the topology is a SP VBP Σ - Δ modulator, whereas if the blue route is selected, the topology is a 2P VBP Σ - Δ modulator. The configurability of the SP and 2P options makes the circuit suitable for multi-band systems. For instance, if a narrow-band operation is required the user can select the SP option hence saving power. On the other hand, the 2P option can double the conversion band without doubling the clock frequency therefore enabling relatively wider band signal conversion. It should be noted that both SP and 2P options perform the same symbolic NTF given in (85).

$$NTF_{flex}(z) = \frac{1 + L_1 z^{-1} + z^{-2}}{1 + (K_1 + L_1) z^{-1} + (K_2 + L_2) z^{-2}}$$
(85)

As mentioned earlier, the node-equation method shares the data samples between the subsequent and/or adjacent channels. Hence, the required delayers are reduced by half for a 2P design. The switches in the configurable topology basically determine which path utilizes the delayers. It should be noted that, in Simulink the sample time is set internally for each block. However when implemented in VHDL, the clock frequencies of the cross-connected delayers and the LFSR block are also determined by switches. Since the circuit is implemented on Xilinx® SpartanTM-3 Development Kit, the universal clock frequency of 66 MHz is applied to the SP topology. When the 2P option is activated, each internal path is clocked at 33 MHz resulting in an overall clock frequency of 66 MHz. If more clock options are available within the overall system, the designer can implement them as needed by using multiplexers [102]. At the present, clock glitches due to switches are not an issue. This is because the $\Sigma - \Delta$ modulator resets itself, if an on-going path-mode is changed. If the modulator does not reset but continues to operate when the path-mode is changed, stability issues may occur. This is due to the fact that the D/A Σ - Δ modulators' stability is highly affected by the initial conditions [103]. Moreover, the transient response of the newly selected path-topology will be influenced by the steady state response of the formerly operating path-topology. The statespace equations need to be analysed if path selection is needed without reseting the circuit.

The NTF given in (85) is applied to Butterworth coefficients. The normalised bandwidth of the NTF is selected to be 0.02. The centre frequency is swept between [0.03-0.47] with a 0.01 step size. This results in 45 different centre frequencies, hence 45 different filter coefficients. The fractional resolution of the coefficients is chosen to be 15-bits. These filter coefficients K_1 , K_2 and L_1 are saved in a look up table in VHDL. At the moment the designer can select the centre-frequency from this look up table and synthesize the circuit. However, for future work the Σ - Δ modulator can calculate the coefficients itself as the Butterworth function can be solved by VHDL coding. This in return will enable the user to enter the centre-frequency and bandwidth of the modulator from a chosen peripheral.

To sum-up, the user can select the path-mode and centre-frequency of the $\Sigma-\Delta$ modulator modulator by external signals that can be determined by peripherals. At the moment, these control signals are attained internally within the VHDL code as the output data is being saved in the RAM. Therefore for a chosen path-mode and centre frequency, the circuit is synthesized on the board. The output data is saved in the RAM and read through RS232. The read output data is processed by MATLAB routines. The resulting output plots of the second-order tunable and reconfigurable D/A $\Sigma-\Delta$ modulator for different centre frequencies are depicted in Figure 127 and Figure 128. The outputs are obtained when the $\Sigma-\Delta$ modulators are excited by zero input. The limit cycle tones observed at 0.2 and 0.4 result from the repetitive output responses of $\Sigma-\Delta$ modulators as they are being excited by a steady input. DC inputs tend to produce limit cycle tones since the $\Sigma-\Delta$ modulators are actually finite-state machines.



Figure 127: Output Plots for Different Centre Frequencies, SP Topology is Activated



Figure 128: Output Plots for Different Centre Frequencies, 2P Topology is Activated

In conclusion a flexible, tunable and reconfigurable topology is designed and implemented. This circuit supports the main theme of this thesis which involved the design, evaluation and implementation of generalized TI VBP Σ - Δ modulators. Moreover, the inherent flexibility of the topology made it possible for designers to utilise different types of filters in order to maintain stable outputs for a wide range of centre frequencies.

4.8 Conclusion

The node equation method was applied in this chapter to the design and implement of D/A VBP Σ - Δ modulators that can accomplish noise-shaping for any centre frequency and bandwidth.

Two fundamental Σ - Δ modulator topologies are discussed. Contrary to the LP Σ - Δ modulator applications, it is shown that the OF topology is well suited for BP Σ - Δ modulators due to the BS NTF resonant behaviour. The OF topology not only discards some of the feedback

multipliers of the loop-filter but also works relatively well with lower resolution of the feedforward coefficients.

In addition, the developed mathematical model for quantization tones produced in A/D VBP Σ - Δ modulators is further extended to the D/A VBP Σ - Δ modulators. Since these topologies' input signal is also quantized, tones caused by the input-amplitude's finite wordlength are investigated. It is concluded that the resulting tones are masked by the NTF therefore they do not cause any stability and/or SNR deterioration. In the meantime, the internal path finite wordlength is also examined. Rather than causing distinct tones at the output spectrum, internal path truncation causes increased noise floor or noise-shaping degradation. The recursive nature of Σ - Δ modulators whitens the expected distinct quantization tones as they are being added up and multiplied to each other.

The 2nd-, 4th- and 6th-order of SP, 2P and 4P generalized D/A VBP Σ - Δ modulators are designed for the OF and EF topologies. These structures are capable of employing Butterworth, Chebyshev, Inverse-Chebyshev and Elliptical filters. The filter performances and resulting SNR values are also discussed. Once again, detailed behavioural-level simulations show that Inverse-Chebyshev filters provide the highest SNR values.

The 4th-order SP, 2P and 4P topologies are implemented on the Xilinx® SpartanTM-3 Development Kit. The output plots of these circuits and the behavioural-level design simulations are compared. However, it is observed that the EF topology does not support high SNR values when Inverse-Chebyshev filters are employed.

A second order tunable and reconfigurable VBP Σ - Δ modulator is designed and implemented. The topology can operate either on a SP mode or a 2-path mode enabling multi-band operation through the same D/A converter. Moreover the designer can select 45 different centrefrequencies each being placed with a step size of 0.01 to each other and within a normalised frequency range of [0.03 0.47]. This design supports the idea of highly flexible and tunable VBP Σ - Δ modulators are realisable. To the best knowledge of the author, this is the first reconfigurable TI topology reported in the open literature.
Chapter 5. Graphical User Interface

A user-friendly design tool created in the MATLAB/Simulink environment has been developed to accelerate the design, analysis and evaluation of single-stage and TI VBP Σ - Δ modulators.

 Σ - Δ modulators' mathematical analysis depends on white-noise assumption and extensive simulations examine the stability. Hence, from the designer perspective, it is very handy to have a toolbox or GUI for the intended work. A toolbox or GUI helps users to design a wide range of TI VBP Σ - Δ modulators. It helps them to evaluate these modulators and make fairly swift comparisons. The GUI also motivates the users in the area who do not have any expertise of code writing.

There are various design tools in the literature focusing on different problems. In [104], the pole-zero optimization of LP modulator's NTF is applied to obtain stable Σ - Δ modulators that are capable to produce higher SNRs. In [105], LP and bandpass Σ - Δ modulators' behavioural level analysis are combined in a single GUI. In [106], a toolbox is constructed to speed up the design of continuous-time bandpass Σ - Δ modulators.

In this study, a novel GUI is built where the designed A/D and D/A SP and TI VBP Σ - Δ modulators' Simulink models cooperate with the Matlab simulation codes. This GUI is the first example of its kind as it enables users to study not only SP but also 2-path and 4path TI structures. The user can select conversion type, topology, loop-filter type, order, path number and OSR to plot the output frequency and SNR responses. The GUI is constructed using Matlab R2010b 64-bit.

5.1 Installation

Since the GUI is built in a Matlab environment there is no need to install an application to the computer. The provided *"VBP SDMs"* folder should be copied somewhere in the computer, preferably to the root directory as it is faster for Matlab to compute. The folder's size is 2.6 MB.

Thereafter the "*GUI_VBP.fig*" file should be executed with Matlab GUIDE. This can be done easily by following the steps listed below.

- 1. Open Matlab.
- 2. Open the directory of the folder in Matlab.
- 3. Double click on "GUI VBP.fig".

Or simply right click on "*GUI_VBP.fig*" and select Open in Matlab. The GUI menu should be seen on screen as depicted in Figure 129.

00	GUI_VBP	
Please o	choose the following optic	ons accordingly.
	Select the conversion	\$
	Select a topology	\$
	Select a loop-filter	\$
	Select the order	\$
	Select path number	\$
	Select an operation	\$
	Select an OSR	\$
	Center Frequency	
	Bandwidth	
	Pass Band Ripple	
	Stop Band Ripple	
	EXECUTE	

Figure 129: GUI Menu

5.2 The Menu Bar

The menu is designed to be as basic as possible. The user selects the conversion type by 'Select the conversion' bar in the first place. If the 'Analog to Digital' conversion is selected, the GUI simulates the CR-RFB topologies only for the ideal-case. If the 'Digital to Analog' conversion is selected, the GUI can simulate both the EF and OF topologies for 15-bits of fractional resolution.

00	GUI_VBP	1	00	GUI_VBP
Please choose	e the following ontions accordingly.		Please cho	oose the following options accordingly.
✓ An	alog to Digital			Digital to Analog
Se	lect a topology			Select a topology
Se	lect a loop-filter			Error Feedback
Se	lect the order		[Select the order \$
Sel	lect path number 💠			Select path number \$
Se	lect an operation \$			Select an operation \$
Se	lect an OSR \$			Select an OSR \$
Cen	ter Frequency		0	Center Frequency
Ban	dwidth		E	Bandwidth
Pas	s Band Ripple		F	Pass Band Ripple
Stop	Band Ripple		s	Stop Band Ripple
	EXECUTE			EXECUTE

Figure 130: Selecting the Conversion

Thereafter, the loop-filter needs to be selected from the options: Butterworth, Chebyshev, Inverse-Chebyshev and Elliptical. Once the order and path number are also chosen, the GUI is able to decide on the overall structure.

GUI_VBP	GUI_VBP	e o Cui_VBP
Please choose the following options accordingly.	Please choose the following options accordingly.	Please choose the following options accordingly.
Digital to Analog	Digital to Analog ‡	Digital to Analog \$
Output Feedback	Output Feedback \$	Output Feedback \$
✓ Select a loop-filter	Chebyshev \$	Chebyshev ‡
Chebyshev Inverse Chebyshev	✓ Select the order	4 \$
Elliptical th number	4 lect path number	✓ Select path number
Select an operation ‡	Select an operation =	2-path TI Dutput Spectrum
Select an OSR \$	Select an OSR \$	4-path TI object an Con
Center Frequency	Center Frequency	Center Frequency
Bandwidth	Bandwidth	Bandwidth
Pass Band Ripple	Pass Band Ripple	Pass Band Ripple
Stop Band Ripple	Stop Band Ripple	Stop Band Ripple
EXECUTE	EXECUTE	EXECUTE

Figure 131: Selecting the Specifications

There are four editable text boxes within the GUI where the user can input data and determine the NTF centre frequency, bandwidth and pass-band/stop-band attenuation parameters. The GUI calculates the filter coefficients depending on these data.

The GUI performs two operations. It can plot the output response in the frequency domain or the SNR versus input amplitude response of the selected modulator. If the SNR plot is selected, the user must choose an OSR within the specified values of 8, 16, 32, 64,128 and 256.

GUI_VBP	⊖ ⊖ ⊂ GUI_VBP	GUI_VBP
Please choose the following options accordingly.	Please choose the following options accordingly.	Please choose the following options accordingly.
Error Feedback +	Error Feedback \$	Digital to Analog
Chebyshev \$	Chebyshev \$	Chebyshev ‡
2-path TI	2-path TI ‡	2-path TI ‡
Plot the Output Spectrum Plot the SNR	✓ Select an OSR	✓ Select an OSR
Center Frequency 0.3 Bandwidth 0.02	32 - Frequency 1 0.3] 64 128 - 64	16 32 or Frequency 0.3 64 128
Pass Band Ripple 2	256 Pass Band Ripple 2	256 Pass Band Ripple 2
Stop Band Ripple	Stop Band Ripple	Stop Band Ripple
EXECUTE	EXECUTE	EXECUTE

Figure 132: Selecting the Specifications

If all the inputs are provided for the GUI the '*EXECUTE*' push button functions properly. Otherwise an error message appears on the screen saying '*You must enter all required inputs*' as shown in Figure 133. Note that the Matlab Command Window also produces an error summary as the GUI does not function properly in the case of a missing input.

	GUI_VE	3P
Please c	hoose the following	options accordingly.
	Digital to Analog	÷
•	Fror Feedback	*]
	You must enter	a numeric value
	ОК	
	Select an OSR	\$
	Center Frequency	
	Bandwidth	0.05
	Pass Band Ripple	
	Stop Band Ripple	
	EXECUT	E

Figure 133: Error Message

5.3 Outputs

The output spectrum is plotted within a frequency interval of [0 0.5]. If A/D conversion is chosen, the input amplitude equals to 0.5 and the dithering amplitude equals to 0.05. The '*Plot the Output Spectrum*' simulation takes around 15 seconds of running time. The selected topology's Simulink model opens during simulation and it closes when the simulation is completed. In this way the user can also confirm that the chosen topology is being simulated.

If the user wishes to display the determined coefficients of the selected topology, this can be done by simply using the '*display (variable)*' command of MATLAB. For instance, to display the K_1 coefficient, the user needs to write '*display (K1)*' on the command window.

The Simulink blocks' names are generated such as:

'Conversion TI.Path Topology Order.mdl'.

Refer to

Table 25 for all Simulink blocks' names. Moreover, one of the GUI output plot is given in Figure 134. Note that the title of the plot is set such that the selected conversion, topology, order and filter are defined.



Figure 134: GUI Output Plot

The SNR calculation takes around 40 minutes of running time. Depending on the order and topology, the simulation time varies. The generation of the SNR plot takes considerably more time when compared to the production of the output spectrum. This is due to the fact that the SNR values are calculated individually for 48 input amplitude values. Therefore the DR of the modulator, maximum and minimum achievable SNR values are depicted in a single plot. In Figure 135, an SNR plot is given. Again the title of the plot is set such that the selected conversion, topology, order and filter are defined.



Figure 135: GUI SNR Plot

Table 25: Simulink Blocks' Names

Simulink Block Name	Conversion	Path	Topology	Order
DAC_SP_OF_2 nd .mdl	D/A	SP	OF	2
DAC_SP_OF_4 th .mdl	D/A	SP	OF	4
DAC_SP_OF_6 th .mdl	D/A	SP	OF	6
DAC_SP_EF_2 nd .mdl	D/A	SP	EF	2
DAC_SP_EF_4 th .mdl	D/A	SP	EF	4
DAC_SP_EF_6 th .mdl	D/A	SP	EF	6
AD_SP_RESFB_2 nd .mdl	A/D	SP	CR-RFB	2
AD_SP_RESFB_4 th .mdl	A/D	SP	CR-RFB	4
AD_SP_RESFB_4 th .mdl	A/D	SP	CR-RFB	6
DAC_TI2_OF_2 nd .mdl	D/A	2P	OF	2
DAC_TI2_OF_4 th .mdl	D/A	2P	OF	4
DAC_TI2_OF_6 th .mdl	D/A	2P	OF	6
DAC_TI2_EF_2 nd .mdl	D/A	2P	EF	2
DAC_TI2_EF_4 th .mdl	D/A	2P	EF	4
DAC_TI2_EF_6 th .mdl	D/A	2P	EF	6
AD_TI2_RESFB_2 nd .mdl	A/D	2P	CR-RFB	2
AD_TI2_RESFB_4 th .mdl	A/D	2P	CR-RFB	4
AD_TI2_RESFB_4 th .mdl	A/D	2P	CR-RFB	6

DAC_TI4_OF_2 nd .mdl	D/A	4P	OF	2
DAC_TI4_OF_4 th .mdl	D/A	4P	OF	4
DAC_TI4_OF_6 th .mdl	D/A	4P	OF	6
DAC_TI4_EF_2 nd .mdl	D/A	4P	EF	2
DAC_TI4_EF_4 th .mdl	D/A	4P	EF	4
DAC_TI4_EF_6 th .mdl	D/A	4P	EF	6
AD_TI4_RESFB_2 nd .mdl	A/D	4P	CR-RFB	2
AD_TI4_RESFB_4 th .mdl	A/D	4P	CR-RFB	4
AD_TI4_RESFB_4 th .mdl	A/D	4P	CR-RFB	6

5.4 **Programmer Guide**

This section covers some of the basics of the GUI's software hence providing an opportunity to the programmers to further develop the GUI and/or to build up their own simulation environment.

The GUI is formed from several subprograms. The main body of the GUI is the '*GUI.VBP.m*' file where all the input data are collected, the related loop-filter transfer function is calculated and the output is plotted. Moreover, if there is any missing input data or invalid input values, the error messages are produced in this file. The '*GUI.VBP.m*' file has its own comments inside the programmers guide, such that they have better understanding of the variable names and control flags.

Loop-filter Transfer Function Calculation: The noise transfer function is calculated by Matlab using the input data. The NTF calculation is performed regardless of the selected topology. The given codes below are for the four different NTFs.

```
%Butterworth NTF
[num, den] = butter(order, [freq1, freq2], 'stop');
%Chebyshev NTF
[num, den] = cheby1(order, PBA, [freq1, freq2], 'stop');
%Inverse-Chebyshev NTF
[num, den] = cheby2(order, SBA, [freq1, freq2], 'stop');
%Elliptical NTF
[num, den] = ellip(order, PBA, SBA, [freq1, freq2], 'stop');
136
```

Once the NTF is obtained, it is normalized for causality of the loop-filter. Thereafter the loop-filter's transfer function is calculated accordingly. Since the CR-RFB topology is also an OF structure, there are two different loop-filter transfer functions regardless of the conversion selection. The loop-filter transfer function is calculated using the AWGN model and resulting equations are given in (87) and (89).

$$NTF_{OF}(z) = \frac{1}{1 + LF_{OF}(z)}$$
(86)

$$LF_{OF}(z) = \frac{1 - NTF_{OF}(z)}{NTF_{OF}(z)} = \frac{num_{NTF} - den_{NTF}}{num_{NTF}}$$
(87)

$$NTF_{EF}(z) = 1 - LF_{EF}(z)$$
(88)

$$LF_{EF}(z) = 1 - NTF_{EF}(z) = \frac{den_{NTF} - num_{NTF}}{den_{NTF}}$$
(89)

The Matlab functions to calculate the loop-filter transfer function are listed in Table 26.

Function File	Corresponding Operation
butter_BP_Hz.m	Butterworth, the EF topology's loop-filter
cheb_BP_Hz.m	Chebyshev, the EF topology's loop-filter
cheb2_BP_Hz.m	Inverse-Chebyshev, the EF topology's loop-filter
ellip_BP_Hz.m	Elliptical, the EF topology's loop-filter
butter_BP_LF.m	Butterworth, the OF topology's loop-filter
cheb_BP_LF.m	Chebyshev, the OF topology's loop-filter
cheb2_BP_LF.m	Inverse-Chebyshev, the OF topology's loop-filter
ellip_BP_LF.m	Elliptical, the OF topology's loop-filter

Table 26: Functions for Loop-filter Transfer Function Calculation

det_block.m : This function returns the selected Simulink block's name. The block name is not global, so the user cannot display it on the command window. However, as mentioned earlier, the selected topology is opened during simulation and is closed before the plot appears.

det_variables.m: The input and dither amplitude values for the output spectrum plot and fractional wordlength of the D/A modulators' coefficients are set by this function. If the programmer is interested in investigating the finite wordlength effect of the D/A modulator blocks or the modulators' responses to different input and dither amplitudes, the variables inside this function need to be changed. Variable names are listed in Table 27. Note that dither noise amplitude only affects the A/D modulators' performance since a LFSR block with fixed amplitude of output noise is employed for the D/A modulators as covered in Section 4.5. To understand the fixed-point arithmetic used for the D/A modulators go to Section 4.5.1.

Variable Name	Corresponding Datum
INPUT	Input signal's amplitude
KDit	Dither Noise amplitude
R	Input signal's fractional wordlength
RK	Feedforward Coefficients' fractional wordlength
RL	Feedback Coefficients' fractional wordlength

Table 27: Variable Names Used in det_variables.m

det_coef.m: This function does not return any value. All the coefficients are calculated within this function and are set to be global for the Simulink blocks to use, Hence the user can display them on the command window. This function calls nine different functions. Those are responsible for calculating the individual topology's coefficients. These functions are listed in Table 28. Note that the employed coefficient equations are given in appendices and calculated using the AWGN model for the quantizer behaviour.

Function Name	Corresponding Operation
ad_resfb_2nd_coef.m	A/D, 2 nd -order, the CR-RFB Topology coefficients' calculation
$ad_{resfb_4th_coef.m}$	A/D, 4 th -order, the CR-RFB Topology coefficients' calculation
ad_resfb_6th_coef.m	A/D, 6 th -order, the CR-RFB Topology coefficients' calculation
<pre>dac_OF_2nd_coef.m</pre>	D/A, 2 nd -order, the OF Topology coefficients' calculation
<pre>dac_OF_4th_coef.m</pre>	D/A, 4 th -order, the OF Topology coefficients' calculation
<pre>dac_OF_6th_coef.m</pre>	D/A, 6 th -order, the OF Topology coefficients' calculation
dac_EF_2nd_coef.m	D/A, 2 nd -order, the EF Topology coefficients' calculation
dac_EF_4th_coef.m	D/A, 4 th -order, the EF Topology coefficients' calculation
<pre>dac_EF_6th_coef.m</pre>	D/A, 6 th -order, the EF Topology coefficients' calculation

Table 28: Functions called by det_coef.m

PlotOut.m: This function is called if the '*Plot the Output Spectrum*' operation is chosen. Basically the function simulates the chosen Simulink block. A Hanning window is applied to the time-domain output data for 2^{20} FFT points. The negative frequency of the output spectrum is excluded and the remaining positive frequency output spectrum is plotted on the screen. The comments within the function provide easy-to-follow explanations for the programmers.

SNRplot.m: This function is called if the *'Plot the SNR'* operation is chosen. First of all, the function calculates the noise floor within the signal-band by simply applying zero-input to the selected modulator. The noise floor is then averaged around the input signal's frequency for a very narrow band; around 0.0012. Thereafter the iterative simulations start.



a) Zero-Input Response b) The in-band Noise To Be Averaged Figure 136: Average Noise Calculation Plots

Iterative Simulations of SNR Values:

1. The input amplitude is increased in every iteration. The selected input amplitude interval is [0.00001 1.2]. This interval covers the DR of the all modulators.

2. The input and output signals' spectrums are calculated in every iteration. Once again, a Hanning window for 2^{20} FFT points is employed.

3. The averaged noise floor obtained with zero-input signal is replaced with the input signal. Finally, the overall in-band noise is calculated according to the chosen OSR and path-number.



Figure 137: Clearing The Input Signal

4. The SNR value is calculated since the in-band noise and signal amplitude values are known.

Once iterations are completed, the SNR array is put through another process where the negative SNR values are excluded hence providing the user to observe the positive SNR values versus input signal amplitude plot. In this way it is easy to distinguish the overall DR value as well.

set_title.m: This function returns the title of the plot where the selected conversion type, topology, order and the filter are written.

5.5 Conclusion

As a final product of the overall work, 27 topologies are wrapped up in a single GUI environment. The GUI enables the users to specify the centre frequency, bandwidth, pass-band/ stop-band attenuation parameters as well as the OSR value for a selected topology. The GUI combines 1857 lines of code to operate in the same environment.

The GUI serves the purpose of this thesis, as it enables the analysis of VBP Σ - Δ modulators for any centre frequency and bandwidth. It does not require any detailed knowledge of Σ - Δ modulators therefore enabling the beginners to simulate VBP Σ - Δ modulators. It also helps users to further their understanding of the TI VBP Σ - Δ modulators. In addition, for those who are willing to develop their own simulation routines and/or GUI environment, it has a wellstructured code that is composed of sub-functions.

This GUI is a unique environment for the analysis of VBP Σ - Δ modulators. Moreover it is promising that the GUI itself can be extended to a toolbox where the user can employ different functions to form their own models and codes.

Chapter 6. CONCLUDING REMARKS and RECOMMENDATIONS FOR FUTURE WORK

In this work, single-path and multi-path VBP $\Sigma - \Delta$ modulators are studied. The main motivation of the thesis was to design, evaluate and implement generalized TI VBP $\Sigma - \Delta$ modulators that can accomplish noise-shaping for a chosen design specifications such as the centre frequency, bandwidth, pass-band and stop-band attenuations. These generalized TI VBP $\Sigma - \Delta$ modulators have the ability to provide flexible structures where multi-band and/or multi-frequency systems can achieve data conversion through a single A/D and/or D/A converter. Moreover, their capacity to perform noise-shaping at the required centre frequency eliminates the passive circuitry of the conventional superheterodyne and homodyne receivers that use LP and BP $\Sigma - \Delta$ modulators. Excluding the passive circuitry such as mixers and filters makes it possible for the receiver to be integrated to the overall system at a higher level.

The study resulted in four publications listed below.

[1] I. Kalafat Kızılkaya, M. Al-Janabi, and I. Kale, "Design and Evaluation of Time-Interleaved Variable Center-Frequency Sigma-Delta Modulators," DSP 2013, Greece, July 2013

[2] I. Kalafat Kızılkaya, M. Al-Janabi, and I. Kale, "Novel time-interleaved variable-center frequency sigma-delta modulators - design, analysis and critical evaluation," I2MTC 2013, USA, May 2013

[3] I. Kalafat Kızılkaya, M. Al-Janabi, and I. Kale, "Novel Time-Interleaved Variable Centre-Frequency, Single-Bit A/D and D/A Sigma-Delta Modulator Topologies," IMEKO 2013, Spain, July 2013

[4] I. Kalafat Kızılkaya, M. Al-Janabi, and I. Kale, "Design and implementation of novel FPGA based time- interleaved variable centre-frequency digital sigma-delta modulators", in ACTA IMEKO on, vol. 4, no.1, pp.68-75, Feb 2015

In addition to the already published papers, three more papers are being written.

[5] I. Kalafat Kızılkaya, M. Al-Janabi, and I. Kale, "Hardware Reduction in The Implementation of Digital Variable-Banspass Time-Interleaved Sigma Delta Modulators", is ready to be submitted.

[6] A GUI paper is in progress.

[7] A paper on the reconfigureable and tunable 2nd-order D/A TI Sigma-Delta modulator will be written afterwards.

6.1 Concluding Remarks

In *Chapter 1*, the benefits of VBP $\Sigma-\Delta$ modulators were discussed. In addition, the fundamental aims of the thesis were explained as well as the limitations of conventional LP and BP $\Sigma-\Delta$ modulators. It was stated that $\Sigma-\Delta$ modulators utilize high OSRs to obtain higher SNRs. However, the use of high OSRs restricted the conversion bandwidth of conventional LP and BP $\Sigma-\Delta$ modulators to narrow-band applications. Hence, it was shown that the TI topologies offered an elegant situation as they employ P mutually cross-connected $\Sigma-\Delta$ modulators operating together in order to obtain a sampling frequency of Pxf_s , where f_s is the individual path sampling frequency. Therefore, it was concluded that building generalized TI VBP $\Sigma-\Delta$ modulators not only provides flexibility but also offers a suitable solution to the narrow-band limitation of LP and BP $\Sigma-\Delta$ modulators.

It was demonstrated that noise-shaping at frequencies close to Nyquist or DC resulted in unequal shoulder gain levels if resonator-based NTFs were used. This may cause instability and/or ear-detectable noise. The unequal shoulder gain levels of the mid-band BP Σ - Δ modulators were overcome by applying different filtering types to the loop-filter such as the Butterworth, Chebyshev, Inverse-Chebyshev and Elliptical filters. These filters also enable the designer to control the bandwidth, stop-band and pass-band attenuation parameters of the NTF.

In *Chapter 2*, the basic principles of data converters were explained. The sampling and the quantization operations were explained and their mathematical analysis was given. These two

blocks were deemed important as they formed the basis for conventional PCM converters. The underpinning principles of the Σ - Δ modulation technique as well as the noise-shaping and oversampling were explained and were shown to achieve high resolution i.e., high SNRs and high DRs. The system-level diagrams of single-path A/D and D/A based Σ - Δ modulators were also presented and discussed in Chapter 2.

After explaining the fundamentals of LP and BP Σ - Δ modulators, the motivations to design TI VBP Σ - Δ modulators were discussed. To do so, a detailed up-to-date literature review and discussion were given. The TI VBP Σ - Δ modulators were presented as suitable candidates for today's emerging technologies as they offered high resolution and relatively simple hardware whilst providing flexibility and higher integration-levels. The novelties and the developed contributions of this thesis were listed and detailed in Chapter 2. Finally Chapter 2 covered the performance metrics of evaluating the SNR, DR, stability and tonality of Σ - Δ modulators.

In *Chapter 3*, the detailed design, analysis, evaluation and comparison of A/D TI VBP $\Sigma - \Delta$ modulator topologies with possible applications in superheterodyne and homodyne receivers were presented. Generalised yet flexible TI VBP $\Sigma - \Delta$ modulator modulators were developed where the designer is able to specify the centre frequency bandwidth, stop-band and pass-band attenuations from a menu of filters. This in return provided:

- a. More stable Σ - Δ modulator topologies,
- b. Improved integration for IC circuits,
- c. Improved SNR and DR.

The narrow band restriction was overcome by the time-interleaving technique thus enabling higher frequency conversion.

Thereafter, a step-by-step design method was provided that enables the design of generalized TI VBP Σ - Δ modulators. The main points are summarised below:

a. The user is able to select a generalized NTF that can accomplish Lth-order Butterworth, Chebyshev, Inverse-Chebyshev and Elliptical noise shaping, where the NTF is applied to a block filter in Simulink before choosing a topology. Therefore the filter coefficients and orders are determined to ensure adequate SNR and as well as stability.

- b. A topology is selected to map the chosen NTF.
- c. The node equation technique is applied accordingly for a chosen path-number.

In [20], the node-equation method to design TI LP Σ - Δ modulators developed. The nodeequation method is an easy to apply technique resulting in simpler hardware when compared to the polyphase decomposition method. Therefore the node equation method was chosen for this work and was applied to VBP Σ - Δ modulators. This extension of the node-equation method enabled the design of multi-path VBP Σ - Δ modulators. It should be noted that polyphase decomposition of VBP Σ - Δ modulators especially for orders four and above is cumbersome and quite hard to solve. Even if done properly, each design specification would result in different TI topologies as their NTFs are changed. However, the node-equation method is relatively easy to apply to build-up generalized TI VBP Σ - Δ modulators. By simply changing the feedback and feedforward coefficients of the designed TI VBP Σ - Δ modulators, the designers can accomplish noise-shaping at a desired centre-frequency with the chosen design specifications.

In this thesis, the node-equation method was applied to two different topologies; the CI-FBFF and the CR-RFB. The main rationale for choosing these topologies was to compare the integrator-based and resonator-based structures. The CI-FBFF topology is formed of cascaded integrators with feedback and feedforward coefficients, whereas the CR-RFB topology is formed of cascaded resonators with local feedback coefficients. Both topologies were built for 2nd-, 4th-, 6th- and 8th- orders as well as for different path numbers such as the SP, 2-path and 4-path structures. Moreover Butterworth, Chebyshev, Inverse-Chebyshev and Elliptical filters were applied to these generalized topologies. In Chapter 3, all structures are given. The CI-FBFF and the CR-RFB were compared in terms of their coefficients, SNRs, DRs and tonality.

The cascaded nature of the CI-FBFF topology resulted in signal accumulation at the internal nodes. Hence, the feedback and feedforward coefficients inevitably increase to balance this accumulation. Extensive simulations revealed that the CI-FBFF topology is not realizable for centre-frequencies [0.25 0.5] especially for 6th-order and above. However, the local feedbacks

of the CR-RFB topology prevent signal accumulation at the internal nodes therefore the coefficients do not exponentially increase.

Butterworth-based VBP Σ - Δ modulators were able to provide stable frequency responses for relatively wider bandwidths i.e., 0.1 of normalized bandwidth. On the other hand, Inverse-Chebyshev-based VBP Σ - Δ modulators with orders higher than 4 could easily go unstable for normalized bandwidths larger than 0.08. Elliptical and Chebyshev-based VBP Σ - Δ modulators actually produced stable outputs with quite strong out-of-band tones for relatively larger bandwidths when compared to Inverse-Chebyshev filters. Despite their narrow band restriction, Inverse-Chebyshev filters presented the highest SNR values due to their spread zeros over the signal band.

Moreover, the SP, 2-path and 4-path structures were compared in terms of their immunity to non-idealities. First of all, their non-idealities were defined and modelled at the behavioural level. Thereafter simulations were run to identify the tolerable percentage of non-idealities for each structure. The SP CI-FBFF and the CR-RFB topologies were investigated for non-ideal conditions by using non-ideal integrators and resonators. Higher-order SP CI-FBFF topologies were more sensitive to non-idealities as they could tolerate 0.1 % of non-idealities. On the other hand, the CR-RFB topologies were shown to be more immune to non-ideal resonators. The TI structures were investigated by using non-ideal adders as the node-equation method distributed the resonators and integrators over the interleaving paths. It was reported that due to their cross-connectivity, non-ideal adders resulted in an increased level of noise floor. Hence the TI structures' immunity to non-idealities is not dependent on the SP topology they are derived from. The path-number and filter order are the defining parameters for the TI structures.

Finally, the tonal behaviour of quantizers when excited by single-sinusoids was mathematically modelled to calculate the amplitudes and frequencies of the tones. This mathematical model was named as the sawtooth quantization noise model. The calculated results were compared with the simulation results. It was shown that, the derived model can calculate the amplitudes and the frequencies of these tones if the input signal's frequency is not irrational. A summary of the sawtooth quantization noise model can be found in Page 88.

Moreover, the SP and multi-path structures' tonal behaviour was investigated separately. The sawtooth quantization noise model was applied to SP VBP Σ - Δ modulators and further extended for the TI VBP Σ - Δ modulators.

The quantization tones seen in the SP topologies were also observed in TI topologies. Yet the shifted copies of these tones were also observed. More importantly, they were folded back into the signal band thereby resulting in in-band tones. In [81], these tones were observed and referred to as limit-cycle tones caused by the path mismatches. However, it was proved that these tones could be observed even for the ideal-case simulations. These tones' frequencies were also calculated by the sawtooth quantization noise model that was extended for the TI topologies.

In *Chapter 4*, generalized D/A TI VBP Σ - Δ modulators were designed, built and simulated. The provided design technique in Chapter 3 was applied to 2nd-, 4th- and 6th-order EF and OF topologies. Their 2-path and 4-path counterparts were also built. Once again these D/A TI VBP Σ - Δ modulators could perform Butterworth, Chebyshev, Inverse-Chebyshev and Elliptical noise-shaping.

Firstly, the EF and the OF topologies were compared. With the proper design of the loopfilter, the multipliers of the OF topology can be reduced by $P \times L/2$ where P is the path number and L is the order of the loop-filter. The mathematical analysis of this reduction was also given in Chapter 4. It was demonstrated that half of the feedback coefficients of the EF topology were eliminated. In addition, some particular frequencies are listed in Table 17 where a further reduction in the number of multipliers for the EF topology is possible.

Secondly, the developed sawtooth quantization noise model was applied to D/A SP and TI VBP Σ - Δ modulators. The results differed from the A/D VBP Σ - Δ modulators as the finite coefficients of the D/A VBP Σ - Δ modulators caused noise-floor elevation. It was also depicted that the discrete tones caused by the input signal's finite wordlength were masked by the NTF. In conclusion, the quantization tones' frequencies can be calculated as in A/D VBP Σ - Δ modulators since the 1-bit quantizer is the only source of these tones. However, it must be realised that the increased noise floor may diminish some of these expected tones.

Thereafter the designed 4th-order topologies were implemented in VHDL and synthesized on the Xilinx® SpartanTM-3 Development Kit for a chosen set of design spesifications. These spesifications and the resulting filter coefficients were listed in Table 18, Table 19 and Table 20. The universal clock of the Xilinx® SpartanTM-3 Development Kit was 66 MHz. Hence, the resulting clock frequencies were 33MHz and 16.5MHz for the 2-path and 4-path modulators' individual paths respectively.

Fixed-point arithmetic was applied to the implemented circuits. The fractional resolution of the coefficients was chosen to be 15-bits. Moreover, it was shown that the OF topology's feedforward coefficients' resolution could be reduced to 8-bits, where the SNR and DR varied by 2-4 dBs whilst maintaining modulator stability.

The implemented circuit performance and simulation results were listed in Table 21 and Table 22. The Inverse-Chebyshev filter could not perform high SNR values for the EF circuits whilst providing the expected SNRs for the OF circuits. It was attributed to the aggressive nature of the designed Inverse-Chebyshev filter. The feedback and feedforward coefficients of the EF topology had a combined effect on the resonant frequency resolution. Hence, the narrow-band design of the Inverse-Chebyshev filter is more likely to deteriorate in noise-shaping performance when non-idealties apply.

Finally, a tunable and reconfigurable 2^{nd} -oder 2P VBP Σ - Δ modulator was designed and implemented. The topology offered a highly flexible set of choices for designers. It could either operate either in SP mode or 2-path mode enabling multi-band implementations on a single D/A VBP Σ - Δ modulator. Moreover, the NTF can be tuned to 45 different centre frequencies in intervals of 0.01 within the normalised frequency range [0.03 0.47]. Therefore, the designed circuits can be utilized for multi-band and multi-frequency systems.

In *Chapter 5*, a novel GUI was built where the designed A/D and D/A SP and TI VBP the Σ - Δ modulators' Simulink models could work with with the Matlab simulation routines developed by the author. The user can select conversion type, topology, loop-filter type, order, path number and OSR to plot the output frequency and SNR responses. A user guide was given in Chapter 5 for the designed GUI in which the installation and user options were explained.

Moreover, the sub-functions and overall Matlab code was clarified for those who are interested in adding more functions to this tool.

6.2 **Recommendations for Future Work**

A wide range of VBP Σ - Δ modulators were designed and evaluated in this study. The comparisons were made to cover the fundamental design aspects of the Σ - Δ modulators such as the building block type, applied filter type, loop-filter order and topology type. Based on the experience of this research, the following suggestions are presented for future work.

- Implement the A/D TI VBP Σ–Δ modulators in SC circuits to realize fully tunable and reconfigurable circuits.
- Apply the provided design methodology of the TI VBP Σ-Δ modulators to MASH topologies to build up higher-order Σ-Δ modulators. Hence, the accumulation resulting from the cascaded nature of the CI-FBFF topology can be avoided.
- Implement higher-orders tunable and reconfigurable D/A VBP Σ - Δ modulators.
- Design and implement higher path numbers for the configurable D/A VBP $\Sigma \Delta$ modulators. This is to switch from single- to P-path for the same topology.
- Mathematically analyse the tunable and reconfigurable D/A VBP Σ - Δ modulators. Hence switching without reset may be possible.

Appendix A

The designed A/D CI-FBFF VBP Σ - Δ modulators are depicted in this Appendix. Their symbolic NTFs are also provided.

Symbolic NTF for 2nd-order CI-FBFF Topology:

$$NTF_{CI-FBFF,2^{nd}}(z) = \frac{1 + (K_1 - 2)z^{-1} + z^{-2}}{1 + L_1 z^{-1} + L_2 z^{-2}}$$
(90)



Figure 138: 2nd-order, SP CI-FBFF Topology



Figure 139: 2nd-order, 2-path CI-FBFF Topology



Figure 140: 2nd-order, 4-path CI-FBFF Topology

Symbolic NTF for 4th-order CI-FBFF Topology:

$$NTF_{CI-FBFF,4^{th}}(z) = \frac{1 + (K_1 - 4)z^{-1} + (K_2 - 2K_1 + 6)z^{-2} + (K_3 + K_1 - K_2 - 4)z^{-3} + z^{-4}}{1 + L_1 z^{-1} + (L_2 - 2L_1)z^{-2} + (L_3 + L_1 - L_2)z^{-3} + L_4 z^{-4}}$$
(91)



Figure 141: 2nd-order, SP CI-FBFF Topology



Figure 142: 4th-order, 2-path CI-FBFF Topology



Figure 143: 4th-order, 4-path CI-FBFF Topology

Symbolic NTF for 6th-order CI-FBFF Topology:

$$NTF_{CI-FBFF,6^{th}}(z) = \frac{1 + n_2 z^{-1} + n_3 z^{-2} + n_4 z^{-3} + n_5 z^{-4} + z^{-5}}{1 + d_2 z^{-1} + d_3 z^{-2} + d_4 z^{-3} + d_5 z^{-4} + d_6 z^{-5}}$$
(92)

$$n_2 = K_1 - 6 \tag{93}$$

$$n_3 = K_2 - 4K_1 + 15 \tag{94}$$

$$n_4 = K_3 + 6K_1 - 3K_2 - 20 \tag{95}$$

$$n_5 = K_4 - 4K_1 + 3K_2 - 2K_3 + 15 \tag{96}$$

$$n_6 = K_5 + K_1 - K_2 + K_3 - K_4 - 6 \tag{97}$$

$$d_2 = L_1 \tag{98}$$

$$d_3 = L_2 - 4L_1 \tag{99}$$

$$d_4 = L_3 + 6L_1 - 3L_2 \tag{100}$$

$$d_5 = L_4 - 4L_1 + 3L_2 - 2L_3 \tag{101}$$

$$d_6 = L_5 + L_1 - L_2 + L_3 - L_4 \tag{102}$$

$$d_7 = L_7 \tag{103}$$



Figure 144: 6th-order, SP CI-FBFF Topology







Figure 146: 6th-order, 4-path CI-FBFF Topology

Symbolic NTF for 8th-order CI-FBFF Topology:

$$NTF_{CI-FBFF,8^{th}}(z) = \frac{1 + n_2 z^{-1} + n_3 z^{-2} + n_4 z^{-3} + n_5 z^{-4} + n_6 z^{-5} + n_7 z^{-6} + n_8 z^{-7} + z^{-8}}{1 + d_2 z^{-1} + d_3 z^{-2} + d_4 z^{-3} + d_5 z^{-4} + d_6 z^{-5} + d_7 z^{-6} + d_8 z^{-7} + d_9 z^{-8}}$$
(104)

$$n_2 = K_1 - 8 \tag{105}$$

$$n_3 = K_2 - 6K_1 + 28 \tag{106}$$

$$n_4 = K_3 + 15K_1 - 5K_2 - 56 \tag{107}$$

$$n_5 = K_4 - 20K_1 + 10K_2 - 4K_3 + 70 \tag{108}$$

$$n_6 = K_5 + 15K_1 - 10K_2 + 6K_3 - 3K_4 \tag{109}$$

$$n_7 = K_6 - 6K_1 + 5K_2 - 4K_3 + 3K_4 - 2K_5 + 28$$
(110)

$$n_7 = K_7 + K_1 - K_2 + K_3 - K_4 + K_5 - K_6 - 8$$
(111)

$$d_2 = L_1 \tag{112}$$

$$d_3 = L_2 - 6L_1 \tag{113}$$

$$d_4 = L_3 + 15L_1 - 5L_2 \tag{114}$$

$$d_5 = L_4 - 20L_1 + 10L_2 - 4L_3 \tag{115}$$

$$d_6 = L_5 + 15L_1 - 10L_2 + 6L_3 - 3L_4 \tag{116}$$

$$d_7 = L_6 - 6L_1 + 5L_2 - 4L_3 + 3L_4 - 2L_5 \tag{117}$$

$$d_8 = L_7 + L_1 - L_2 + L_3 - L_4 + L_5 - L_6$$
(118)

$$d_8 = L_8 \tag{119}$$



Figure 147: 8th-order, SP CI-FBFF Topology 159







Figure 149: 8th-order, 4-path CI-FBFF Topology

Appendix **B**

The designed A/D CI-RFB VBP Σ - Δ modulators are depicted in this Appendix. Their symbolic NTFs are also provided.

Symbolic NTF for 2nd-order CR-RFB Topology:

$$NTF_{CR-RFB,2^{nd}}(z) = \frac{1+K_1z^{-1}+z^{-2}}{1+(L_1+K_1)z^{-1}+(L_2+1)z^{-2}}$$
(120)



Figure 150: 2nd-order, SP CR-RFB Topology



Figure 151: 2nd-order, 2-path CR-RFB Topology



Figure 152: 2nd-order, 4-path CR-RFB Topology

Symbolic NTF for 4th-order CR-RFB Topology: For the 4th-, 6th- and 8th-orders of the CR-RFB topologies, the MATLAB codes are provided those are used to calculate the NTFs' coefficients.

* * *

$$NTF_{CR-RFB,4^{\#}}(z) = \frac{\left(1 + K_1 z^{-1} + z^{-2}\right) \left(1 + K_2 z^{-1} + z^{-2}\right)}{1 + d_2 z^{-1} + d_3 z^{-2} + d_4 z^{-3} + d_5 z^{-4}}$$
(121)

```
[num1, den1] = butter(order1, [freq1*2, freq2*2], 'stop');
num1=num1/num1(1);
num=num1;
den=den1;
Z1=roots(num);
Z1_1=[Z1(1),Z1(2)];
Z1_2=[Z1(3), Z1(4)];
zer_pol1=poly(Z1_1);
zer_pol2=poly(Z1_2);
K1=zer_pol1(2);
K2=zer_pol2(2);
A=[0 1 0 0; 1 K2 0 1; K2 1 1 0; 1 0 0 0];
B=[(den(5)-1); (den(4)-K1-K2); (den(3)-K1*K2-2); (den(2)-K1-K2)];
C=linsolve(A,B);
L1=C(1);
L2=C(2);
L3=C(3);
L4 = C(4);
                                      * * *
```



Figure 153: 4th-order, SP CR-RFB Topology



Figure 154: 4th-order, 2-path CR-RFB Topology


Figure 155: 4th-order, 4-path CR-RFB Topology

Symbolic NTF for 6th-order CR-RFB Topology:

$$NTF_{CR-RFB,6^{th}}(z) = \frac{\left(1 + K_1 z^{-1} + z^{-2}\right) \left(1 + K_2 z^{-1} + z^{-2}\right) \left(1 + K_3 z^{-1} + z^{-2}\right)}{1 + d_2 z^{-1} + d_3 z^{-2} + d_4 z^{-3} + d_5 z^{-4} + d_6 z^{-5} + d_7 z^{-6}}$$
(122)

* * *

```
[num1, den1] = butter(order1, [freq1*2, freq2*2], 'stop');
num1=num1/num1(1);
num=num1;
den=den1;
Z1=roots(num);
Z1_1=[Z1(1),Z1(2)];
Z1_2=[Z1(3), Z1(4)];
Z1_3=[Z1(5), Z1(6)];
zer_pol1=poly(Z1_1);
zer_pol2=poly(Z1_2);
zer_pol3=poly(Z1_3);
K1=zer_pol1(2);
K2=zer_pol2(2);
K3=zer_pol3(2);
A=[0 1 0 0 0 0; 1 K2+K3 0 1 0 0;
K2+K3 K2*K3+2 1 K3 0 1;
2+K2*K3 K2+K3 L3 1 1 0 ;
K2+K3 1 1 0 0 0;
1 0 0 0 0 0];
B=[(den(7)-1);(den(6)-K1-K2-K3);
(den(5)-K1*K2-K1*K3-K2*K3-3);
(den(4)-2*K1-2*K2-2*K3-K1*K2*K3);
(den(3)-K1*K2-K1*K3-K2*K3-3);
(den(2)-K1-K2-K3)];
C=linsolve(A,B);
L1=C(1);
L2=C(2);
L3=C(3);
L4=C(4);
L5=C(5);
L6=C(6);
                                      * * *
```

```
167
```



Figure 156: 6th-order, SP CR-RFB Topology



Figure 157: 6th-order, 2-path CR-RFB Topology



Figure 158: 6th-order, 4-path CR-RFB Topology

Symbolic NTF for 8th-order CR-RFB Topology:

$$NTF_{CR-RFB,8^{th}}(z) = \frac{\left(1 + K_1 z^{-1} + z^{-2}\right) \left(1 + K_2 z^{-1} + z^{-2}\right) \left(1 + K_3 z^{-1} + z^{-2}\right) \left(1 + K_3 z^{-1} + z^{-2}\right)}{1 + d_2 z^{-1} + d_3 z^{-2} + d_4 z^{-3} + d_5 z^{-4} + d_6 z^{-5} + d_7 z^{-6} + d_8 z^{-7} + d_9 z^{-8}}$$

* * *

(123)

```
[num1, den1] = butter(order1, [freq1*2, freq2*2], 'stop');
num1=num1/num1(1);
num=num1;
den=den1;
Z1=roots(num);
Z1_1=[Z1(1),Z1(2)];
Z1_2=[Z1(3), Z1(4)];
Z1_3=[Z1(5), Z1(6)];
Z1_4 = [Z1(7), Z1(8)];
zer_pol1=poly(Z1_1);
zer_pol2=poly(Z1_2);
zer_pol3=poly(Z1_3);
zer_pol4=poly(Z1_4);
K1=zer_pol1(2);
K2=zer_pol2(2);
K3=zer_pol3(2);
K4=zer_pol3(2);
A = [0 1 0 0 0 0 0 0;
1 (K2+K3+K4) 0 1 0 0 0;
(K2+K3+K4) (K2*K3+K2*K4+K3*K4+3) 1 (K3+K4) 0 1 0 0; (K2*K3+K2*K4+K3*K4+3)
(2*K2+2*K3+K2*K3*K4+2*K4) (K3+K4) (K3*K4+2) 1 K4 0 1;
(2*K2+2*K3+2*K4+K2*K3*K4) (K2*K3+K2*K4+K3*K4+3) (K3*K4+2) (K3+K4) K4 1 1 0;
(K2*K3+K2*L4+K3*K4+3) (K2+K3+K4) (K3+K4) 1 1 0 0 0; (K2+K3+K4) 1 1 0 0 0 0
0;
1 0 0 0 0 0 0 0];
```

```
B=[(den(9)-1);
(den(8)-K1-K2-K3-K4);
(den(7)-K1*K2- K1*K3-K1*K4-K2*K3-K2*K4-K3*K4-4);
(den(6)-3*K1 - 3*K2 - 3*K3 - 3*K4 - K1*K2*K3 - K1*K2*K4 - K1*K3*K4 -
K2*K3*K4);
(den(5)-2*K1*K2-2*K1*K3-2*K1*K4-2*K2*K3-2*K2*K4-2*K3*K4-6-K1*K2*K3*K4);
(den(4)-3*K1-3*K2-3*K3-3*K4-K1*K2*K3-K1*K2*K4-K1*K3*K4-K2*K3*K4 );
(den(3)-K1*K2-K1*K3-K1*K4-K2*K3-K2*K4-K3*K4-4); (den(2)-K1-K2-K3-K4)];
```

```
C=linsolve(A,B);
L1=C(1);
L2=C(2);
```

L3=C(3); L4=C(4); L5=C(5);

L6=C(6);

L7=C(7);

L8=C(8);



Figure 159: 8th-order, SP CR-RFB Topology



Figure 160: 8th-order, 2-path CR-RFB Topology



Figure 161: 8th-order, 4-path CR-RFB Topology

Appendix C

The designed D/A VBP Σ - Δ modulators of the EF topology are depicted in this Appendix. Their symbolic NTFs are also provided.

Symbolic NTF for 2nd-order EF Topology:

$$NTF_{EF,2^{nd}}(z) = \frac{1 + (L_1 - K_1)z^{-1} + (L_2 - K_2)z^{-2}}{1 + L_1 z^{-1} + L_2 z^{-2}}$$
(124)



Figure 162: 2nd-order, SP EF Topology



Figure 163: 2nd-order, 2-path EF Topology



Figure 164: 2nd-order, 4-path EF Topology

Symbolic NTF for 4th-order EF Topology:

$$NTF_{EF,4^{th}}(z) = \frac{1 + (L_1 - K_1)z^{-1} + (L_2 - K_2)z^{-2} + (L_3 - K_3)z^{-3} + (L_4 - K_4)z^{-4}}{1 + L_1z^{-1} + L_2z^{-2} + L_3z^{-3} + L_4z^{-4}}$$
(125)



Figure 165: 4th-order, SP EF Topology



Figure 166: 4th-order, 2-path EF Topology



Figure 167: 4th-order, 4-path EF Topology

Symbolic NTF for 6th-order EF Topology:

$$NTF_{EF,6^{th}}(z) = \frac{1 + (L_1 - K_1)z^{-1} + (L_2 - K_2)z^{-2} + (L_3 - K_3)z^{-3} + (L_4 - K_4)z^{-4} + (L_5 - K_5)z^{-5} + (L_6 - K_6)z^{-6}}{1 + L_1z^{-1} + L_2z^{-2} + L_3z^{-3} + L_4z^{-4} + L_5z^{-5} + L_6z^{-6}}$$





Figure 168: 6th-order, SP EF Topology



Figure 169: 6th-order, 2-path EF Topology



Figure 170: 8th-order, 4-path EF Topology

Appendix D

The designed D/A VBP Σ - Δ modulators of the OF topology are depicted in this Appendix. Their symbolic NTFs are also provided.

Symbolic NTF for 2nd-order OF Topology:

$$NTF_{OF,2^{nd}}(z) = \frac{1 + L_1 z^{-1} + z^{-2}}{1 + (L_1 + K_1) z^{-1} + (1 + K_2) z^{-2}}$$
(127)



Figure 171: 2nd-order, SP OF Topology



Figure 172: 2nd-order, 2-path OF Topology





Symbolic NTF for 4th-order OF Topology:

$$NTF_{OF,4^{\#}}(z) = \frac{1 + L_1 z^{-1} + L_2 z^{-2} + L_1 z^{-3} + z^{-4}}{1 + (L_1 + K_1) z^{-1} + (L_2 + K_2) z^{-2} + (L_1 + K_3) z^{-3} + (1 + K_4) z^{-4}}$$
(128)



Figure 174: 4th-order, SP OF Topology



Figure 175: 4th-order, 2-path OF Topology



Figure 176: 4th-order, 4-path OF Topology

Symbolic NTF for 6th-order OF Topology:

$$NTF_{OF,6^{th}}(z) = \frac{num_{6^{th}}}{den_{6^{th}}}$$
(129)

$$num_{6^{th}} = 1 + L_{1}z^{-1} + L_{2}z^{-2} + L_{3}z^{-3} + L_{2}z^{-4} + L_{1}z^{-5} + z^{-6}$$

$$den_{6^{th}} = 1 + (L_{1} + K_{1})z^{-1} + (L_{2} + K_{2})z^{-2} + (L_{3} + K_{3})z^{-3} + (L_{2} + K_{4})z^{-4} + (L_{1} + K_{5})z^{-5} + (1 + K_{6})z^{-6}$$

(130)



Figure 177: 6th-order, SP OF Topology 189



Figure 178: 6th-order, 2-path OF Topology



Figure 179: 6th-order, 4-path OF Topology

REFERENCES

[1] Yong Ping Xu; Rui Yu, "Electromechanical resonator based bandpass sigma-delta modulator for wireless transceivers," *in Radio-Frequency Integration Technology: Integrated Circuits for Wideband Communication and Wireless Sensor Networks, 2005. Proceedings.* 2005 IEEE International Workshop on, vol., no., pp.101,104, 30 Nov.-2 Dec. 2005

[2] Inhee Lee; Gunhee Han; Youngcheol Chae, "A 2 mW, 50 dB DR, 10 MHz BW 5 x Interleaved Bandpass Delta-Sigma Modulator at 50 MHz IF," *in Circuits and Systems I: Regular Papers, IEEE Transactions on,* vol.62, no.1, pp.80, 89, Jan. 2015

[3] Liang Zou; Karthaus, U.; Sukumaran, D.; Mehrtash, N.; Wagner, H., "A 6th order, 700– 1100 MHz, 3.6 Gb/s RF bandpass ΣΔ ADC with two-tone SFDR 67.2 dB in 65nm CMOS," *in Circuits and Systems (ISCAS), 2014 IEEE International Symposium on,* vol., no., pp.714, 717, 1-5 June 2014

[4] Ucar, A.; Cetin, E.; Kale, I., "On the implications of Analog-to-Digital Conversion on variable-rate BandPass Sampling GNSS receivers," *in Signals, Systems and Computers, 2008 42nd Asilomar Conference on*, vol., no., pp.2081, 2085, 26-29 Oct. 2008

[5] Elsayed, F.; Helaoui, M., "Linearized Multi-Level Σ - Δ Modulated Wireless Transmitters for SDR Applications Using Simple DLGA Algorithm," *in Emerging and Selected Topics in Circuits and Systems, IEEE Journal on*, vol.3, no.4, pp.594, 601, Dec. 2013

[6] Chen-Yen Ho; Wei-Shan Chan; Yung-Yu Lin; Tsung-Hsien Lin, "A Quadrature Bandpass Continuous-Time Delta-Sigma Modulator for a Tri-Mode GSM-EDGE/UMTS/DVB-T Receiver," *in Solid-State Circuits, IEEE Journal of,* vol.46, no.11, pp.2571, 2582, Nov. 2011

[7] Norsworthy, S., Schreier, R., Temes, G., "Delta-Sigma Data Converters: Theory, Design, and Simulation". "*Wiley-IEEE Press*", 1997

[8] Candy, J., Temes, G., "Oversampling Delta-Sigma Data Converters: Theory, Design, and Simulation". "*Wiley-IEEE Press*", 1992

[9] Ardalan, S.H.; Paulos, J.J., "An analysis of nonlinear behavior in delta - sigma modulators," *in Circuits and Systems, IEEE Transactions on*, vol.34, no.6, pp.593, 603, Jun 1987

[10] Candy, J., Temes, G., "Design of Stable High Order 1 Bit Sigma Delta Modulators". "Wiley-IEEE Press", 1992

[11] Namdar, A.; Leung, B.H., "A 400-MHz, 12-bit, 18-mW, IF digitizer with mixer inside a sigma-delta modulator loop," *in Solid-State Circuits, IEEE Journal of,* vol.34, no.12, pp.1765, 1776, Dec 1999

[12] Shibata, H.; Schreier, R.; Wenhua Yang; Shaikh, A.; Paterson, D.; Caldwell, T.C.; Alldred, D.; Ping Wing Lai, "A DC-to-1 GHz Tunable RF Σ - Δ ADC Achieving DR=74 dB and BW=150 MHz at f₀ = 450 MHz Using 550 mW," *in Solid-State Circuits, IEEE Journal of,* vol.47, no.12, pp.2888, 2897, Dec. 2012

[13] Inhee Lee; Gunhee Han; Youngcheol Chae, "A 2 mW, 50 dB DR, 10 MHz BW 5xInterleaved Bandpass Delta-Sigma Modulator at 50 MHz IF," *in Circuits and Systems I: Regular Papers, IEEE Transactions on,* vol.62, no.1, pp.80, 89, Jan. 2015

[14] Bing Li; Kong-Pang Pun, "A High Image-Rejection SC Quadrature Bandpass DSM for Low-IF Receivers," *in Circuits and Systems I: Regular Papers, IEEE Transactions on,* vol.61, no.1, pp.92, 105, Jan. 2014

[15] S. Jantzi, K. Martin, and A. Sedra, "Quadrature bandpass delta; sigma; modulation for digital radio," *in Solid-State Circuits, IEEE Journal of*, vol. 32, no. 12, pp. 1935–1950, Dec 1997.

[16] Sobot, R.; Stapleton, S., "Design Methodology for $\Sigma\Delta$ Modulators," *Research in Microelectronics and Electronics 2006, Ph.D.*, vol., no., pp.121, 124, 12-15 June 2006

[17] Kitayabu, T.; Ishikawa, H., "Generalized architecture of concurrent dual-band transmitter for spectrum aggregation system," *in Personal Indoor and Mobile Radio Communications* (PIMRC), 2010 IEEE 21st International Symposium on, vol., no., pp.111, 116, 26-30 Sept. 2010

[18] Kumar, Y.B.N.; Patra, A.; Maloberti, F., "Reconfigurable multi-band quadrature $\Sigma\Delta$ modulators," *Ph.D. Research in Microelectronics and Electronics (PRIME), 2011 7th Conference on*, vol., no., pp.237, 240, 3-7 July 2011

[19] A. Eshraghi and T. Fiez, "A comparative analysis of parallel delta- sigma ADC architectures," *in Circuits and Systems I: Regular Papers, IEEE Transactions on*, vol. 51, no. 3, pp. 450 – 458, March 2004

[20] M. Kozak and I. Kale, "Novel topologies for time-interleaved delta- sigma modulators," *in Circuits and Systems II: Analog and Digital Signal Processing, IEEE Transactions on*, vol. 47, no. 7, pp. 639–654, July 2000.

[21] Lorenz, M.; Ritter, R.; Anders, J.; Ortmanns, M., "Estimation of Non-Idealities in Sigma-Delta Modulators for Test and Correction Using Unscented Kalman Filters," in *Circuits and Systems I: Regular Papers, IEEE Transactions on*, vol.62, no.5, pp.1240-1249, May 2015

[22] R.J. Marks II: Introduction to Shannon Sampling and Interpolation Theory, *Spinger-Verlag*, 1991

[23] Black, H.S.; Edson, J.O., "Pulse Code Modulation," *American Institute of Electrical Engineers, Transactions of the*, vol.66, no.1, pp.895, 899, Jan. 1947

[24] IEEE Standard for Terminology and Test Methods for Analog-to-Digital Converters, IEEE Standard 1241-2010

[25] Smith, Steven W., The Scientist's and Engineer's Guide to Digital Signal Processing (Second Edition), *San Diego CA: California Technical Publishing*, 1999 (ISBN 0-9660176-6-8), p. 60.

[26] Inose, H.; Yasuda, Y.; Murakami, J., "A Telemetering System by Code Modulation – Δ - Σ Modulation," *in Space Electronics and Telemetry, IRE Transactions on*, vol. SET-8, no.3, pp.204, 209, Sept. 1962

[27] Heng-Yu Jian; Xu, Z.; Yi-Cheng Wu; Chang, M.-C.F., "A Fractional-N PLL for Multiband (0.8–6 GHz) Communications Using Binary-Weighted D/A Differentiator and Offset-Frequency Δ - Σ Modulator," *in Solid-State Circuits, IEEE Journal of,* vol.45, no.4, pp.768, 780, April 2010

[28] X. Wu, V. A. Chouliaras, J. L. Nunez, and R. M. Goodall, "A novel control system processor and its VLSI implementation," *IEEE Trans. Very Large Scale Integr. (VLSI) Syst.*, vol. 16, no. 3, pp. 217–228, Mar. 2008

[29] Ostrovskyy, P.; Gustat, H.; Ortmanns, M.; Christoph Scheytt, J., "A 5-Gb/s 2.1–2.2-GHz Bandpass $\Delta\Sigma$ Modulator for Switch-Mode Power Amplifier," *Microwave Theory and Techniques, IEEE Transactions on*, vol.60, no.8, pp.2524, 2531, Aug. 2012

[30] Angus, A.C.R.; Reekie, H.M., "STF behaviour in optimised for ELD cascaded CT Delta-Sigma Modulators," *in Circuits and Systems (ISCAS), 2012 IEEE International Symposium on,* vol., no., pp.57, 60, 20-23 May 2012

[31] Van Tam Nguyen; Khushk, H.A.; Jabbour, C.; Loumeau, P., "High pass filter implementation comparison in unity STF high pass $\Delta\Sigma$ modulator," *in Electronics, Circuits and Systems (ICECS), 2011 18th IEEE International Conference on*, vol., no., pp.101, 104, 11-14 Dec. 2011

[32] Vaidyanathan, P.P., "Multirate digital filters, filter banks, polyphase networks, and applications: a tutorial," *Proceedings of the IEEE*, vol.78, no.1, pp.56, 93, Jan 1990

[33] Park, S., "A real-time implementation of half-band filters to obtain 18-20 bit resolution from the DSP56ADC16 sigma-delta A/D converter," *in Acoustics, Speech, and Signal Processing, 1990. ICASSP-90., 1990 International Conference on,* vol., no., pp.989, 992 vol.2, 3-6 Apr 1990

[34] Horbach, U., "Design of a 20 bit sigma-delta A/D-converter for audio applications," *in Circuits and Systems, 1990., IEEE International Symposium on,* vol., no., pp.2789, 2792 vol.4, 1-3 May 1990

[35] Shiang-Hwua Yu, "Analysis and design of single-bit sigma-delta Modulators using the theory of sliding modes," Control Systems Technology, IEEE Transactions on, vol.14, no.2, pp.336, 345, March 2006

[36] Schreier, R., "An empirical study of high-order single-bit delta-sigma modulators," *in Circuits and Systems II: Analog and Digital Signal Processing, IEEE Transactions on*, vol.40, no.8, pp.461, 466, Aug 1993

[37] Bichan, M.; Dunwell, D.; Qiwei Wang; Carusone, A.C., "A passive resonant clocking network for distribution of a 2.5-GHz clock in a flash ADC," *in Circuits and Systems (ISCAS)*, 2014 IEEE International Symposium on, vol., no., pp.1320,1323, 1-5 June 2014

[38] Megha, R.; Pradeepkumar, K.A., "Implementation of low power flash ADC by reducing comparators," *in Communications and Signal Processing (ICCSP), 2014 International Conference on*, vol., no., pp.443, 447, 3-5 April 2014

[39] Inamdar, A.; Sahu, A.; Jie Ren; Setoodeh, S.; Mansour, R.; Gupta, D., "Design and Evaluation of Flash ADC," *Applied Superconductivity, IEEE Transactions on*, vol.25, no.3, pp.1, 5, June 2015

[40] S. Rylov et al., "3.1 A 25Gb/s ADC-based serial line receiver in 32nm CMOS SOI," 2016 *IEEE International Solid-State Circuits Conference (ISSCC)*, San Francisco, CA, 2016, pp. 56-57

[41] X. Yang, G. Cui, Y. Zhang, J. Ren and J. Liu, "A Metastability Error Detection and Reduction Technique for Partially Active Flash ADCs," in *IEEE Transactions on Circuits and Systems II: Express Briefs*, vol. 63, no. 4, pp. 331-335, April 2016

[42] G. Tretter, M. M. Khafaji, D. Fritsche, C. Carta and F. Ellinger, "Design and Characterization of a 3-bit 24-GS/s Flash ADC in 28-nm Low-Power Digital CMOS," in *IEEE Transactions on Microwave Theory and Techniques*, vol. 64, no. 4, pp. 1143-1152, April 2016.

[43] Lin, J.; Paik, D.; Seungjong Lee; Miyahara, M.; Matsuzawa, A., "An Ultra-Low-Voltage

160 MS/s 7 Bit Interpolated Pipeline ADC Using Dynamic Amplifiers," *Solid-State Circuits, IEEE Journal of*, vol.50, no.6, pp.1399, 1411, June 2015

[44] J. Gao, G. Li, L. Huang and Q. Li, "An Amplifier-Free Pipeline-SAR ADC Architecture With Enhanced Speed and Energy Efficiency," in *IEEE Transactions on Circuits and Systems II: Express Briefs*, vol. 63, no. 4, pp. 341-345, April 2016

[45] Ren, S.; Emmert, J., "Successive approximation pipelined ADC with one clock cycle conversion rate," *Electronics Letters*, vol.48, no.20, pp.1257, 1258, September 27 2012

[46] K. T. Lin, Y. W. Cheng and K. T. Tang, "A 0.5 V 1.28-MS/s 4.68-fJ/Conversion-Step SAR ADC With Energy-Efficient DAC and Trilevel Switching Scheme," in *IEEE Transactions on Very Large Scale Integration (VLSI) Systems*, vol. 24, no. 4, pp. 1441-1449, April 2016

[47] Jian Li; Xiaoyang Zeng; Lei Xie; Jun Chen; Jianyun Zhang; Yawei Guo, "A 1.8-V 22mW 10-bit 30-MS/s Subsampling Pipelined CMOS ADC," *Custom Integrated Circuits Conference, 2006. CICC '06. IEEE*, vol., no., pp.513, 516, 10-13 Sept. 2006

[48] Min Gyu Kim; Gil-Cho Ahn; Hanumolu, P.K.; Sang-Hyeon Lee; Sang-Ho Kim; Seung-Bin You; Jae-Whui Kim; Temes, G.C.; Un-Ku Moon, "A 0.9 V 92 dB Double-Sampled Switched-RC Delta-Sigma Audio ADC," *Solid-State Ci*rcuits, IEEE Journal of, vol.43, no.5, pp.1195, 1206, May 2008

[49] Gil-Cho Ahn; Dong-Young Chang; Brown, M.E.; Ozaki, N.; Youra, H.; Yamamura, K.; Hamashita, K.; Takasuka, K.; Temes, G.C.; Un-Ku Moon, "A 0.6-V 82-dB delta-sigma audio ADC using switched-RC integrators," *Solid-State Circuits, IEEE Journal of,* vol.40, no.12, pp.2398, 2407, Dec. 2005

[50] Benabes, P.; Kielbasa, R., "Low-consumption sigma-delta DAC for audio mobile applications," *Electronics Letters*, vol.44, no.17, pp.1010, 1011, Aug. 14 2008

[51] Hart, A.; Voinigescu, S.P., "A 1 GHz Bandwidth Low-Pass Sigma-Delta ADC With 20– 50 GHz Adjustable Sampling Rate," *Solid-State Circuits, IEEE Journal of,* vol.44, no.5,

pp.1401, 1414, May 2009

[52] Gupta, S.; Gangopadhyay, D.; Lakdawala, H.; Rudell, J.C.; Allstot, D.J., "A 0.8–2 GHz Fully-Integrated QPLL-Timed Direct-RF-Sampling Bandpass Sigma Delta ADC in 0.13 μ m CMOS," *Solid-State Circuits, IEEE Journal of,* vol.47, no.5, pp.1141, 1153, May 2012

[53] Schreier, R.; Lloyd, J.; Singer, L.; Paterson, D.; Timko, M.; Hensley, M.; Patterson, G.; Behel, K.; Zhou, J., "A 10-300-MHz IF-digitizing IC with 90-105-dB dynamic range and 15-333-kHz bandwidth," in *Solid-State Circuits, IEEE Journal of*, vol.37, no.12, pp.1636-1644, Dec 2002

[54] R. Ritter, J. G. Kauffman, J. Becker and M. Ortmanns, "A 10 MHz Bandwidth, 70 dB SNDR Continuous Time Delta-Sigma Modulator With Digitally Improved Reconfigurable Blocker Rejection," in *IEEE Journal of Solid-State Circuits*, vol. 51, no. 3, pp. 660-670, March 2016

[55] T. Oshita *et al.*, "A Compact First-Order Sigma-Delta Modulator for Analog High-Volume Testing of Complex System-on-Chips in a 14 nm Tri-Gate Digital CMOS Process," in *IEEE Journal of Solid-State Circuits*, vol. 51, no. 2, pp. 378-390, Feb. 2016

[56] Inhee Lee; Gunhee Han; Youngcheol Chae, "A 2 mW, 50 dB DR, 10 MHz BW 5 x Interleaved Bandpass Delta-Sigma Modulator at 50 MHz IF," *Circuits and Systems I: Regular Papers, IEEE Transactions on,* vol.62, no.1, pp.80, 89, Jan. 2015

[57] T. Podsiadlik and R. Farrell, "Time-Interleaved Sigma Delta Modulators for FPGAs," in *IEEE Transactions on Circuits and Systems II: Express Briefs*, vol. 61, no. 10, pp. 808-812, Oct. 2014

[58] A. Olmos, T. Miyashita, M. Nihei, E. Charry and Y. Watanabe, "A 5 GHz continuous time sigma-delta modulator implemented in 0.4 μm InGaP/InGaAs HEMT technology," *Circuits and Systems, 1998. ISCAS '98. Proceedings of the 1998 IEEE International Symposium on*, Monterey, CA, 1998, pp. 575-578 vol.1

[59] A. Labaziewicz and R. J. Baker, "A 2 GHz effective sampling frequency K-Delta-1-Sigma analog-to-digital converter," *2011 IEEE 54th International Midwest Symposium on Circuits and Systems (MWSCAS)*, Seoul, 2011, pp. 1-4

[60] Minho Kwon; Jungyoon Lee; Gunhee Han, "A time-interleaved recursive loop band-pass delta-sigma modulator for a digital IF CDMA receiver," *Circuits and Systems II: Express Briefs, IEEE Transactions on,* vol.52, no.7, pp.389, 393, July 2005

[61] Martens, E.; Bourdoux, A.; Couvreur, A.; Fasthuber, R.; Van Wesemael, P.; Van der Plas, G.; Craninckx, J.; Ryckaert, J., "RF-to-Baseband Digitization in 40 nm CMOS With RF Bandpass Sigma Delta Modulator and Polyphase Decimation Filter," *Solid-State Circuits, IEEE Journal of,* vol.47, no.4, pp.990, 1002, April 2012

[62] Kye-shin Lee; Maloberti, F., "Time-interleaved sigma-delta modulator using output prediction scheme," *Circuits and Systems II: Express Briefs, IEEE Transactions on,* vol.51, no.10, pp.537, 541, Oct. 2004

[63] Hamidi, B.; Miar-Naimi, H., "Extended Noise Shaping in Sigma-Delta Modulator Using Cross-Coupled Paths," *Circuits and Systems I: Regular Papers, IEEE Transactions o*, vol.61, no.6, pp.1675, 1686, June 2014

[64] Bhide, A.; Ojani, A.; Alvandpour, A., "Effect of Clock Duty-Cycle Error on Two-Channel Interleaved Sigma Delta DACs," *Circuits and Systems II: Express Briefs, IEEE Transactions on*, vol.62, no.7, pp.646, 650, July 2015

[65] Khoini-Poorfard, R.; Lim, L.B.; Johns, D.A., "Time-interleaved oversampling A/D converters: theory and practice," *Circuits and Systems II: Analog and Digital Signal Processing, IEEE Transactions on*, vol.44, no.8, pp.634, 645, Aug 1997

[66] E. I. Jury and B. W. Lee., "The absolute stability of systems with many nonlinearities," *Automcrtw~ Remofe Conrr.*, vol. 26, pp. 943-961, 1965

[67] B.P. Agrawal and K. Shenoi, "Design Methodology for Sigma Delta Modulators," *IEEE Transactions on Communications*, vol. COM-31, pp.360-370, March 1983

[68] J. D. Reiss and M. B. Sandler, "A Mechanism for the Detection and Removal of Limit Cycles in the Operation of Sigma–Delta Modulators," British Patent Application No. 0514677.4, UK patent filed July 18th 2005

[69] J. M. de la Rosa, B. Perez- Verdu, F. Medeiro, R. del Rio, and A. Rodriguez-Vazquez, "Analysis and Experimental Characterization of Idle Tones in 2nd-Order Bandpass Sigma– Delta Modulators—A 0.8 μm CMOS Switched-Current Case Study," *Proc. IEEE Int. Sym. on Circuits and Systems, ISCAS 2001*, pp. 774–777 (2001)

[70] A portable radio direction finder," in *A.I.E.E., Journal of the*, vol.46, no.2, pp.131-131, Feb. 1927

[71] Tucker, D.G., "The history of the homodyne and synchrodyne," *Radio Engineers, Journal of the British Institution of*, vol.14, no.4, pp.143, 154, April 1954

[72] Crols, J.; Steyaert, M.S.J., "Low-IF topologies for high-performance analog front ends of fully integrated receivers," *Circuits and Systems II: Analog and Digital Signal Processing, IEEE Transactions on*, vol.45, no.3, pp.269, 282, Mar 1998

[73] Zhu, Y.; Al-Sarawi, S.F.; Lim, C.C.; Liebelt, M.J., "Fourth-Order Discrete-Time Variable Centre Frequency Bandpass Sigma-Delta Modulator," *Circuits and Systems, 2006. APCCAS* 2006. *IEEE Asia Pacific Conference on*, vol., no., pp.820, 823, 4-7 Dec. 2006

[74] Yamamoto, K.; Carusone, A.C.; Dawson, F.P., "A Delta-Sigma Modulator With a Widely Programmable Center Frequency and 82-dB Peak SNDR," *Solid-State Circuits, IEEE Journal of,* vol.43, no.8, pp.1772, 1782, Aug. 2008

[75] Shibata, H.; Schreier, R.; Wenhua Yang; Shaikh, A.; Paterson, D.; Caldwell, T.C.; Alldred, D.; Ping Wing Lai, "A DC-to-1 GHz Tunable RF Sigma Delta ADC Achieving DR - 74 dB and BW - 150 MHz at - 450 MHz Using 550 mW," *Solid-State Circuits, IEEE Journal of,* vol.47, no.12, pp.2888, 2897, Dec. 2012

[76] Alam, M.K.; Khan, F.H., "A high-efficiency modular switched-capacitor converter with continuously variable conversion ratio," *Control and Modeling for Power Electronics (COMPEL), 2012 IEEE 13th Workshop on,* vol., no., pp.1,5, 10-13 June 2012

[77] Lee, S.-S.; Uchida, H.; Soda, S.; Nishino, T.; Inoue, H.; Izuo, S.; Yoshida, Y.; Miyazaki, M., "An Rfmems Switched Capacitor Array for a Tunable Band Pass Filter," *Solid-State Sensors, Actuators and Microsystems Conference, 2007. TRANSDUCERS 2007. International,* vol., no., pp.157,160, 10-14 June 2007.

[78] Jensen, J.F.; Fields, C.H.; Chang, M.F., "A 2GS/s 3b ΔΣ-modulated DAC with a tunable switched-capacitor bandpass DAC mismatch shaper," *Solid-State Circuits Conference, 2004. Digest of Technical Papers. ISSCC. 2004 IEEE International, vol.*, no., pp.366, 533 Vol.1, 15-19 Feb. 2004

[79] Shih-Jui Chen,; Chuang-yuan Lee; Eun Sok Kim, "Integration of tunable capacitors and bonded-wires for contactless RF switch and tunable filter," *Solid-State Sensors, Actuators and Microsystems Conference, 2009. TRANSDUCERS 2009. International,* vol., no., pp.545, 548, 21-25 June 2009

[80] Dae-Hyun Baek; Youngkee Eun; Dae-Sung Kwon; Min-Ook Kim; Taeyoung Chung; Jongbaeg Kim, "Widely Tunable Variable Capacitor With Switching and Latching Mechanisms," *Electron Device Letters, IEEE*, vol.36, no.2, pp.186, 188, Feb. 2015

[81] Mladenov, V.; Hegt, H.; van Roermund, A., "On the stability of high order Sigma-Delta modulators," *Electronics, Circuits and Systems, 2001. ICECS 2001. The 8th IEEE International Conference on,* vol.3, no., pp.1383, a1386 vol.3, 2001

[82] Beydoun, A.; Benabes, P.; Oksman, J., "Optimization of the noise transfer function of Extended-Frequency-Band-Decomposition sigma-delta A/D converters," *Circuits and Systems and TAISA Conference, 2008. NEWCAS-TAISA 2008. 2008 Joint 6th International IEEE Northeast Workshop on*, vol., no., pp.193,196, 22-25 June 2008
[83] Pham, D.-K.G.; Desgreys, P.; Loumeau, P.; Ridgers, T., "Multi-stage noise band cancellation $\Sigma\Delta$ modulator for digitisation of distorted signals," *Electronics Letters*, vol.48, no.10, pp.560,562, May 10 2012

[84] Yazkurt, U.; Dundar, G.; Talay, S.; Beilleau, N.; Aboushady, H.; de Lamarre, L., "Scaling Input Signal Swings of Overloaded Integrators in Resonator-based Sigma-Delta Modulators," *in Electronics, Circuits and Systems, 2006. ICECS '06. 13th IEEE International Conference on,* vol., no., pp.966-969, 10-13 Dec. 2006

[85] Yang Xu; Baoyong Chi; Zhihua Wang, "Power-scalable multi-mode reconfigurable continuous-time lowpass/quadrature bandpass sigma-delta modulator for zero/low-IF receivers," *in Circuits and Systems (ISCAS), 2012 IEEE International Symposium on,* vol., no., pp.293-296, 20-23 May 2012

[86] Wulff, C.; Ytterdal, T., "Resonators in Open-Loop Sigma–Delta Modulators," *in Circuits and Systems I: Regular Papers, IEEE Transactions on*, vol.56, no.10, pp.2159-2172, Oct. 2009

[87] Chon-In Lao; Seng-Pan U; Martins, R.P., "A novel effective bandpass semi-MASH sigma-delta modulator with double-sampling mismatch-free resonator," *in Circuits and Systems, 2006. ISCAS 2006. Proceedings. 2006 IEEE International Symposium on,* vol., no., pp.4 pp.-, 21-24 May 2006

[88] Xiaowei Liu; Honglin Xu; Jiajun Zhou; Song Chen; Jiang Yang, "An improved fourthorder band-pass sigma-delta modulator with a novel resonator," *in Optoelectronics and Microelectronics (ICOM), 2012 International Conference on,* vol., no., pp.510-513, 23-25 Aug. 2012

[89] Salo, T.; Lindfors, S.; Halonen, K., "A low-voltage single-Opamp 4th-order band-pass $\Delta\Sigma$ -modulator," *in Circuits and Systems, 2001. ISCAS 2001. The 2001 IEEE International Symposium on*, vol.1, no., pp.352-355 vol. 1, 6-9 May 2001

[90] Gautam, J.K.; Kumar, A.; Saxena, R., "On the modified Bartlett-Hanning window (family)," *in Signal Processing, IEEE Transactions on*, vol.44, no.8, pp.2098-2102, Aug 1996

[91] S. Lee, J. Choi and S. Kim, "Pascal's triangle-based multihop range-free localization for anisotropic sensor networks," *2014 IEEE Wireless Communications and Networking Conference (WCNC)*, Istanbul, 2014, pp. 2739-2744

[92] Khoini-Poorfard, R.; Johns, D.A., "Mismatch effects in time-interleaved oversampling converters," *in Circuits and Systems, 1994. ISCAS '94., 1994 IEEE International Symposium on,* vol.5, no., pp.429-432 vol.5, 30 May-2 Jun 1994

[93] Galton, I., "Delta-sigma data conversion in wireless transceivers," in *Microwave Theory* and *Techniques, IEEE Transactions on*, vol.50, no.1, pp.302-315, Jan 2002

[94] Taleie, S.M.; Copani, T.; Bakkaloglu, B.; Kiaei, S., "A Linear Sigma Delta Digital IF to RF DAC Transmitter With Embedded Mixer," in *Microwave Theory and Techniques, IEEE Transactions on*, vol.56, no.5, pp.1059-1068, May 2008

[95] Hua Qian; Juan Chen; Saijie Yao; Zhenhua Yu Zhang; Hua Zhang; Wei Xu, "One-Bit Sigma-Delta Modulator for Nonlinear Visible Light Communication Systems," *in Photonics Technology Letters, IEEE*, vol.27, no.4, pp.419-422, Feb.15, 15 2015

[96] Athar, S.; Siddiqi, M.A.; Masud, S., "Design and FPGA Implementation of a 2nd Order Adaptive Delta Sigma Modulator with One Bit Quantization," *in Field Programmable Logic and Applications (FPL), 2010 International Conference on*, vol., no., pp.388-393, Aug. 31 2010-Sept. 2 2010

[97] Candy, J.; Temes, G., "Technology Scaling and Performance Limitations in DeltaSigma AnalogDigital Converters," in *Oversampling Delta-Sigma Data Converters:Theory, Design, and Simulation*, 1, Wiley-IEEE Press, 1992, pp.219-222

[98] Jintae Kim; Modjtahedi, S.; Chih-Kong Ken Yang, "Flexible-Assignment Calibration Technique for Mismatch-Constrained Digital-to-Analog Converters," in *Very Large Scale Integration (VLSI) Systems, IEEE Transactions on*, vol.22, no.9, pp.1934-1944, Sept. 2014 [99] Galton, I., "One-bit dithering in delta-sigma modulator-based D/A conversion," in *Circuits and Systems, 1993., ISCAS '93, 1993 IEEE International Symposium on,* vol., no., pp.1310-1313 vol.2, 3-6 May 1993

[100] Oppenheim, A.V., and Weinstein, C.J.: 'Effects of finite register length in digital signal processing', *Prc. IEEE, 1972,* 60, (8), pp. 957–976

[101] Liu, B.: 'Effects of finite word length on the accuracy of digital filters – a review', *IEEE Trans.*, *1971*, CT-18, (6), pp. 670–677

[102] Alser, M.H.; Assaad, M.; Hussin, F.A.; Yohannes, I., "Design and FPGA implementation of PLL-based quarter-rate clock and data recovery circuit," in *Intelligent and Advanced Systems (ICIAS), 2012 4th International Conference on*, vol.2, no., pp.825-830, 12-14 June 2012

[103] Fitzgibbon, B.; Kennedy, M.P., "Calculation of Cycle Lengths in Higher Order Error Feedback Modulators With Constant Inputs," in *Circuits and Systems II: Express Briefs, IEEE Transactions on*, vol.58, no.1, pp. 6-10, Jan. 2011

[104] S. Jantzi, C. Ouslis, and A. Sedra, "Transfer function design for $\Sigma\Delta$ converters," IEEE Proc. ISCAS, pp. 433-436, May 1994.

[105] Altinok, G.D.; Al-Janabi, M.; Kale, I., "A GUI driven Σ - Δ modulator design, evaluation and measurement tool with a view to practical implementation," in *Instrumentation and Measurement Technology Conference, 2009. I2MTC '09. IEEE*, vol., no., pp.1022-1025, 5-7 May 2009

[106] F. Medeiro, B. Perez-Verdu, and A. Rodriguez-Vazquez, "Top-down design of high-performance modulators," *Kluwer Academic Publishers*, Netherlands, 1999

BIOGRAPHY

Işıl Kalafat Kızılkaya was born in Sinop in 1985. She received the B.E. degree from Yıldız Technical University, İstanbul, Turkey, in 2008. She received the M.E. degree from Boğaziçi University, İstanbul, Turkey in 2011.

Her master's thesis is named 'RF Front-End of a Receiver for a 3 Tesla Magnetic Resonance Imager'. The thesis mainly focused on the design of a low-power fully differential Low Noise Amplifier (LNA), operating at 123 MHz.

From 2009 to 2011 she worked as a research assistant in Yıldız Technical University. In September 2011, she joined the Department of Engineering, University of Westminster, as a research scholar. Her main areas of research interest are analog and digital signal processing circuits' design and implementation, sigma-delta based A/D and D/A converters.